PNWD-3437 WTP-RPT-122, Rev. 0

Test Summary: WTP Flowsheet Evaluation of Mercury- Containing Hanford Waste Simulant

R. W. Goles* W.C. Buchmiller* L.K. Jagoda* B.D. MacIsaac** S. Rassat*

*Battelle—Pacific Northwest Division **Battelle Memorial Institute

April 2004

Prepared for Bechtel Hanford Incorporated under Contract 24590-101-TSA-W000-00004

LEGAL NOTICE

This report was prepared by Battelle Memorial Institute (Battelle) as an account of sponsored research activities. Neither Client nor Battelle nor any person acting on behalf of either:

MAKES ANY WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, with respect to the accuracy, completeness, or usefulness of the information contained in this report, or that the use of any information, apparatus, process, or composition disclosed in this report may not infringe privately owned rights; or

Assumes any liabilities with respect to the use of, or for damages resulting from the use of, any information, apparatus, process, or composition disclosed in this report.

Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by Battelle. The views and opinions of authors expressed herein do not necessarily state or reflect those of Battelle.



WTP Flowsheet Evaluation of Mercury-Containing Hanford Waste Simulant

R. W. Goles* W. C. Buchmiller* L.K. Jagoda* B.D. MacIsaac** S. Rassat*

*Battelle—Pacific Northwest Division **Battelle Memorial Institute

April 2004

Prepared for Bechtel National Inc. under Contract 24590-101-TSA-W000-00004

Test Specification: 24590-HLW-TSP-RT-03-004, Rev. 0 Test Plan: TP-RPP-WTP-252 Test Exceptions: None R&T Focus Area: HLW Vitrification Test Scoping Statement(s): B-91

54/204 for W. Tampaitis

ACCEPTED FOR PROJECT USE

Battelle—Pacific Northwest Division Richland, Washington 99352

Completeness of Testing

This report describes the results of work and testing specified by Test Specification 24590-PTF-3PI-MEVV-00001, Rev. A and Test Plan TPR-RPP-WTP-252. The work and any associated testing followed the quality assurance requirements outlined in the Test Specification/Plan. The descriptions provided in this test report are an accurate account of both the conduct of the work and the data collected. Test plan results are reported. Also reported are any unusual or anomalous occurrences that are different from expected results. The test results and this report have been reviewed and verified.

Approved:

Gordon H. Beeman, Manager WTP R&T Support Project

Summary

Mercury is found in the Hanford tank farm wastes. The chemistry of mercury under thermal processing conditions suggests that several different chemical species of mercury can be formed. The physical properties of those species provide different pathways for the mercury in the Waste Treatment Plant (WTP), which require different abatement schemes to control mercury releases.

Under current WTP operating conditions, the portion of the mercury that is scrubbed from the process exhaust will be periodically recycled to the Pretreatment Plant where it will be chemically adjusted, concentrated, and further partitioned between high-level waste (HLW), low-activity waste (LAW), the liquid effluent retention facility/effluent treatment facility (LERF/ETF), and pretreatment's vessel vent (VV) streams. The noncondensable mercury fraction in the melter's off-gas stream, on the other hand, will be actively abated using activated-charcoal sorption beds that form the initial processing component in the melter's secondary off-gas processing system. Without adequate abatement, mercury vapor, depending on its concentration, can threaten both catalytic unit operations in the Plant's secondary off-gas system and clean-air emission limits for mercury as well as other catalytically-abated, regulated volatiles (organics and NO_x).

Beyond process clean-air limits, there is a possibility that the mercury content of the LERF/ETF streams could exceed those facility's discharge limits. Consequently, partitioning fractions and chemical species produced by vitrification and subsequent pretreatment operations need to be quantified to determine off-gas design criteria that will mitigate the threat of excessive airborne and aqueous plant releases.

Toward this end, experimental evaluations were conducted to elucidate the vitrification flowsheet behavior of a mercury-containing Hanford-waste simulant (C-104/AY-101) under representative WTP operating conditions using a range of mercury, halogen, and reductant concentrations as testing variables. Specifically, Battelle—Pacific Northwest Division's Research-Scale Melter (RSM) was used to conduct seven separate processing campaigns over a ~120-h period that involved selected combinations of three mercury, two halogen, and two reductant feed concentrations. In addition, the secondary aqueous waste stream generated by the melter's quench-scrubber was used in a separate, post-melter test evaporator study to project mercury partitioning in LERF/ETF and VV streams.

To be meaningful, experimental testing conditions were designed to simulate as closely as practical those key WTP conditions that may affect mercury speciation. For vitrification and process condensate evaporator tasks, the key process conditions identified, their associated target values, and the average values achieved during testing are summarized in Table S.1. All operational constraints were successfully achieved except for the evaporator concentration factor which, nevertheless, is well within the conceptual design operating range (1.25 to 5) of vacuum evaporators.

To satisfy the technical objectives of this mercury-flowsheet-testing task as discussed above, continuous emission monitoring (CEM) for volatiles (H₂, O₂, CO, CO₂, NO, NO₂, SO₂, volatile organics, and gaseous forms of mercury) in the unquenched melter exhaust was conducted during all seven test conditions evaluated during this 5-day, 24-h-per-day, process-evaluation study. Beyond the CEM studies, four discrete U.S. Environmental Protection Agency Method-29 sampling campaigns were conducted to characterize the melter-effluent source and the chemical species of effluent mercury in particular. Secondary waste streams were also routinely sampled to further elucidate the fate and behavior of mercury and all other waste-simulant species.

	Parameter	Target	Actual			
Melter						
	Glass melt temperature, °C	1,150	1,150			
	Plenum temperature range, °C	400-450	422			
	Post film cooler temperature, C°	200-250	222			
	Melter-EVS off-gas residence time, s	1	1			
	Specific Glass Agitation Flow, scfm/m ²	1.0-2.0	0 0.99			
	Glass Production Rate, MT/d/m ²	0.4-0.8	0.64			
	Glass Fe ⁺² /Fe ^{total} ratio	≤0.3	0.055			
Evaporator		-	pH=8.4	pH=13		
	Boiler Temperature, °C	49 ± 10	47–49	48-49		
	Reflux Condenser, °C	49 ± 10	47–49	49		
	Condenser Temperature, °C	30 ± 5	30	30		
	Vacuum, Torr	60-100	71-90	67-85		
	Scaled Inleakage Rate, Scc/min	8-9	8.75	8.5		
	Concentration Factor	2	1.7 2.7			

Table S.1. C-104/AY-101 Flowsheet Evaluation Test Conditions

Because the performance of the mercury CEM was flawed during C-104/AY-101 melter-flowsheet evaluations, correlations existing between the parameters of the test (Hg, Cl, and reductant concentrations) and the melter-effluent characteristics of mercury, if any, were not directly identifiable using continuous monitor results alone. However, when the continuous monitoring information was combined with off-gas sampling results and waste-stream compositional data, an apparent correlation between feed chloride content and the melter's mercury-effluent source term was revealed. Specifically, these combined data suggest that when significant chlorine is present in a mercury-containing feed stream that is vitrified under WTP processing conditions, the formation of HgCl₂ is both thermodynamically and kinetically favored. With an Hg:Cl molar ratio of ~0.1, conversion to chemically combined mercury was essentially complete, whereas a 0.4 molar ratio produced a mixed mercury-effluent source with an appreciable elemental component. On the other hand, for test segments with feed Hg:Cl molar ratios in excess of 2.5, the elemental form was found to be the dominant melter mercury-effluent source. These results suggest that when mercury-containing melter feed with a very low Hg:Cl molar ratio is vitrified under WTP processing conditions, a chemically combined (HgCl₂) mercury-effluent source results; otherwise, depending on the magnitude of the Hg:Cl molar ratio, a mixed or totally elemental source will dominate the melter mercury source term. Mercury CEM sampling results downstream of the Ejector Venturi Scrubber (EVS) and high-efficiency mist eliminator (HEME) off-gas treatment devices were consistent with corresponding melter-source-term results previously discussed.

Although the apparent relationship between the Hg:Cl molar ratio and the melter's mercury source term observed during the current test should be generally true and independent of waste-batch identity, any waste constituent capable of affecting the chemical states of the reactants (halogens) can alter the melter's off-gas source term. Thus, if gas-phase reactions are responsible for the formation of halogenated mercury, any substance capable of forming refractory nonvolatile compounds with the halides under vitrification conditions, for example, will affect reactant availability and the resultant yield of halogenated mercury compounds. As a result, unique features of the melter feeds being processed need

to be examined in addition to Hg:Cl molar ratios before conclusions can be drawn regarding the likely composition of mercury off-gas effluents.

Post-test evaluation of the melter's secondary waste streams has shown that \sim 70% of the mercury fed to the melter penetrated the off-gas system's quench scrubber and accumulated in the HEME's deep-bed filter, suggesting a primarily elemental mercury effluent source. Indeed, 76% of the total mercury processed during Research-Scale Melter (RSM) testing occurred during the last two test segments when the melter's mercury-effluent source was predominantly in the elemental state. Of the 27% of mercury collected in the quench-scrubber's condensate, only 9% of the quench-scrubber's mercury inventory was found to be soluble.

The addition of sugar reductant had no discernable influence on processing rates, although it did measurably affect the glass-oxidation state and nitrate reduction. The major effluent gases observed during all phases of melter testing were CO_2 and NO. The combustible gas CO was barely detectable (<2 ppm), except when sugar was added to the feed, and H₂ was not detectable (<10 ppm) under any of the test-processing conditions. Throughout all phases of processing, the CO concentration (by volume) averaged only 1.7 ppm, and the maximum concentration recorded, 0.0019%, occurred during the processing of feed containing 5 g of sugar/L of feed. These concentrations are well below the lower flammability limit (15.5 vol%) of this combustible gas.

Melter partitioning of individual feed constituents, derived from both off-gas sampling and secondary waste-stream analysis, revealed that with the exception of boron, mercury, sulfur (feed impurity detected only in off-gas), and the halogens, essentially all feed constituents (excluding, for example, C, N, and H_2O) were found to be primarily in a condensed state downstream of the film cooler. Overall, the element-specific decontamination factors (DFs) recorded during RSM testing are reasonably close to general expectations and are generally consistent with previous RSM testing results. The very reasonable mass closure demonstrated for most of the feed constituents for which complete analytical data exist suggests that the current melter test has successfully characterized the C-104/AY-101 vitrification flowsheet.

Representative glass samples generated under worst-case, high-mercury testing conditions were subjected to the U.S. Environmental Protection Agency's Toxicity Characteristic Leach Procedure test (TCLP 1992) to demonstrate land-disposal suitability of the simulated waste glass produced during RSM testing. The concentrations of all hazardous analytes, except for Ba and Cr (for which estimates are provided), were found to be below instrument detection limits, and all were below their respective universal treatment standard (UTS) limits. The extremely low mercury concentrations found in the glass presents no adverse leaching problems that would preclude the glass-vitrification product from conforming with all existing Resource Conservation and Recovery Act land-disposal limits (40 CFR 268).

To project HLW/LAW, LERF/ETF, and VV Hg partitioning during pretreatment secondary-waste concentration, representative samples of the quench-scrubber's condensate/scrubbing liquor were vacuum evaporated and concentrated with and without initial pH adjustment. Of the two tests conducted, the results obtained from the pH-adjusted (13) evaluation are considered to be most representative of projected WTP evaporation conditions. The analyses of evaporator condensate and off-gas sample solutions suggest that the mercury partitioning to the Pretreatment Plant's condensate collection and VV

systems could be greater than the current WTP design value for the contract maximum mercury feed rate (Cramer 2001). Specifically, the observed evaporator mercury DF (~22) for the unblended RSM condensate is significantly less than the corresponding reference concentrator DF of 997 projected for a blended evaporator feed stream. Of the partitioned mercury, 5.4% was collected in the overhead condensate fraction with the remaining 94.6% being carried off by the gaseous exhaust (vessel vent system). The chemical nature of the evaporator's mercury off-gas source term was evaluated by observing the mercury distribution across the off-gas chemical scrubbers employed during the laboratory-scale tests. The results obtained suggest a mixed volatile oxide/elemental mercury source term (25% / 75%), dominated by the elemental form (3×). Post-test analyses of the mercury remaining in the evaporator concentrate after both the pH=8.4 and pH=13 tests demonstrated that most (86% and 93%, respectively) of the mercury present was associated with undissolved solids. If representative, this result suggests that all but 7% to 14% of the mercury present in WTP evaporator bottoms will be recycled to the HLW melter, with the remainder contributing to the LAW stream.

Since the Pretreatment Plant's evaporator influent stream will be composed of more than just the secondary, aqueous waste generated by HLW melters, the mercury-partitioning results (i.e., mercury DF) established by the above laboratory-scale evaporator tests may not be totally representative of actual plant operations. All factors that can affect the chemical composition of mercury in the evaporator's influent stream need to be considered before accurate Pretreatment Plant projections of mercury off-gas and solution concentrate behavior can be established.

Acronyms/Abbreviations

APEL	Applied Process Engineering Laboratory
BNI	Bechtel National, Inc.
CEM	continuous emissions monitor
DF	decontamination factor
DOE	U.S. Department of Energy
EPA	U.S. Environmental Protection Agency
ETF	Effluent Treatment Facility
EVS	Ejector Venturi Scrubber
Fd	feed
FIA	flame ionization analyzer
FY	Fiscal Year
GC	gas chromatograph
HEME	high-efficiency mist eliminator
HEPA	high-efficiency particulate air (filter)
Hi	High
HLW	high-level waste
Idle	melter non feeding period
IR	injection rate
LAW	low-activity waste
LERF	Liquid Effluent Retention Facility
LFCM	Liquid Fed Ceramic Melter
Lo	Low
LOD	loss on drying
LOI	loss on ignition
MACT	Maximum Achievable Control Technology
MOG	melter off-gas
Mx	maximum
NA	not applicable
ND	not detected
NDIR	non-dispersive infrared

ORP	Office of River Protection
PLC	Programmable Logic Controller
PNWD	Battelle—Pacific Northwest Division
POG	process off-gas
ppm	parts per million (by volume)
QA	quality assurance
QAPjP	quality assurance project plan
RCRA	Resource Conservation and Recovery Act
Red	reductant
RSM	Research-Scale Melter
SBS	submerged-bed scrubber
scfm	standard cubic foot per minute
SCR	silicon-controlled rectifier
SRTC	Savannah River Technology Center
ShakeDwn	melter/off-gas system operational alignment activity
StDev	standard deviation
TCD	thermal conductivity detector
TCLP	toxicity characteristic leach procedure
THC	total hydrocarbon
UDS	undissolved solids
UTS	Universal Treatment Standard
VOC	volatile organic compound
VV	Vessel Ventilation
WTP	Waste Treatment Plant
WTPSP	Waste Treatment Plant Support Project
	not applicable/available/detected/measured

Acknowledgments

The following Battelle—Pacific Northwest Division (PNWD) and Bechtel National Inc. (BNI) staff are recognized for key contributions to this vitrification flowsheet evaluation.

Task Execution

The successful completion of the recent Research-Scale Melter (RSM) flowsheet evaluation of simulated, mercury-containing Hanford waste formulation is due entirely to the combined skills and personal commitments of the test contributors. PNWD operators and support staff had to shoulder a greater than anticipated work load due to the presence and needs of commercial service providers that participated in this melter test. With only two qualified staff per shift, it was the resourcefulness and perseverance (and sweat) of the shift personnel, Bill Buchmiller, Jim Davis, Steve Halstead, Chuck Hymas, Harry Smith, and John Tixier, Jr., that allowed the testing schedule to be maintained and successfully completed in a totally safety-conscious and environmentally-responsive manner.

- Test Reporting Support
 - Wayne Cosby for dedicated editorial assistance and a willingness to work with a non-resident author during weekend and off-hour periods (PNWD).
 - Renee Russell for a thorough technical review of this report and numerous Project related documents, technical support of laboratory activities, and a willingness to guide a somewhat naive principal investigator through a maze of confusing Project procedures (PNWD).
 - o Barry Sacks for providing thorough but practical quality assurance support to this Task (PNWD).
 - May-Lin Thomas and Sally Slate for the detailed and complete analysis of a confusing array of experimental test samples (PNWD).
- Task Guidance:
 - Larry Petkus and Joe Perez for technical guidance in establishing and/or resolving experimental design conditions and/or parameters (BNI).
 - o Gordon Beeman and Juai Jao for providing Project support to this Task (PNWD).

Sum	mary			iii	
Acro	onyms	Abbrev	iations	vii	
Ack	nowle	dgments		ix	
1.0	Intro	duction.		1.1	
2.0	Test	Objectiv	/es	2.1	
3.0	0.0 Melter Test Program				
	3.1	Test-Pa	arameter-Range Selection		
		3.1.1	Mercury		
		3.1.2	Chlorine		
		3.1.3	Reductant		
		3.1.4	Test-Condition Duration		
	3.2	Test-M	atrix Glass Composition		
4.0	Expe	erimenta	l Equipment Description	4.1	
	4.1	RSM S	ystem Description	4.1	
		4.1.1	Melter	4.1	
		4.1.2	Feed System		
		4.1.3	Off-Gas Processing System		
		4.1.4	Off-Gas Sampling System	4.7	
	4.2	Data-A	cquisition and Process-Control System	4.15	
	4.3	Conder	nsate Evaporator	4.16	
		4.3.1	Evaporator Design Criteria	4.16	
		4.3.2	Normal Evaporator Operation Parameters and Resulting Testing Parameters	4.17	
		4.3.3	Laboratory-Scale Evaporator Description	4.17	
		4.3.4	Analytical Objectives	4.18	
5.0	C-10	04/AY-1	01 Simulant, Melter Feed, and Product Glass	5.1	
	5.1	C-104/	AY-101 Waste Surrogate and Feed Mixture	5.1	
	5.2	Feed M	lixture/Test Condition Description	5.5	
	5.3	Melter	Feed Characteristics	5.5	
	5.4	Produc	t Glass Characteristics	5.7	
		5.4.1	Oxidation State Results	5.8	

Contents

		5.4.2	Compositional Data	5.10
		5.4.3	TCLP Results	5.10
6.0	Proc	ess Cha	racterization	6.1
	6.1	Proces	sing Observations and Parameters	6.1
	6.2	Operat	ing Parameters	6.4
		6.2.1	Process Temperatures	6.6
		6.2.2	Process Pressures	6.9
		6.2.3	Melter Electrical Data	6.9
		6.2.4	Bubbler Glass Agitation	6.13
		6.2.5	Melter Off-Gas Temperature and Residence Time Characteristics	6.13
		6.2.6	EVS Condensate Tank, Film Cooler Injection Air	6.16
7.0	Mel	ter Off-(Gas Emission Characterization	7.1
	7.1	Flue-C	Gas Effluent	7.2
	72	Mercu	ry-Vapor Emissions	75
		721	Thermodynamic Model of Mercury Speciation	7.6
		7.2.2	CEM Mercury Measurements	7.6
	73	Conde	nsed-Phase Effluents	7 12
	1.5	7 3 1	Aerosol Mass DFs	7 13
		732	Aerosol Flemental DFs	7 14
		7.3.3	Volatile Partitioning and Total Elemental DFs	7.14
	7 /	Droces	s Wasta Stream Composition	7 20
	/.4	7 / 1	FVS Condensate Composition	7.20
		7.4.1	Overall Melter DE	7.21
		7.4.2	HEME Effluent Catch	7.23
		7.4.5	Off Gas Line Effluent Accumulations	7.24
		/.4.4	On-Oas Enic Efficient Accumulations	1.23
8.0	Вур	roducts,	Residuals, Mass Balance, and Volume Reduction	8.1
	8.1	Off-Ga	as Line Deposits	8.1
	8.2	Proces	s Mass Balance	8.2
	8.3	Vitrifi	cation Process Statistics	8.3
	8.4	Mercu	ry Partitioning Under Vacuum-Evaporator Concentration of Melter-Generated	
		Conde	nsate	8.4
		8.4.1	Experimental Observations and Results	8.4
		8.4.2	Mercury Partitioning	8.7
	8.5	Proces	s Summary	8.11

9.0	QA Requirements	. 9.1
10.0	References	10.1
Appe	endix A: Evaporator Hardware Photographs	A.1
Appe	endix B: Feed Batching and Data Sheets	.B.1
Appe	endix C: Graphical Temperature, Electrical Parameter, and Noncondensable Off-Gas Data	.C.1
Appe	endix D: Piping Hardware Photographs	D.1

Figures

Figure 4.1. Photogra	aph of the Research-Scale Melter Demonstration Unit	4.2
Figure 4.2. Schemat	tic of the Research-Scale Melter Processing System	4.3
Figure 4.3. Plan and	l Elevation View of Temporary Melter Hood Enclosure	4.4
Figure 4.4. Exterior	View of Melter Enclosure	4.5
Figure 4.5. Interior	View of Enclosed Melter	4.5
Figure 4.6. Cross Se	ection View of the Research-Scale Melter (not drawn to scale)	4.6
Figure 4.7. Elevated	Melter Feed Tank, Secondary Containment, and Load Cell Platform Scale	4.8
Figure 4.8. Melter-V	Valve-Station Configuration and Flow Logic	4.9
Figure 4.9. Gas-Dis	tribution System Supplying Continuous-Emission Monitors	4.10
Figure 4.10. Mercur	ry Monitor Flow and Analysis Schematic	4.13
Figure 4.11. EPA M	Iethod 29, Total Mercury Sampling Train	4.13
Figure 4.12. Sample	e Train Collection Module	4.14
Figure 4.13. Sample	e-Train Flow-Control Module	4.14
Figure 4.14. Melter	Off-Gas Line Sampling Port	4.15
Figure 4.15. Vacuur	n-Evaporator and Mercury-Vapor Chemical Scrubbers	4.19
Figure 6.1. Process	History of RSM Feed Batches and Average Overall Processing Rate	6.2
Figure 6.2. Processi	ng History of the First Quasi-Continuous Processing Campaign	6.2

Figure 6.3. Processing History of the Second Quasi-Continuous Processing Campaign	
Figure 6.4. C-104/AY-101 Glass-Accumulation Data	
Figure 6.5. Melter Vacuum and Off-Gas Pressure Drop Data	6.10
Figure 6.6. RSM Electrode Operating Characteristics	6.10
Figure 6.7. Flow Rate Behavior of Glass Agitation Bubblers	6.17
Figure 6.8. Unquenched Off-Gas Temperatures and Residence Times During RSM Testing	6.17
Figure 6.9. EVS Condensate Tank Temperature and Volume	6.18
Figure 7.1. Temporal Behavior of Major Process Effluent Gases	7.4
Figure 7.2. Thermodynamic Predictions of Speciation of Mercury Compounds	
Figure 7.3. Mercury-Vapor Concentration in the Unquenched Melter Exhaust	
Figure 7.4. Mercury-Vapor Concentration in the EVS Exhaust	
Figure 7.5. Mercury-Vapor Concentration in the HEME Exhaust	
Figure 7.6. Mercury CEM Response to a Step Change in Feed Composition	
Figure 7.7. EVS Condensate Anion Concentrations at the Conclusion of Sequential Test Conditions	
Figure 8.1. Melter Off-Gas Jumper Configuration and Sample Site Locations	

Tables

Table 3.1. Target Test Matrix for the June 2003 RSM Testing Campaign	
Table 3.2. Target RSM Operating Conditions	
Table 4.1. RSM Dimensions and Operational Specifications	
Table 4.2. RSM's Effluent Gas Analyzers	
Table 4.3. Typical Range of Temperature and Volume Conditions Seen by the	ne 242-A Evaporator 4.18
Table 5.1. Compositional Summary (oxide basis) of the C-104/AY-101 HLV Additives, and the Glass for Melter Tests	V Simulant, Glass 5.2
Table 5.2. Composition of Melter Feed to Produce 1 Metric Ton of Target GC-104/AY-101 HLW Simulant (20 wt% suspended solids)	lass from
Table 5.3. As Received Simulant Feed Sample Composition	
Table 5.4. Physical Properties of Baseline, C-104/AY-101 Surrogate Melter	Feed5.4
Table 5.5. Target Test Condition, Feed Additives, Concentrations, and Proce	ssing Expectations 5.5
Table 5.6. Actual RSM Testing Conditions	
Table 5.7. Physical Properties of Surrogate C-104/AY-101 Melter Feeds	
Table 5.8. Melter-Feed-Batch Composition	
Table 5.9. Soluble Anionic Feed Species	
Table 5.10. C-104/AY-101 Sugar Loadings and Glass-Oxidation States	
Table 5.11. Oxide-Composition of Melter C-104/AY-101 Glass Samples	
Table 5.12. TCLP Leachate Concentrations from C-104/AY-101 Glasses	
Table 6.1. C-104/AY-101 Feed Processing Rates and Derived Glass-Product	ion Rates 6.4
Table 6.2. Experimental C-104/AY-101 Glass-Production-Rate Data	
Table 6.3. RSM's Operating Temperature Characteristics	
Table 6.4. Melter Kiln, Pour Spout, and Canister Oven Temperatures	
Table 6.5. RSM Electrode Circuit Operating Characteristics	
Table 6.6. Specific Process Energy Requirements for C-104/AY-101 Feeds .	

Table 6.7. Operational Characteristics of Melter Kiln and Overflow Heaters	6.14
Table 6.8. Off-Gas Flow Rate, Residence Time, and Temperature Test-Condition Statistics	6.15
Table 7.1. Test-Segment Trace-Additive Summary	7.1
Table 7.2. Unquenched Melter Off-Gas Composition	7.2
Table 7.3. Actual Vs. Calculated CO _x and NO _x Process Off-Gas Concentration	7.5
Table 7.4. Volatile Mercury Concentration During RSM Test Conditions	7.8
Table 7.5. Paired Elemental and Total-Mercury-Vapor Measurements	7.11
Table 7.6. Gross Melter Aerosol Emission Characteristics	7.13
Table 7.7. Oxide Composition of Melter-Generated Aerosols and Melter Feed	7.15
Table 7.8. Normalized Oxide Composition of Melter-Generated Aerosols	7.16
Table 7.9. Elemental Melter DFs Associated With Aerosol Emissions	7.17
Table 7.10. Off-Gas Sampler Impinger Solution Composition	7.18
Table 7.11. Volatile Melter DFs from Impinger Solution Data	7.18
Table 7.12. Total Individual Elemental Melter DF Values	7.19
Table 7.13. Distribution of Mercury Effluent Across Sampler Components	
Table 7.14. Projected Elemental Mercury Distribution Across Sampler Components	7.20
Table 7.15. EVS Condensate Soluble Effluent Composition	7.22
Table 7.16. EVS Condensate Soluble-Anion Concentrations	
Table 7.17. Composition of EVS Undissolved Solids	7.24
Table 7.18. EVS Waste-Stream Effluent Distribution	7.25
Table 7.19. Off-Gas Sampler and Waste-Stream Total Melter DF Values	7.25
Table 7.20. RSM Off-Gas Pipe Deposit Composition	
Table 7.21. Pipe Solids Soluble Anion Composition	7.27
Table 7.22. Secondary Waste Stream Quantities	7.27
Table 8.1. Distribution of Melter Off-Gas Line Deposits	8.1

Table 8.2.	Stream-Dependent %-Partitioning of C-104/AY-101 Melter-Feed Constituents	8.3
Table 8.3.	Evaporator Design and Test Conditions	8.5
Table 8.4.	Mercury Partitioning Study, Mass Balance (pH=8.4)	8.6
Table 8.5.	Mercury Partitioning Study, Mass Balance (pH=13)	8.7
Table 8.6.	Evaporator Component Composition for pH 8.4 Test (mg/L)	8.8
Table 8.7.	Evaporator Component Composition for pH 13 Test (mg)	8.9

1.0 Introduction

Mercury is found in the Hanford tank farm wastes. The Best Basis Inventory^(a) suggests that tank-farm wastes contain approximately 2000 kg of mercury. A study of mercury's effect on the Waste Treatment Plant (WTP) process (Cramer 2001) suggested that vapor-state mercury is likely to interfere with catalytic off-gas treatment devices and to exceed clean-air release limits if vapor-state emissions were left unabated. However, an examination of process alternatives and flowsheet changes designed to mitigate these potential problems require an accurate knowledge of the partitioning behavior of mercury; unfortunately, the distribution of mercury compounds through the WTP is not well understood.

The chemistry of mercury under thermal processing conditions suggests that several different chemical species of mercury can be formed. The physical properties of those species provide different pathways for the mercury in the WTP, which require different abatement schemes to control mercury releases. Most of the incoming mercury should go directly to the high-level waste (HLW) melter where it will vaporize and become part of the melter off-gas. The chemical nature of the mercury effluent as it enters the WTP's off-gas quencher, the submerged-bed scrubber (SBS), will establish how it partitions and affects unit off-gas treatment operations and secondary waste streams in the WTP.

Under current Plant operating conditions, the portion of the mercury that is scrubbed from the process exhaust will be periodically recycled to the Pretreatment Plant where it will be chemically adjusted, concentrated, and further partitioned between HLW, low-activity waste (LAW), the Liquid Effluent Retention Facility/Effluent Treatment Facility (LERF/ETF), and the pretreatment's vessel vent (VV) streams. On the other hand, the noncondensable mercury fraction will penetrate the WTP's primary offgas treatment system and, depending on its magnitude, can threaten vitrification plant catalytic unit operations and clean-air emission limits for mercury as well as other regulated volatiles for which catalytic abatement processes are used. As a result, activated carbon absorbers have been subsequently added to the melter off-gas system designs to protect the catalytic units and reduce mercury emissions to below the Maximum Achievable Control Technology (MACT) standards. However, mercury-vapor speciation and corresponding influent rates are not well understood. In addition, there is a possibility that the mercury content of the LERF/ETF streams, discussed above, could exceed those facility's discharge limits. Consequently, partitioning fractions and chemical species produced by vitrification and subsequent pretreatment operations need to be quantified to determine off-gas design criteria that will mitigate the threat of excessive airborne and aqueous plant releases.

This summary report documents the experimental evaluations that were expressly conducted to elucidate the vitrification flowsheet behavior of a mercury-containing Hanford-waste simulant under representative WTP operating conditions using a range of mercury, halogen, and reductant concentrations as testing variables. Specifically, Battelle—Pacific Northwest Division's (PNWD's) Research-Scale Melter (RSM) was used to conduct seven separate processing campaigns over an ~120-h period that involved selected combinations of three mercury, two halogen, and two reductant feed concentrations. In addition, the secondary aqueous waste stream generated by the melter's quench-scrubber was used in a separate, post-

⁽a) Go to the following link for the Best Basis Summary database: <u>http://twins/data/getLookupFields3.exe?table=tcd.dbo.v_best_basis_summary&whatsnew=Best+Basis+Inventory</u>. This page has all 177 tanks across the top, and down the left are all of the best basis standard constituents, which would include mercury, in kilograms. The total, 1840 Kg, is shown at the right.

melter-test evaporator study to project mercury partitioning in LERF/ETF streams. The results of these experimental efforts will now be discussed.

2.0 Test Objectives

The major objectives of this test were to characterize the off-gas behavior of mercury under representative WTP vitrification processing conditions over selected waste-loading ranges of mercury, halogen, and reductant and to project mercury-partitioning characteristics during simulated secondary-waste pretreatment concentration operations using condensate generated during the vitrification test conditions. The activities conducted to accomplish these objectives included:

- 1. Designing experimental testing conditions to simulate as closely as practical those key WTP conditions that may affect mercury speciation. For vitrification testing, the key process conditions identified and their associated target values include:
 - a glass temperature of 1150°C
 - forced-air glass agitation at 1 to 2 scfm/m², using two bubblers
 - a plenum temperature of 400°C under steady-state processing
 - a post film-cooler temperature of 200°C to 260°C
 - a melter/quench-scrubber off-gas transit time of 1 second
 - a steady-state specific glass production rate between 0.4 and 0.8 MT/d/m²
 - a glass with $Fe^{+2}/Fe^{tot} \le 0.3$.

For the laboratory-scale mercury evaporator partitioning studies, the key evaporator operational parameters selected included:

- a vacuum of 60 to 100 Torr
- a 49°C evaporator feed temperature
- a 30°C condenser temperature
- a scaled 8 to 9 Scc/min simulated inleakage rate
- a feed pH of 8.4 and 13^(a)
- a feed concentration factor of 2.
- 2. Performing a research-scale melter test to sample and characterize mercury speciation in off-gas and condensate streams over a range of vitrification process conditions involving feed concentrations of mercury, chlorine, and reductant. This off-gas characterization objective was accomplished by:
 - conducting quasi-continuous determinations of vapor-state concentrations of elemental and oxidized (e.g., HgCl₂) forms of mercury before and after each off-gas processing device
 - taking periodic secondary waste-stream supernatant samples for subsequent off-line analyses

⁽a) Since Site evaporators utilize a pH range of 7 to 10, but pH=13 is considered WTP prototypic, two evaporator tests were conducted at pH=8.4 and at pH=13.

- collecting and analyzing all quench-scrubber undissolved solids (UDS) and supernatant fluids at the conclusion of testing
- characterizing the melter-effluent source term using a mercury-specific (40 CFR 60, Appendix A, Method 29) sampling train.
- 3. Sampling and characterizing mercury speciation in evaporator condensable and noncondensable streams when concentrating mercury-containing condensate solutions generated during the vitrification testing phase. Projecting partitioning behavior of mercury under simulated pretreatment evaporation operations was accomplished by:
 - collecting a representative sample of the melter's quench-scrubber waste stream to be concentrated
 - making appropriate pH and chemical adjustment to the condensate fraction
 - performing concentration under the above evaporator reference conditions
 - collecting condensate fraction
 - scrubbing the noncondensable condenser exhaust stream with sequential chemical traps (H₂O₂/HNO₃ and KMnO₄/H₂SO₄) designed to selectively trap oxidized and elemental forms of mercury, respectively
 - determining the mercury content of the evaporator condensate and chemical-trap solutions.

In addition to and in support of the above-discussed primary objectives, the compositions and masses of all process influent and effluent streams were also characterized, allowing glass leachability to be assessed and a melter mass balance to be derived. To accomplish this:

- Continuous emission monitors were used to monitor melter noncondensable flue-gas emissions
- Samples of feed, glass, and all aqueous streams generated from off-gas treatment devices were periodically collected in support of post-test analyses and subsequent flowsheet characterization.

3.0 Melter Test Program

As discussed in Section 2, waste-composition ranges of mercury, chlorine, and reductant were chosen as test parameters in the vitrification flowsheet evaluations. High (Hi) and low (Lo) concentration values of each of these variables were established on the basis on waste-tank compositions and likely staging strategies. In addition to Hi/Lo values, an engineering maximum (Mx) condition for mercury was also evaluated.

To satisfy the technical objectives of this mercury-flowsheet-testing task, seven test conditions were identified for evaluation requiring \sim 5 days of continuous, 24-h-per-day melter operation. A target test-condition matrix defining the technical activity schedule is shown in Table 3.1. This matrix is designed to satisfy the test objectives described in Section 2.

Test Condition Levels (Hg/Cl/Redox)	Glass Fe ⁺² :Fe ^{tot} Ratio	Reductant Additive	Feed Rate Target	Cl-Fd Oxide Wt%	Hg Oxide Wt%	Off-Gas Samples	Duration	Cumu- lative hours ^(a)
#1 Lo/Lo/Lo	Not	None	Highest	0.009			18	18
#2 Lo/Hi/Lo	(ND)	None	optimum at 90%	0.009	0.05		18	36
#3 Lo/Hi/Hi	≥20% ≤30%	Sucrose ~5 g/L	to 95% Cold-	0.06			6	42
#4 Hi/Lo/Lo	ND	None	Cap Coverage	0.009	0.15	Х	18	60
#5 Hi/Hi/Lo	Not	None	0.4-0.8		0.15	Х	18	78
#6 Mx/Hi/Lo	Detectable	INDIR	MT/d/m ²	0.06	0.05	Х	18	96
#7 Mx/Hi/Hi	≥20% ≤30%	Sucrose ~5 g/L	1.0–1.5 L/h		0.75	X	6	102
(a) The 18 hours of contingency time allows for startup, feed-adjustment activities, and unscheduled delays.								

Table 3.1. Target Test Matrix for the June 2003 RSM Testing Campaign

3.1 Test-Parameter-Range Selection

To establish a solid basis for the Hg and Cl concentration ranges to be used during RSM testing, an evaluation was conducted of the 84 waste batches that have been identified under the River Protection Project's System Plan Case-3 that will allow completion of all tank-waste vitrification activities by 2028 (ORP 2003). The results of this evaluation are discussed below.

3.1.1 Mercury

It was determined that, except for the beginning and ending phases of processing, most waste batches exhibit a fairly consistent HgO content. The average HgO waste content, 0.19 wt%, would correspond, at 30 wt% waste loading, to a feed concentration of \sim 0.05 wt%, which is an analytically reasonable choice for the low-Hg-concentration parameter. On the other hand, the highest batch Hg concentration, 0.5 wt%, corresponding to a feed concentration of 0.15 wt%, was a logical choice for the high concentration value to be used during RSM flowsheet tests.

In addition to the Hi/Lo mercury conditions described above, a design maximum condition (Mx) was also established on the basis of creating a sufficient melter mercury off-gas flux to allow mercury-vapor-saturation conditions $(30^{\circ}C)$ to exist at the outlet of the HEME for:

- a 0.8 MT/d/m² glass-production rate
- quencher and high-efficiency mist eliminator (HEME) decontamination factors (DFs) of 6 and 3.5, respectively.

The equivalent oxide concentration of mercury in the melter feed stream that satisfies these conditions at a projected noncondensable off-gas flow rate of 5 scfm is 0.95 wt%, which became the adopted mercury Mx value.

3.1.2 Chlorine

To determine the Hi/Lo candidate values for chlorine, waste-batch-composition data for chlorine were similarly examined. It is clear from this evaluation that the median value of 0.03 wt% is representative of the chlorine content of most of the Case-3 batches (ORP 2003) and, therefore, appeared suitable as a low-value test parameter. If the Case-3 batch maximum for chlorine of 0.19 wt% is used for the upper bound, a suitable Cl:Hg molar ratio range of 7 to 0.5 will be established. Since this is also representative of the span of molar ratios exhibited by Case-3 batches, it also appeared appropriate to choose 0.06 wt% and 0.009 wt% as the Hi/Lo chlorine values for melter-feed-test parameters.

3.1.3 Reductant

The glass redox state, which is measured by the Fe^{+2} : Fe^{Tot} ratio, is determined by the relative concentrations of oxidizing and reducing agents in the melter feed stream. Oxalate reductant in the baseline feed (see Section 5) is not expected to create a Fe^{+2} : Fe^{Tot} ratio above its detection limit of about 0.01. However, since reductant levels can be varied to improve process throughput, it is important to determine the influence of reductant levels, if any, upon the off-gas distribution Hg chemical species. Consequently, during part of the test, sucrose was added to the melter-feed stream in sufficient quantities (~5g/L) to produce a measurable Fe^{+2} : Fe^{Tot} ratio that is considered safe (≤ 0.3) from a glass-durability standpoint.

3.1.4 Test-Condition Duration

The ability of the Table 3.1 test series to accomplish all planned test objectives depends on how rapidly the melter-glass bath, the cold-cap, and the off-gas respond to step changes that occur for each of the

different test conditions. The target operating conditions based on RSM design parameters and melter test specification conditions are summarized in Table 3.2.

Parameter	Target	
Melt surface area, cm ²	182	
Melt volume, L	1.4	
Glass specific gravity (20°C)	2.6	
Glass inventory, kg	3.6	
Minimum expected glass rate, MT/d/m ²	0.40	
Minimum expected glass rate, kg/h	0.30	
Maximum expected glass rate, MT/d/m ²	0.80	
Maximum expected glass rate, kg/h	0.61	
Minimum expected feed rate, L/h ^(a)	0.55	
Minimum expected feed rate, kg/h ^(b)	0.77	
Maximum expected feed rate, L/h ^(a)	1.11	
Maximum expected feed rate, kg/h ^(b)	1.55	
Maximum melter glass turnover rate, h	12.0	
Minimum melter glass turnover rate, h	6.0	
Glass Fe ⁺² :Fe ^{tot} ratio	≤0.3	
Specific glass agitation flow, scfm/m ²	1.0 to 2.0	
Glass agitation flow rate, sL/min	0.52 to 1.0	
Glass melt temperature, °C	1,150	
Plenum temperature range, °C	400 to 500	
Plenum pressure, inches water	-0.5 to -1.5	
Post film cooler temperature range, °C	200 to 250	
Air inleakage rate, scfm	1	
Post Film-cooler off-gas flow rate, acfm	10.5	
Melter-EVS off-gas residence time, s	1	
Initial scrub solution volume, L	60	
Initial scrub solution pH	7	
(a) Based on a feed oxide loading of 550 g-ox/L.		
(b) Based on a feed density of 1.4 kg/L.		

Table 3.2. Target RSM Operating Conditions

At a given set of operating conditions, some operating time is needed to allow time for the melt-bath composition to approach a new equilibrium after step changes in the feed composition that affect the melt composition. Based on the RSM molten-glass volume and the acceptable production-rate-range

previously discussed, a 6- to 12-h melt-cavity turnover frequency is suggested. Since up to three bathvolume turnovers are needed to achieve steady-state composition of the melter's glass inventory, a minimum of 18 processing hours, at a 0.8 MT/d/m^2 production rate, will be needed to reach true steadystate processing conditions after a feed-stream change has occurred.

However, the above considerations only pertain to feed constituents that are incorporated in the glass product. Because mercury glass partitioning is vanishingly small, changes in its feed concentration will have no impact upon steady-state melter-glass composition. Similarly, since changes in feed-reductant concentrations affect existing as well as newly produced melter glass, changes in melter-glass oxidation state occur over much shorter periods (3 to 6 hours) than the time required to produce three melter turnovers. Although changes in chloride concentrations will affect melter-glass composition and impact the time required to attain steady-state processing conditions, the impact of glass composition, if any, upon the melter's mercury emission source will be derivable from the trending data generated by the quasi-continuous measurements that will be conducted.

Thirty hours were originally allowed for the first test condition to provide sufficient time to:

- establish processing stability
- purge the melter of its startup glass
- record emission results under representative processing conditions.

However, since it was possible to replace the RSM's existing glass inventory with representative C-104/AY-101 glass supplied by Bechtel National, Inc. (BNI), the period for the first test condition was reduced to 18 hours.

Six hours has been allowed for establishing the impact of the changes in reductant concentrations. As discussed above, this time is not based on melter-glass turnovers, but is nominally based on time needed to:

- make the feed change
- change the oxidation state of melter glass
- characterize the system response to the change
- accumulate adequate continuous off-gas monitoring data.

All other test conditions are scheduled for 18-h periods.

3.2 Test-Matrix Glass Composition

The test-parameter conditions detailed in Table 3.1 will be established by adding appropriate quantities of mercury, chlorine, and/or sucrose to an invariant HLW simulant feed batch representing \sim 30 wt% C-104/AY-101 feed formulation to be discussed in Section 5. Since mercury and reductant are not incorporated in the glass to any significant extent, and the chlorine content of the glass remains at trace levels under all conditions, the glass composition is expected to remain nominally invariant throughout all phases of testing.

4.0 Experimental Equipment Description

Experimental equipment used to support the objectives of this flowsheet evaluation task includes the RSM processing system and a laboratory-scale evaporator. The RSM facility was used to model WTP's HLW processing conditions to determine the chemistry and resultant off-gas fate and behavior of waste-constituent mercury. The redistribution of mercury in the RSM's secondary waste between HLW, LAW, LERF/ETF, and off-gas streams was subsequently evaluated under WTP pre-treatment (concentration) conditions using a laboratory-scale evaporator. The equipment to be used in these evaluations is described below.

4.1 RSM System Description

PNWD's RSM facility is located in the Applied Process Engineering Laboratory (APEL) building in Richland, Washington. Figure 4.1 is a photograph of the RSM system as it nominally appeared during melter testing, and Figure 4.2 schematically illustrates the system components and their relationships to one another. Because mercury was part of the C-104/AY-101 vitrification flowsheet evaluation test, a temporary enclosure was constructed around the RSM to contain and exhaust any melter emissions created by inadvertent melter pressurizations. A plan and elevation view of this enclosure is illustrated in Figure 4.3, while interior and exterior photographs of a similar, previously constructed walk-in hood appear in Figure 4.4 and Figure 4.5, respectively.

The RSM processing system provides a continuous, Joule-heated vitrification capability, which is key for

- developing process flowsheets
- characterizing relationships between feed composition and the properties of the final glass produced
- establishing the fate and behavior of process effluent.

This melter system's capability to produce glass in a continuous manner is also essential for modeling the behavior of a full-scale system. Moreover, the size of the RSM allows the impacts of process variables upon melter performance or glass quality to be quickly and efficiently evaluated without undue expense or waste generation.

4.1.1 Melter

The RSM itself is a small Joule-heated melter that is capable of processing melter feed on a continuous basis. The body of the RSM is an Inconel[®] closed-ended cylinder lined with Alfrax® refractory and containing a Monofrax® K3 refractory melt cavity. An Inconel[®] overflow tube discharges molten glass into a stainless steel canister. An electric kiln surrounds the melter body and minimizes heat loss from the melter body during operation, and auxiliary heaters are used to heat the melter's discharge section to facilitate pouring of the glass. The stainless steel glass receipt canister sits inside a clam-shell furnace maintained between 700°C and 900°C to promote uniform canister filling. A platform scale, forming part of a hydraulic jack assembly that supports the canister inside the furnace, allows glass-canister accumulations to be monitored as necessary. Two top-entering Inconel[®] 690 electrodes (7.6-cm square × 0.64-cm thick [3-in. square × ¼-in. thick]), that are suspended in the glass, supply Joule-heating power to the RSM. The electrode's connecting tubular busbars also serve as thermowells that allow continuous

measurement of the glass-pool temperatures. Beyond the pair of electrodes and the exhaust port, the melter lid also provides melter access for a pair of glass agitation bubbler (Inconel[®]) tubes and a water-cooled feed nozzle. Figure 4.6 provides a cross-sectional view of the melter vessel, illustrating its refractory makeup, while Table 4.1 summarizes the RSM's dimensions and other operational features.



Figure 4.1. Photograph of the Research-Scale Melter Demonstration Unit



Figure 4.2. Schematic of the Research-Scale Melter Processing System



Figure 4.3. Plan and Elevation View of Temporary Melter Hood Enclosure

4.1.2 Feed System

The melter-feed system is located on the elevated steel platform adjacent to the melter (see Figure 4.1). Two agitated conical bottom tanks were staged on this platform during the current test. A master (55 gal) tank accommodated the baseline feed formulation that was periodically transferred to the melter's 15-gal feed tank where it was mixed with appropriate spike additives (Hg, NaCl, and/or sugar) to create the seven feed conditions that formed the basis of the experimental studies of the current test. The melter-feed tanks, the variable-speed agitators, the peristaltic-feed pump, and the valve-control station are attached to a steel pallet that allows the melter-feed-tank system, which includes a secondary containment vessel and load-cell platform scales, to be lifted from the platform with a forklift. The electronic record of time-dependent feed-tank weights, generated by the system's load-cell platform scales, provides for redundant feed-transfer measurements and a direct measure of the melter mass-feeding rate. Figure 4.7 provides a photographic perspective of this elevated tank and feed-delivery system.





Figure 4.4. Exterior View of Melter Enclosure

Figure 4.5. Interior View of Enclosed Melter

Two peristaltic pumps were used to extract and deliver feed from the 15-gal tank to the melter. As schematically illustrated in Figure 4.8, a large pump was used to recirculate feed from the bottom to the top of the conical, bottom-drain tank, while a smaller pump was used to extract a slip stream from the larger recirculation line and deliver it to the melter. A valve-control station associated with the melter feed delivery system allowed feed to be either sampled or delivered to the water-cooled feed nozzle that extended through the melter lid into the melter's plenum. The valve station, illustrated in Figure 4.8, also permits feed lines to be flushed with air and/or water without resorting to disassembly. A computer/pump interface allowed the pump's feeding rate to be controlled remotely, thus facilitating necessary adjustments required to maintain a steady melting process.

4.1.3 Off-Gas Processing System

Melter off-gas is treated by an off-gas treatment system consisting of a film cooler, Ejector Venturi Scrubber (EVS), HEME, and high efficiency particulate air (HEPA) filter (see Figure 4.2). The film cooler, located at the melter's exhaust port, injects room-temperature building air into the off-gas pipe to

- cool and solidify entrained vitreous matter to minimize pipe-wall particle adhesion
- speed aerosol transport to the EVS quench scrubber to minimize aerosol-settling losses in horizontal off-gas line (2 in.) pipe runs.



Figure 4.6. Cross Section View of the Research-Scale Melter (not drawn to scale)

The EVS used a high-pressure aqueous scrubbing liquor (condensate) spray to contact the process exhaust stream to quench it and to remove steam, large-diameter aerosols, and some condensable and/or acid gases. A 90-L (24-gal) charge of water was put in the scrubbing liquor/condensate collection tank at the start of the test. Off-gas condensate supernatant samples were collected at the conclusion of each discrete test segment while the UDS that accumulated within the condensate tank throughout the duration of the test were fully collected and sampled at the end of the test. A water-cooled heat exchanger located in the EVS's spray circuit was used to maintain nominal room-temperature scrubbing-liquor conditions.

Parameter	Value
Melter cavity diameter	15 cm
Melter cavity height	17 cm
Melter inside volume	4.5 L
Glass pool surface area	182 cm^2
Nominal glass depth	7.6 cm
Melter glass inventory volume	1.4 L
Nominal molten glass mass	3.6 kg
Glass turnover rate @ nominal feed rate	4.5 h
of 1.5 L/h of feed with 0.6 kg/L oxides	
Maximum operating temperature	1,200°C
Nominal operating temperature	1,150°C
Electrode Dimensions	7.6 cm × 7.6 cm
Electrode Material	Inconel [®] 690
Electrode melt-cavity bottom clearance	0 cm
Electrode current (average)	90 A
Electrode voltage (average)	25 V
Electrode current density (average/maximum)	$1.6/2.0 \text{ A/cm}^2$

 Table 4.1. RSM Dimensions and Operational Specifications

The HEME uses a deep, regenerable fibrous bed to remove both liquid aerosols generated by the highpressure EVS spray and submicron condensed-phase aerosols that successfully penetrate the lowefficiency quench (EVS) scrubber. The demisted and relatively clean HEME exhaust is then heated before being treated with a certified HEPA filter to remove all significant remaining concentrations of aerosol matter before the process exhaust is released to the environment.

Since the RSM off-gas processing system provides unit off-gas treatment operations of quenching, wet scrubbing, and high-efficiency filtration similar/equivalent to that provided by the WTP's primary melter off-gas system, comprehensive effluent partitioning behavior can be directly assessed under representative conditions. It should be noted that the aqueous quench-scrubber employed in the current test (EVS) has been previously shown to be functionally equivalent to the SBS technology (Goles and Schmidt 1992) that will be used in the WTP.

4.1.4 Off-Gas Sampling System

Process off-gas sampling during the current test was limited to characterizing the melter source term and establishing the Hg-emission-abatement performance of the melter's aqueous quench scrubber and HEME as a function of processing conditions, i.e., mercury, chlorine, and reductant concentrations. These data were collected in support of engineering flowsheet design and are not intended to support

WTP regulatory or environmental-release modeling activities. Melter exhaust gases and effluents with significant room-temperature vapor pressures were monitored continuously with gas analyzers described in Table 4.2, while four limited discrete sampling campaigns were conducted to characterize condensed-phase effluents and condensable/scrubbable gases. These sampling data will also provide a cross check for contemporaneous mercury monitoring results.



Figure 4.7. Elevated Melter Feed Tank, Secondary Containment, and Load Cell Platform Scale

4.1.4.1 Continuous Emissions Monitoring

Continuous-emission monitoring was conducted for flue gases (H₂, O₂, N₂, CO, CO₂, NO, NO₂ and total hydrocarbons [THCs]) and volatile forms of mercury. Although the continuous-emission monitors (CEMs) were configured as shown in Figure 4.9, flue-gas monitoring was exclusively conducted between the film-cooler and quench scrubber. Mercury monitoring was also selectively conducted between the EVS and the HEME and downstream of the HEME, independent of the flue-gas monitoring system.

Analyzer/EPA Method	Targeted Effluent Gases
Gas Chromatograph	H_2 & He
Oxygen/3A	O ₂
Carbon Monoxide/10	СО
Carbon Dioxide/3A	CO ₂
Nitrogen Oxide/7E	NO & NO ₂
Sulfur Oxide/6C	SO ₂
Total Hydrocarbon/25A	Volatile Hydrocarbons
Gas Phase Mercury Analyzer	Hg (oxidized and elemental)

Table 4.2. RSM's Effluent Gas Analyzers



Figure 4.8. Melter-Valve-Station Configuration and Flow Logic
Testing and analysis guidelines used in the performance of this continuous monitoring activity are presented in the July 1, 2002, edition of the U.S. Environmental Protection Agency (EPA) document Title 40, Code of Federal Regulations, Part 60 (40 CFR 60), Appendix A. Details for the instrument-specific methods employed are described below.

Oxygen and Carbon Dioxide: Method 3A was performed to determine the concentrations of oxygen (O_2) and carbon dioxide (CO_2) . O_2 was measured using a paramagnetic analyzer. CO_2 was measured using a non-dispersive infrared (NDIR) analyzer. Gas measurements were recorded once every 10 seconds on a continuous basis during each emission test period and were averaged into 1-min readings.



Figure 4.9. Gas-Distribution System Supplying Continuous-Emission Monitors

The instruments were calibrated daily using EPA Protocol One certified gas. A 3-point calibration error check of each analyzer was performed before commencing testing.

Nitrogen Oxides: Method 7E was performed to quantify emissions of speciated nitrogen oxides (NO and NO_2) using a chemiluminescent analyzer. The NO_x concentrations were recorded in dry ppm once per minute using a data-acquisition system. The analyzer was calibration bias and drift checked using EPA Protocol One certified gas daily, and a 3-point calibration error check was performed before commencing testing. The nitrogen dioxide (NO_2) to nitrogen oxide (NO) conversion-efficiency test discussed in Section 5.6 of Method 20 was performed onsite before beginning the test sequence.

Carbon Monoxide: EPA Method 10 was performed to quantify emissions of carbon monoxide (CO) using a gas-filter correlation NDIR analyzer. CO concentrations were recorded in dry parts per million (ppm) at least once per minute using a data-acquisition system, and averaged. EPA Method 10 testing was conducted with the same rigorous bias and drift requirements as found in EPA Method 6C. Before testing, an analyzer calibration error check was performed using zero-, mid-, and high-range EPA Protocol One calibration gases. A system calibration was performed daily.

Total Gaseous Organic Concentration: Method 25A was performed to quantify emissions of volatile organic compounds (VOC) measured as total gaseous organic concentrations, as propane, using a flame ionization analyzer (FIA). For the Method 25A, a heated sample line introduced gas to the FIA analyzer on a hot, wet basis. VOC concentrations were recorded in wet ppm at least once per minute, using a data-acquisition system, and averaged. Before testing, an analyzer calibration error test was performed using zero-, low-, mid-, and high-range EPA Protocol One calibration gases. The analyzer was calibrated daily.

Hydrogen and Helium: A gas chromatograph (GC) was used to quasi-continuously determine the concentration of hydrogen and helium in the off gas. This was accomplished by directly injecting the off gas into an onsite portable GC equipped with a thermal conductivity detector (TCD) and a set of micro columns designed to separate and analyze hydrogen and helium. Results were generated on a quasi-continuous basis at approximate 5-min intervals.

Continuous Mercury Monitoring: The mercury analyzer employed is a PS Analytical Sir Gallahad instrument designed to selectively measure elemental (Hg^o) and total vapor concentrations of mercury (Hg^T). The chemically combined fraction, referred to as oxidized mercury, is derived by difference. This unit operates semi-continuously in 5-min cycles, alternating between the Hg^T and the Hg^o modes. In each cycle, a 1-min-average data point for either Hg^T or Hg^o is determined.

In operation, sample gas is withdrawn continuously through a heated sample line by the CEM's heated vacuum pump. The sample gas is diluted, if appropriate, and split evenly into two streams. One stream is continuously passed through an impinger containing $2 \text{ w/v} \% \text{ SnCl}_2/5 \text{ w/v} \%$ NaOH, where oxidized forms of mercury are reduced to Hg°. The other stream flows continuously through an impinger containing 5 w/v % NaOH, where oxidized forms of Hg are scrubbed from the gas into the impinger solution.

In the first cycle, the sample gas from the $SnCl_2/NaOH$ impinger is passed into a cold-wall condenser to remove water vapor and through a bed of gold-coated sand where the total mercury as Hg° is sorbed. This is the collection cycle for the Hg^{T} measurement. The sorbent is then heated and purged with N_2 to release the total Hg into an atomic-fluorescence measurement cell. In this cycle, all chemically combined forms of Hg in the sample gas are reduced to Hg° in the $SnCl_2/NaOH$ impinger and detected and measured along with any elemental mercury that may have been originally present.

In the next cycle, the sample gas from the NaOH impinger is passed through the cold-wall condenser and into the gold-coated sand adsorber. Since the NaOH impinger removes oxidized Hg species from the sample gas, only Hg^o is collected in the gold sorbent. When thermally desorbed into the measurement cell, only Hg^o is detected.

The instrument alternates between measuring total and elemental mercury and displays the data in $\mu g/m^3$ in the form of a trend graph and stores the data in an Excel file. During every 10-min period, the instrument reports a single 1-min-average total Hg value, a single 1-min-average Hg^o value, and total oxidized Hg, determined by the difference between the two other values. The two 1-min-average values are measured 5 minutes apart, so the reported value for oxidized Hg is not a true difference of simultaneous Hg^T and Hg^o values. A diagram of the instrument's flow and measurement system is illustrated in Figure 4.10.

The mercury analyzer is calibrated using a mercury-vapor injection system supplied by the vendor. This system consisted of a vessel containing liquid mercury from which measured volumes of mercury-saturated air at a measured temperature are withdrawn by hypodermic syringe and injected into an N_2 carrier gas by means of a sample port. The analyzer software calculates the mass of mercury injected, which is converted to a mercury concentration by using both the measured carrier-gas flow rate and the time interval used for sampling.

Prior to sampling the melter off-gas, room air was sampled by the mercury analyzer to set sampling flow rates and establish a baseline. A manual data sheet was used for recording sampling flow rates, condenser temperatures, and observations and comments.

4.1.4.2 Manual Off-Gas Sampling and Analysis

Manual off-gas sampling and analysis were performed to provide off-gas emissions data in addition to that obtained from the CEMS. All manual off-gas sampling was performed at the film-cooler outlet location in order to best characterize the melter source using an EPA Method 29 (40 CFR 60, Appendix A) sampling protocol. The actual system used, illustrated in Figure 4.11, did not employ the standard pitot-tube-equipped stack probe. Rather, helium-dilution flow measurements (see below) were used in conjunction with process line and sampling-tube geometrical factors to establish isokinetic sampling flow rates. This off-gas sampling system was composed of an appropriately sized quartz sampling probe, a heated aerosol collection device, a condenser to remove condensable vapors, and a series arrangement of chemical gas scrubbers used to collect reactive non-condensable gases. Photographs of similar sample-train collection and flow-control equipment to that used during the current test are shown in Figure 4.12 and Figure 4.13, respectively. Since all gas scrubbers were usually immersed in an ice-bath container, the first (empty) scrubber vessel also served as the system's condenser.

Since species-specific mercury-vapor scrubbing was of primary interest in this test, the gas-scrubbing components used consisted of two impingers containing a mixture of 10% hydrogen peroxide (H_2O_2) and 5% nitric acid (HNO_3) and an empty impinger followed by two additional impingers containing 4% potassium permanganate ($KMnO_4$) and 10% sulfuric acid (H_2SO_4). In this arrangement, the first two impingers will selectively remove oxidized forms of mercury while the final two impingers are designed to oxidize and trap elemental vapors penetrating the first two non-oxidizing gas scrubbers. Operational



Figure 4.10. Mercury Monitor Flow and Analysis Schematic



Figure 4.11. EPA Method 29, Total Mercury Sampling Train



Figure 4.12. Sample Train Collection Module



Figure 4.13. Sample-Train Flow-Control Module

conditions allowed an ~120-min Modified Method 29 (40 CFR 60, Appendix A) test to be conducted during four testing segments.

In operation, the Method 29 sampling probe, a straight quartz tube, was inserted coaxially along the centerline of the off-gas pipe into the process off-gas line at an elbow between the film cooler and the EVS (see Figure 4.14). The diameter of the beveled sampling-probe inlet was chosen to allow isokinetic sampling conditions to be achieved with reasonable sampling flow rates. The heated filter assembly employed a quartz-filter media to quantitatively collect particulate matter entrained in the unquenched (225°C), post film-cooler melter exhaust. The filtered gas stream was then subsequently quenched (0°C) and chemically washed to remove reactive gases (e.g., Hg) by a series arrangement of several gas-washing vessels.

To establish isokinetic sampling conditions, the total off-gas flow rate has to be measured. During RSM testing, this was accomplished by injecting a helium tracer into the film-cooler's air-injection stream at a fixed flow rate (1 L/min) and measuring its resultant off-gas concentration with the online gas chromatograph discussed above. The relationship between flow rate (Flw), He injection rate (IR), and resultant He concentration (He_{ppm}) can be expressed as follows: Flw = IR×10⁶/He_{ppm}. Since GC-derived process flow rates are on a dry basis (see Figure 4.9), average steady-state feeding rates and the water content of the feed had to be used to estimate flow rates on a wet basis. Having calculated a total off-gas flow rate, the fraction of the total flow required to establish isokinetic sampling conditions was

determined by the ratio of the geometrical cross-sectional areas of the sampling probe to that of the offgas line where the sampling was being conducted. These process off-gas (POG) flow-rate data were also of fundamental importance in establishing effluent concentrations and emission rates.



Figure 4.14. Melter Off-Gas Line Sampling Port

4.2 Data-Acquisition and Process-Control System

The RSM is controlled and monitored with a Square D, SY/MAX® 400 Programmable Logic Controller (PLC). Operators interface with the PLC using a PC running FIX DMACS® software on a Microsoft NT platform that is serially linked to the PLC. FIX32 provides user-control inputs as well as history logging of the RSM system-process variables.

This data-acquisition and control system monitors and controls the electrodes, the melter and discharge canister kilns, the heater for the discharge section, and the peristaltic pump for the feed system. Data collected include the voltage and current values for major electrical components, temperature at various locations in the system (e.g., molten glass, plenum space in melter, melter kiln, and off-gas treatment system), pressures in the melter and across all off-gas system components, and the weight of the feed tank. Data are typically archived every minute, but are displayed at more frequent intervals to assist the operators.

4.3 Condensate Evaporator

Under WTP operations, aqueous secondary waste from HLW processing will be transferred to the pretreatment facility where it will be concentrated. Since the species of mercury present in this waste stream may influence the manner in which mercury partitions to HLW, LAW, LERF/ETF, and off-gas streams, quench-scrubber condensate generated during RSM testing was concentrated using a lab-scale evaporator that was integrated with the gas-scrubber components of the Method 29 (40 CFR 60, Appendix A) sampler discussed in the previous section. The design criteria, mode of operation, lab-scale system description, and analytical objectives of this study are discussed below.

4.3.1 Evaporator Design Criteria

The operating parameters to be subsequently discussed were based on information from design specifications and current working-evaporator operations detailed in the following documents:

- Engineering Specifications for the Forced Circulation Vacuum Evaporator System: DIM No. 24590-PTF-3PI-MEVV-00001, Rev A.^(a)
- 2. Process Data for Waste Feed Evaporator, Feed Vessels, and Feed/Concentration Pumps: Calculation Sheet – Calc No. 24590-PTF-MEC-FEP-00001, Rev B.
- 3. 242-A Evaporator Documented Safety Analysis, Chapter 2 Facility Description.^(b)

In addition, previous WTP-sponsored small-scale evaporation tests documented in the reports listed below were examined to ensure consistency of the current test with previous studies.

- 1. Waste Feed Evaporation: Physical Properties and Solubility Determination (U), Savannah River Technology Center, SRT-RPP-2003-00094, Rev 0.^(c)
- 2. AN-107 (C) Simulant Bench-Scale Law Evaporation with Organic Regulatory Analysis, SRT-RPP-2000-00047, Savannah River Technology Center.^(d)

⁽a) Issued by the River Protection Project - Waste Treatment Plant Project, Richland, WA (June 4, 2002).

⁽b) Waste Management Project, Hanford Site, Rev. 0, Richland, WA (April 10, 2003).

⁽c) Westinghouse Savannah River Company, Savannah River Site, Aiken, SC (May 13, 2003).

⁽d) Westinghouse Savannah River Company, Savannah River Site, Aiken, SC.

4.3.2 Normal Evaporator Operation Parameters and Resulting Testing Parameters

The evaporators currently used on the Hanford Site and the ones planned for use with the WTP are continuous, forced-circulation vacuum systems with recycle. In operation, the feed enters a reboiler where the temperature is raised a few degrees and then pumped into the evaporator tank with a fixed liquid level. The evaporator's vacuum-induced vapor stream travels through de-entrainment pads equipped with countercurrent water sprays and then onto the primary condenser.

The primary condenser is a tube in-shell condenser that uses the raw process water as its cooling fluid. The inlet temperature of this fluid can vary from 2 to 24°C based on the time of year. To be conservative, the test condenser emulating the primary condenser was operated at 30°C. The normal temperature of the condensate collection tank is 38°C with a Hi alarm set for 49°C; a 30°C target value was adopted for this laboratory-scale test.

Since the current site evaporators operate best when maintaining approximately a 50% volumetric-wastereduction target, this value was adopted for this small-scale test. Beyond the concentration factor, there is a range of acceptable evaporator operating conditions that are typified by the 242-A Evaporator conditions summarized in Table 4.3. Apart from temperatures and flow ranges, Hanford Site evaporators maintain feed pH between 7 and 10 and operate at a nominal pressure of 60 torr absolute. However, current WTP pretreatment evaporators are expected to operate at a much higher pH. As a result, duplicate evaporator tests were conducted at two pH conditions: 8.4 and 13.0. Since material and laboratory-scale equipment limitations made it infeasible to operate continuously, a single batch concentration campaign was conducted for each pH condition to satisfy the requirements of this evaluation.

4.3.3 Laboratory-Scale Evaporator Description

The laboratory-scale evaporator schematically illustrated in Figure 4.15 and pictorially depicted in Appendix A was used to concentrate a representative sample of the EVS condensate collected during the RSM testing phase of this BNI-sponsored task. The feed evaporator tank was a 4-L reaction vessel, set up in a heating mantle with an automatic temperature controller with a 49°C setpoint. That is within the normal operational range of the evaporator and representative of the daily operation temperature differential between feed and final condensate.

A standard reflux condenser that returns condensate back into the batch vessel was used to simulate the contact reflux that the de-entrainment pads induce in the vacuum-evaporator units. Its target temperature (47°C to 49°C) was nominally set at 49°C to match the boiler control temperature.

	Temperature	Flow, L/mi	n (gpm)	Specific
Stream	°C (°F)	Average	Range	Gravity
Feed	18 - 49 (65 - 120)	340 (90)	265 – 494 (70 – 130)	1.0 - 1.5
Slurry	18 - 66 (65 - 150)	170 (45)	~115-265 (~30-70)	1.0 - 1.6
Process condensate to LERF ^(a) (boil-off)	27 – 43 (80 – 110)	190 (50)	75 - 230 (20 - 60)	1.0
Raw water (cooling)	2 - 24 (35 - 75)	10,410 (2,750)	na	1.0
(a) LERF – Liquid Efflue	ent Retention Facility			

Table 4.3. Typical Range of Temperature and Volume Conditions Seen by the 242-A Evaporator

The primary condenser temperature target was 30° C, $\pm 5^{\circ}$ C, and the actual range achieved during testing was 30° C $\pm 1^{\circ}$ C. A secondary condenser through which the gas stream passed between the collection vessel and the impinger train also was held at this temperature. These two columns successfully condensed the majority of evaporated water into the condensate collection vessel during both tests. The gas stream then entered a Method-29 mercury-vapor wet scrubbing system, which has been previously described in Section 4.1.4.2.

The scrubbed gas then passed though two Drierite water traps, the second of which also had an ~5-cm layer of "Mersorb" to trap any residual mercury vapor remaining before the gas stream passed through the pump and was vented to a hood.

A controlled air purge of approximately 8 to 9 mL/min was introduced into the evaporator vessel to simulate normal air in-leakage. This rate was based on previous Savannah River Technology Center (SRTC) laboratory evaporator design criteria,^(a) scaled by the relative evaporator vessel volumes. The in-leakage source consisted of a metered flow of room air.

4.3.4 Analytical Objectives

The laboratory-scale evaporator system described above was designed to allow mercury partitioning to HLW/LAW, LERF/ETF, and vessel-vent off-gas streams to be projected from the mercury content in the various aqueous solutions generated by the batch operation of the previously described evaporator/sampler system. Specifically, analyses of soluble/insoluble mercury in the evaporator bottoms will establish the HLW/LAW distribution, while the mercury in the evaporator condensate will project

⁽a) Waste Feed Evaporation: Physical Properties and Solubility Determination (U), Savannah River Technology Center, SRT-RPP-2003-00094, Rev 0. Westinghouse Savannah River Company, Savannah River Site, Aiken, SC (May 13, 2003).

AN-107 (C) Simulant Bench-Scale Law Evaporation with Organic Regulatory Analysis, SRT-RPP-2000-00047, Savannah River Technology Center. Westinghouse Savannah River Company, Savannah River Site, Aiken, SC.

partitioning to the LERF/ETF. The distribution of mercury across the chemical scrubbers, shown in Figure 4.15, will establish species-specific off-gas partitioning of mercury as previously described in Section 4.1.4.2.



Figure 4.15. Vacuum-Evaporator and Mercury-Vapor Chemical Scrubbers

5.0 C-104/AY-101 Simulant, Melter Feed, and Product Glass

The primary objective of the Liquid Fed Ceramic Melter (LFCM) waste-vitrification process is to immobilize toxic and/or hazardous elements and/or radionuclides in a suitable matrix that will control environmental release rates to safe and acceptable levels. The vitrification technology achieves this by incorporating and thereby immobilizing these hazardous-waste constituents within a high quality, durable glass matrix. To create a vitreous waste product, glass-forming chemicals have to be added to the waste before it can be vitrified (calcined and melted) in a high-temperature melter. However, to meet stringent waste-form durability criteria, an appropriate glass composition has to be formulated, and its chemical properties (multi-valent oxidation states) have to be carefully controlled.

The BNI-supplied baseline melter-feed material used during RSM testing was commercially prepared by mixing C-104/AY-101 waste-simulant chemicals (~30wt%, oxide basis) with appropriate quantities of glass-forming chemicals. Portions of this baseline feed formulation were subsequently spiked with mercuric nitrate, sodium chloride, and/or sugar to create each of the seven discrete processing conditions of the test. The following discussion provides detailed information concerning the constituents of the feed-streams processed.

5.1 C-104/AY-101 Waste Surrogate and Feed Mixture

As mentioned above, the surrogate waste that was processed during the current RSM test is C-104/AY-101. The equivalent oxide feed formulation to be processed is composed of \sim 30% waste oxides and \sim 70% glass formers composed of B, Li, Na, Si, and Zn oxides. Table 5.1 summarizes the C-104/AY-101 equivalent waste-oxide composition, the relative proportions of the glass formers used, and the resultant target glass composition to be prepared during melter testing.

Although the equivalent oxide presentation in Table 5.1 provides a clear relationship between waste, glass former, and product glass, the actual feed to be processed is not composed of oxides as Table 5.2 clearly illustrates. As is clear from this table, the feed constituents of interest, Hg, Cl, and sucrose, for the current test are not represented in the baseline feed batch and must be added in the appropriate proportions to establish the various test conditions as will be discussed later in this section. Since chlorine is the only additive that was incorporated into the glass product, and its maximum concentration will not exceed 0.1 wt%, the target glass composition detailed in Table 5.1 will accurately represent the product glass composition throughout all phases of testing.

An as-delivered feed sample was collected and analyzed to validate the composition of the sponsorsupplied feed batch. The results of this analysis, which are summarized in Table 5.3, are compared to the target values previously described in Table 5.1.

	C-104/AY-101	Glass Former	
Oxide	HLW Simulant	(as wt% of Glass)	Melter Glass
Al ₂ O ₃	11.89%	—	3.58%
B ₂ O ₃	0.34%	10.71%	10.81%
CaO	1.60%	—	0.48%
Cr ₂ O ₃	0.21%		0.06%
Cs ₂ O	0.17%		0.05%
CuO	0.10%		0.03%
F	0.39%		0.12%
Fe ₂ O ₃	31.67%		9.54%
Ι	0.33%		0.10%
La ₂ O ₃	0.53%		0.15%
Li ₂ O	0.33%	3.21%	3.31%
MnO	5.04%	—	1.52%
Na ₂ O	5.26%	9.91%	11.49%
Nd ₂ O ₃	0.36%	—	0.11%
NiO	1.55%	—	0.47%
PbO	0.41%	—	0.12%
P_2O_5	0.14%	—	0.04%
SiO ₂	8.24%	43.90%	46.39%
TiO ₂	0.07%	—	0.02%
ZnO	0.06%	2.14%	2.16%
ZrO ₂	31.30%	—	9.43%
TOTAL	100.0%	69.87%	100.0%

Table 5.1. Compositional Summary (oxide basis) of the C-104/AY-101 HLW Simulant,Glass Additives, and the Glass for Melter Tests

Given that this feed evaluation was only generated to support a go/no-go decision, the results were found to be in reasonably good agreement with target-value expectation. Of the major feed constituents, only Al and Zr exhibited significant variances. As subsequent, more rigorous feed and glass analyses suggest, the high alumina value appears real, while the Table 5.3 Zr value appears to be biased high. It should be noted that the feed alumina aggregate size distribution was larger than normal and, as a result, created feed-nozzle blockages that were extremely difficult to clear during the melter testing. The low concentrations of the halides and other anionic constituents are due primarily to the fact that only soluble species were analyzed. The impurity levels of chloride present appear to be less than that indicated in Table 5.3, as suggested by subsequent feed analyses to be discussed in Section 5.3.

C-104/AY-101	HLW Simulant	Glass-Forming Additives				
Starting Materials	Target Weight (kg)	Starting Materials	Target Weight (kg)			
Al(OH) ₃	57.72					
H ₃ BO ₃	1.86	Na ₂ B ₄ O ₇ ·10H ₂ O	296.24			
Ca(OH) ₂	6.50					
Cr_2O_3	0.64					
CsOH (50% solution)	1.06					
CuO	0.30					
NaF	2.65					
Fe(OH) ₃ (13% slurry)	977.65					
NaI	1.19					
La(OH) ₃ ·3H ₂ O	2.41					
Li ₂ CO ₃	2.54	Li ₂ CO ₃	81.48			
MnO_2	18.81					
NaOH	13.98	Na ₂ CO ₃	88.76			
Nd ₂ O ₃	1.09					
Ni(OH) ₂	6.01					
FePO ₄ · <i>x</i> H ₂ O (80%)	1.08					
PbO	1.24					
SiO ₂	25.08	SiO ₂	443.45			
TiO ₂	0.22					
ZnO	0.19	ZnO	21.63			
Zr(OH) ₄ · <i>x</i> H ₂ O (50%)	243.75					
NaNO ₂	1.64					
NaNO ₃	6.77					
$H_2C_2O_4$ ·2 H_2O	2.12					
Water	496.50					
TOTAL	1873.00	TOTAL	931.56			
		FEED TOTAL	2804.56			

Table 5.2. Composition of Melter Feed to Produce 1 Metric Ton of Target Glassfrom C-104/AY-101 HLW Simulant (20 wt% suspended solids)

	C-104/A	Y-101 HLW	Simulant			Conc. (mg	g/g-Fd)	
Oxide	Anal	Nomalized	Target	%Dev	Anion	Analysis ^(a)	Target	%Dev
Al ₂ O ₃	3.770	4.590	3.580	28.3	Cl	0.099		
B ₂ O ₃	8.800	10.700	10.800	-0.8	F	0.410	0.461	-11.00
CaO	0.402	0.490	0.480	2.0	I-	0.230	0.388	-41.00
Cr ₂ O ₃	0.042	0.051	0.060	-16.0	NO ₂ ⁻	0.200	0.421	-52.00
Cs ₂ O			0.050		NO ₃ -	1.900	1.900	-0.05
CuO			0.030		$C_2O_4^=$	0.560	0.570	-1.70
Fe ₂ O ₃	8.250	10.100	9.540	5.4	(a) Sol	uble		
La ₂ O ₃	0.090	0.110	0.150	-27.0				
Li ₂ O	2.300	2.800	3.310	-15.0				
MnO	1.420	1.730	1.520	13.5				
Na ₂ O	9.200	11.200	11.500	-2.5				
Nd ₂ O ₃			0.110					
NiO	0.401	0.488	0.470	3.9				
PbO	0.102	0.125	0.120	3.9				
P_2O_5			0.040					
SiO ₂	40.500	49.400	46.400	6.4				
SrO	0.003	0.003						
TiO ₂			0.020					
ZnO	1.310	1.590	2.160	-26.0				
ZrO ₂	5.480	6.680	9.430	-29.0				
Sum	82.100	100.000	100.000		1			

 Table 5.3. As Received Simulant Feed Sample Composition

Beyond chemical composition, the physical properties of the C-104/AY-101 surrogate melter feed were also measured. Specifically, measurements were conducted on the baseline melter feed to establish density as well as solids, oxide, and water loadings. These values are summarized in Table 5.4.

Table 5.4.	Physical Propertie	s of Baseline, C-10	4/AY-101 Surrogate Melte	r Feed
		Weight Loss (%)	Slurry Feed Loading (g/L)	

		Weight	Loss (%)	Slurry Feed Loading (g/L)			
Sample	Sp. Grav.	Drying	Ignition	H ₂ 0	Solids	Oxides	
RSM-Hg-6	1.44	55.3	61.5	797	644	555	
RSM-Hg-7	1.45	55.1	61.4	798	650	559	
RSM-Hg-8	1.45	55.4	61.7	802	645	555	
Average	1.45	55.3	61.5	799	646	556	
StDev	0.26%	0.31%	0.23%	0.35%	0.51%	0.44%	

5.2 Feed Mixture/Test Condition Description

As mentioned above, the surrogate C-104/AY-101 feed mixture that was processed during the current test is complete and inclusive of all constituents except for mercury, chlorine, and sucrose. To satisfy Test Plan objectives (see Section 2), separate feed batches were prepared to provide the seven different conditions for process-feed compositions previously referred to. The feed additives used and the quantities to be dispensed and processed, assuming maximum expected processing conditions, are summarized in Table 5.5.

	TstCond		Wt%		g/L @ 0.56 kg-Ox/L			Mass Processed (g)			
Batch#	Hg:Cl:Red	Duration	HgO	Cl	Sugar	Hg(NO ₃) ₂ ·H ₂ O	NaCl	Sugar	Hg(NO ₃) ₂ ·H ₂ O	NaCl	Sugar
	Lo:Lo:Lo	18	0.05	0.009	0	0.44	0.083	0	8.8	1.64	0
1	Lo:Hi:Lo	18	0.05	0.06	0	0.44	0.550	0	8.8	10.95	0
	Lo:Hi:Hi	6	0.05	0.06	0.90	0.44	0.550	5	2.9	3.65	33
	Hi:Lo:Lo	18	0.15	0.009	0	1.32	0.083	0	26.3	1.64	0
2	Hi:Hi:Lo	18	0.15	0.06	0	1.32	0.550	0	26.3	10.95	0
	Mx:Hi:Lo	18	0.95	0.06	0	8.37	0.550	0	166.6	10.95	0
3	Mx:Hi:Hi	6	0.95	0.06	0.90	8.37	0.550	5	55.54	3.65	33
	Total	102							295	43.4	66

Table 5.5. Target Test Condition, Feed Additives, Concentrations, and Processing Expectations

The actual processing conditions executed during RSM testing varied little from the above target values, as the assumed and achieved processing rates were essentially identical. Only processing durations varied somewhat from plan values, as shown in Table 5.6; these plan variations, although inconsequential, were prompted by sampling needs. The feed-preparation sheets used to prepare all melter-feed-tank batches throughout RSM testing are summarized in Appendix B.

5.3 Melter-Feed Characteristics

Melter-feed samples were collected from each feed batch prepared during the C-104/AY-101 melter test. These feed samples were subsequently analyzed to determine their physical properties and chemical composition. Table 5.7 presents the physical properties associated with all the feed batches prepared during RSM testing. Also presented in this table, for comparison purposes, are the pre-test evaluation results previously discussed. The reproducibility is remarkable as two separate laboratory groups evaluated the samples collected before and during the melter test. Furthermore, the consistency of the data validates the adequacy of the tank agitations and delivery systems employed during RSM testing.

Т	est Condition	Test I	Period	Duration	Feed
#	Hg/Cl/Red	Start	Stop	(h)	Proc. kg
0	ShakeDwn	07/28/03 09:25	07/28/03 14:58	5.6	11.1
1	Lo/Lo/Lo	07/28/03 16:33	07/29/03 11:22	18.8	28.1
2	Lo/Hi/Lo	07/29/03 11:22	07/30/03 05:47	18.4	26.7
3	Lo/Hi/Hi	07/30/03 05:47	07/30/03 13:26	7.6	10.4
Idle	Transition	07/30/03 13:26	07/30/03 16:16	2.8	
4	Hi/Lo/Lo	07/30/03 16:16	07/31/03 10:55	18.7	27.3
5	Hi/Hi/Lo	07/31/03 10:55	08/01/03 05:26	18.5	26.1
6	Mx/Hi/Lo	08/01/03 05:26	08/01/03 22:20	16.9	27.0
7	Mx/Hi/Hi	08/01/03 22:20	08/02/03 05:11	6.8	10.9
			Total	114.2	167.6

Table 5.6. Actual RSM Testing Conditions

Table 5.7. Physical Properties of Surrogate C-104/AY-101 Melter Feeds

			Weight	Loss (%)	Slurry Feed Loading (g/L)			
Sample	Hg/Cl/Red	Sp. Grav.	Drying	Ignition	H ₂ 0	Solids	Oxides	
RSM-Hg-6	PreTest	1.44	55.3	61.5	797	644	555	
RSM-Hg-7	PreTest	1.45	55.1	61.4	798	650	559	
RSM-Hg-8	PreTest	1.45	55.4	61.7	802	645	555	
RSM-Hg-13	Lo/Lo/Lo	1.46	56.0	61.8	816	642	556	
RSM-Hg-34	Lo/Hi/Lo	1.46	55.4	61.9	809	651	557	
RSM-Hg-42	Lo/Hi/Hi	1.45	55.5	61.8	806	645	554	
RSM-Hg-51	Hi/Lo/Lo	1.45	55.9	62.0	809	638	549	
RSM-Hg-65	Hi/Hi/Lo	1.46	55.9	61.9	813	643	556	
RSM-Hg-80	Mx/Hi/Lo	1.45	55.7	62.0	805	641	549	
RSM-Hg-87B	Mx/Hi/Hi	1.44	55.1	61.8	796	649	552	
	Average	1.45	55.5	61.8	805	645	554	
	StDev	0.44%	0.57%	0.34%	0.85%	0.65%	0.58%	

The feed's total-oxide values derived from laboratory loss-on-drying (LOD, at 110°C) and loss-onignition (LOI of wet waste at 1050°C) tests are found to satisfactorily agree with the feed-formulation expectations of 550 g-Ox/L-Fd (JM Perez, private communication). The total average oxide loading of the feed calculated from all the feed processed and the glass produced throughout the duration of RSM testing, 505 g-oxide/L, also agrees reasonably well with the values derived from LOI measurements, given that significant bubbler-aggravated feed losses to the melter's walls and plenum could not be accounted for, as will be described later.

The oxide-equivalent compositions of cationic species for each of the unique feed batches prepared during RSM testing are summarized in Table 5.8 and compared to target baseline values based upon the

previously defined waste- and glass-composition values. All of the major feed-component concentrations were reasonably consistent throughout the nominal 5 days of testing. With the exception of alumina, which appears to be biased high, and the Ba, K, and Sr chemical impurities, there were no trends or large variations of importance, and most components, except for P and Ti, were reasonably close to their target values. The large variances observed for P and Ti feed constituents may be due to analytical-detection-limit difficulties associated with these elements.

An analysis of soluble anionic species in the feed was also carried out for completeness, and the results obtained are summarized in Table 5.9. The most important of these anions are chloride and nitrate as these are constituents of the feed additives used, i.e., NaCl and Hg(NO₃)₂·H₂O. The as-found concentrations of these anions are found to be reasonably close to target values when the feed additives used are taken into account. The slightly elevated chloride levels present under low-chlorine processing campaigns is most likely due to contributions from chemical impurities that are clearly responsible for the presence of sulfate. The feed nitrite content is shown to be consistently below expectation, as is the concentration of iodide, whose solubility properties may be responsible for the lower-than-expected condition. On the other hand, the soluble fluoride present is reasonably close to the target value as is the reductant oxalate. Being a test-matrix variable, the nitrate agreement in Table 5.9 circumstantially validates the appropriateness of the Hg(NO₃)₂·H₂O additions made throughout the test.

Apart from the problems associated with identifiable analytical biases, the contaminants and trace constituents just discussed, the overall melter-feed composition data agree quite well with feed-formulation expectation values. In general, it appears there was good control over feed composition throughout the entire test period, which will be corroborated when the glass data are subsequently discussed.

5.4 Product Glass Characteristics

Glass grab samples were collected from RSM pours throughout the C-104/AY-101 test period. When operated without glass-pool-bubbler agitation, melter glass discharges nominally occur every 2 hours, whenever the glass-pool head pressure overcomes the flow resistance presented by the slightly cooler overflow channel. However, the effect of bubbler-glass-pool agitation was to create quasi-continuous overflow conditions with very low glass flows. Nevertheless, sufficient glass was collected during each phase of testing to allow required analyses to be conducted. The Fe⁺²:Fe^{Tot} ratio measurements were conducted to assess the impact of changing feed-reductant levels upon the oxidation state of the glass product. In addition, representative glass samples collected from each of the distinct feeds were compositionally analyzed, and some of these were subjected to the toxicity characteristic leach procedure (TCLP). The results of these measurements will now be discussed.

			Feed	Sample W	eight Perc	ent Comp	osition			
	RSM-	RSM-	RSM-	RSM-	RSM-	RSM-	RSM-			
Oxide	13	34	42	51	65	80	87B	Average	Target	%Dev
Al_2O_3	4.350	4.290	4.340	4.280	4.500	4.370	4.460	4.370	3.580	22.100
B_2O_3	10.200	11.000	10.600	10.500	10.300	11.200	10.500	10.600	10.800	-1.900
BaO	0.013	0.014	0.014	0.015	0.014	0.015	0.017	0.015		
CaO	0.514	0.462	0.445	0.451	0.458	0.404	0.461	0.456	0.480	-4.900
Cr ₂ O ₃	0.069	0.061	0.068	0.064	0.065	0.067	0.059	0.065	0.060	8.140
Cs ₂ O	0.036	0.035	0.032	0.030	0.035	0.034	0.033	0.034	0.050	-33.000
CuO	0.076	0.030	0.037	0.035	0.049	0.032	0.024	0.040	0.030	34.400
Fe ₂ O ₃	9.240	9.150	9.290	9.100	9.270	9.230	9.240	9.220	9.540	-3.400
HgO	0.052	0.053	0.043	0.160	0.109	1.020	0.895	0.334		
K ₂ O		0.235	0.292	0.266	0.208			0.250		
La ₂ O ₃	0.121	0.103	0.119	0.119	0.116	0.132	0.126	0.119	0.150	-20.000
Li ₂ O	3.050	3.470	4.160	3.540	4.690	3.710	3.180	3.690	3.310	11.300
MnO	1.680	1.650	1.710	1.650	1.640	1.600	1.620	1.650	1.520	8.590
Na ₂ O	11.600	12.700	12.000	11.700	11.300	11.400	11.300	11.700	11.500	1.870
NiO	0.466	0.466	0.471	0.467	0.473	0.471	0.459	0.468	0.470	-0.510
P_2O_5	0.177	0.216	0.206	0.203	0.215	0.253	0.249	0.217	0.040	443.000
PbO	0.122	0.120	0.125	0.119	0.118	0.120	0.120	0.120	0.120	0.404
SiO ₂	46.200	44.600	44.400	44.700	44.800	44.400	45.300	44.900	46.400	-3.200
SrO	0.007	0.005	0.023	0.007	0.005	0.006	0.004	0.008		
TiO ₂	0.110	0.109	0.113	0.113	0.116	0.107	0.117	0.112	0.020	461.000
ZnO	1.440	1.420	1.450	1.420	1.440	1.440	1.430	1.440	2.160	-34.000
ZrO ₂	10.500	9.820	10.100	11.000	10.100	9.960	10.500	10.300	9.430	9.000
Sum	100.000	100.000	100.000	100.000	100.000	100.000	100.000	100.000	100.000	0.020

Table 5.8. Melter-Feed-Batch Composition

5.4.1 Oxidation State Results

As previously discussed (see Section 5.2), the feed sucrose concentration was an experimental parameter during RSM testing. Reductant additives can sometimes be beneficial in optimizing feed-processing rates and in nitrate destruction (NO_x abatement). The major concern regarding the use of reductant is producing an overly reduced glass that can affect glass durability and accelerate insoluble metal production, settling rates, and resultant melter electrode shorting. Based upon previous melter testing of a similar C-106/AY-102 waste simulant, a sugar loading (5g/L-Fd) was chosen to produce a measurable change in the fraction of iron in its +II oxidation state but without producing an overly reduced glass (Fe⁺²:Fe^{tot} >0.3). The post-test analysis of glass samples collected during the test has been subsequently examined to determine the effects of this incremental change in feed reductant upon the glass-oxidation state.

					Concen	tration ((mg/g)			
Sample ID	Hg/Cl/Red	Cl ⁻¹	NO_2^{-1}	Br ⁻¹	NO ₃ ⁻¹	PO ₄ -3	SO_4^{-2}	$C_2O_4^{-2}$	\mathbf{I}^{-1}	F ⁻¹
RSM-Hg-13	Lo/Lo/Lo	0.049	0.237	< 0.012	1.97	< 0.1	0.110	0.496	< 0.1	0.372
RSM-Hg-34	Lo/Hi/Lo	0.182	0.244	< 0.016	2.03	< 0.1	0.115	0.507	0.150	0.389
RSM-Hg-42	Lo/Hi/Hi	0.172	0.246	< 0.015	2.03	< 0.1	0.110	0.504	< 0.1	0.373
RSM-Hg-51	Hi/Lo/Lo	0.053	0.238	< 0.014	2.16	< 0.1	0.107	0.494	< 0.1	0.371
RSM-Hg-65	Hi/Hi/Lo	0.210	0.256	< 0.011	2.38	< 0.1	0.115	0.576	< 0.1	0.385
RSM-Hg-80	Mx/Hi/Lo	0.211	0.261	< 0.010	3.68	< 0.1	0.110	0.589	< 0.1	0.352
RSM-Hg-87B	Mx/Hi/Hi	0.229	0.279	< 0.013	3.76	< 0.1	0.121	0.612	< 0.1	0.354
	Average		0.251				0.113	0.540		0.371
	Target		0.421					0.570	0.388	0.461
	Lo	0.035			2.010					
	Hi	0.231			2.230					
	Mx				4.000					

Table 5.9. Soluble Anionic Feed Species

To accomplish this, a chemical method for measuring an Fe(II) complex colorimetrically at a wavelength of 515 η m was used to establish the vitreous iron fraction in the +II valance state. This is accomplished by dissolving a powdered-glass sample in H₂SO₄ and HF, buffering the resultant solution with sodium acetate/boric acid solution containing an o-phenanthroline complexing agent, and conducting an Fe(II)-specific absorbance measurement at 515 η m. Ascorbic acid is subsequently used to reduce all remaining iron in the dissolved sample to the Fe(II) state, which allows the total Fe to be measured by a subsequent absorption measurement. These results allow the fraction of iron in the +II valence state in the glass sample to be directly determined.

The oxidation-state results obtained from glass samples obtained during current RSM testing are summarized in Table 5.10 along with the corresponding sugar-loading values used. These data show that the oxalate present in the baseline feed was sufficient to measurably affect the multi-valent distribution of iron in the glass. At first glance, it appears that inconsistent results were obtained when adding sugar to the melter feed. However, grab-glass samples from the melter's overflow were not available during the Mx/Hi/Hi processing campaign. As a result, a post-test glass sample had to be collected from the surface of the glass-collection canister. Since this canister is contained in a clam-shell furnace controlled at an 800°C temperature, partial reoxidation of the collected glass resulted. Since duplicate redox determinations demonstrated relative standard errors of the order of 3%, and the two glasses produced with no sugar additive demonstrated good reproducibility, it is fairly safe to conclude that sugar, at a 5 g/L-Fd loading, produced a measurable (~3.5×) change in the oxidation state of the glass produced and that this change was within acceptable limits, i.e., Fe⁺²:Fe^{Tot} ≤30%.

		Fe ⁺² :Fe ^{Tot}	Sugar
Sample	Hg/Cl/Red	(%)	(g/L-Fd)
RSM-Hg-38	Lo/Hi/Lo	1.49	0
RSM-Hg-43	Lo/Hi/Hi	5.46	5
RSM-Hg-58	Hi/Lo/Lo	1.60	0
RSM-Hg-90	Mx/Hi/Hi	1.95	5

 Table 5.10.
 C-104/AY-101 Sugar Loadings and Glass-Oxidation States

5.4.2 Compositional Data

Glass-product compositional data associated with samples taken throughout all phases of melter processing are summarized in Table 5.11 along with the corresponding target value for the baseline C-104/AY-101 surrogate feed. Apart from the contaminant (e.g., Ba, Sr, S) and detection limit-driven variances that have been previously discussed (Section 5.3), all major oxide constituents compared favorably with their respective target values. However, like the feed data discussed earlier, the glass-compositional data also indicate a higher-than-expected alumina concentration. Although mercury was detected in essentially all C-104/AY-101 glasses produced, it was found to be present at vanishingly low concentrations.

Overall, the target/actual composition comparison for the glass samples analyzed mirror corresponding results obtained from feed data previously discussed. Based on a simple comparison of product-glass analytical and target-glass compositions, the classic volatiles/semi-volatiles feed components (e.g., B, alkalis) do not appear to have partitioned significantly to the process exhaust. Partitioning values or melter DFs will be discussed in a later section dealing with off-gas emission characterization.

5.4.3 TCLP Results

Representative glass samples generated under worst-case testing conditions (Mx/Hi/Lo and Mx/Hi/Hi) were subjected to EPA's TCLP test (TCLP 1992). In that procedure, crushed glass is leached with a sodium acetate buffer solution for 18 hours at 22°C while under constant, end-over-end agitation. The leachate compositions were subsequently analyzed, and the results obtained are summarized in Table 5.12.

The concentrations of all hazardous analytes, except for Ba and Cr (for which estimates are provided), were found to be below their instrumental detection limits, and all were below their respective Universal Treatment Standard (UTS) limits. In the case of Ba and Cr, the estimates provided are at least an order of magnitude below their corresponding UTS limits. The extremely low mercury concentrations found in the glass clearly present no adverse leaching problems that would preclude the glass vitrification product from conforming with all existing Resource Conservation and Recovery Act (RCRA) land-disposal limits (40 CFR 268).

			Normali	zed Glass	Oxide Co	omposition	(Wt%)			
Oxide	RSM-23	RSM-38	RSM-43	RSM-58	RSM-73	RSM-85B	RSM-90	Average	Target	% Dev.
Al_2O_3	4.790	4.760	4.850	4.680	4.680	4.710	4.710	4.740	3.580	32.300
B_2O_3	10.100	10.500	10.800	10.300	10.400	10.500	10.500	10.400	10.800	-3.400
BaO	0.016	0.015	0.014	0.013	0.013	0.014	0.014	0.014		
CaO	0.555	0.551	0.600	0.598	0.549	0.562	0.562	0.568	0.480	18.400
Cr_2O_3	0.167	0.102	0.103	0.086	0.092	0.086	0.086	0.103	0.060	71.800
Cs_2O	0.035	0.032	0.033	0.032	0.032	0.035	0.035	0.033	0.050	-33.000
CuO			0.043	0.028	0.020	0.034	0.034	0.032	0.030	5.850
Fe_2O_3	9.430	9.400	9.690	9.370	9.390	9.460	9.460	9.460	9.540	-0.880
HgO	1.90E-06			1.41E-06	1.49E-06	4.84E-06	4.91E-06	5.08E-06		
La_2O_3			0.123	0.122	0.115	0.119	0.119	0.120	0.150	-20.000
Li ₂ O	3.020	3.060	3.350	3.340	3.090	3.160	3.160	3.170	3.310	-4.300
MnO	1.850	1.760	0.180	1.730	1.720	1.740	1.740	1.530	1.520	0.643
Na ₂ O	10.800	10.600	11.800	11.400	10.800	10.900	10.900	11.000	11.500	-4.000
Nd_2O_3									0.110	
NiO	0.523	0.529	0.509	0.482	0.525	0.523	0.523	0.516	0.470	9.850
P_2O_5			0.243	0.218	0.246	0.279	0.279	0.253	0.040	533.000
PbO			0.129	0.127	0.132	0.129	0.129	0.129	0.120	7.680
SO_3					0.222			0.222		
SiO ₂	47.400	47.400	45.000	45.300	46.600	46.100	46.100	46.300	46.400	-0.290
SrO	0.128	0.043	0.055	0.042	0.025	0.022	0.022	0.048		
TiO ₂	0.118	0.133	0.131	0.127	0.124	0.128	0.128	0.127	0.020	535.000
ZnO	1.400	1.440	1.480	1.440	1.450	1.470	1.470	1.450	2.160	-33.000
ZrO_2	9.750	9.730	10.900	10.600	9.810	10.000	10.000	10.100	9.430	7.140
Sum	100.000	100.000	100.000	100.000	100.000	100.000	100.000	100.000		

Table 5.11. Oxide-Composition of Melter C-104/AY-101 Glass Samples

Table 5.12. TCLP Leachate Concentrations from C-104/AY-101 Glasses

	Leachate Concentration (mg/L): Estimate/Detection-Limit ^(a) Su									
Item	Ag	As	Ba	Cd	Cr	Hg	Pb	Se	(g/L-Fd)	
UTS Limit	0.14	5	21	0.11	0.60	0.025	0.75	5.7	N/A	
Detect Limit	0.015	0.092	0.0023	0.0050	0.0050	0.00050	0.092	0.086	N/A	
RSM-Hg-101	ND	ND	0.24	ND	0.038	ND	ND	ND	0	
RSM-Hg-102	ND	ND	0.30	ND	0.022	ND	ND	ND	5	
(a) $ND = not of$	(a) ND = not detected; N/A = not applicable									

6.0 Process Characterization

RSM evaluation of the simulated, mercury-containing C-104/AY-101 flowsheet was initiated on July 28, 2003, and concluded on August 2, 2003. During this ~120-h period, C-104/AY-101 melter feeds containing varying concentrations of mercury, chlorine, and reductant, but, at a fixed 30-wt% waste loading, were successfully processed. The observations and experimental test results derived from all phases of C-104/AY-101 melter testing will now be discussed. For the interested reader, low resolution, hourly data logs of the most important melter and off-gas process parameters are reproduced in Appendix B.

6.1 Processing Observations and Parameters

The melting characteristics of the C-104/AY-101 feed formulation under study were found to be independent of the range of mercury, chlorine, and sugar concentrations used during melter testing. Bubblers were employed throughout all phases of testing, and glass-pool characteristics remained invariant independent of feed-spike loading. When operated without bubblers, reductant additives are often required to control the oxidation states of multivalent glass constituents to preclude melter-glass foaming that can adversely affect processing rates. Although the feed oxalate concentration was sufficiently high to preclude foaming (see Section 5.4.1), no changes in cold-cap characteristics or processing behavior were observed when the sugar additive was used.

Beyond the influence of sugar upon the glass oxidation state (see Section 5.4.1), the sugar additive was also found to produce noticeable changes in the melter's NO_x source term. In both processing campaigns involving sugar, significant quantities of the nitrate present in the waste appeared to be efficiently reduced to N₂. During all processing campaigns containing no sugar additive, NO dominated the melter's nitrogen oxide emission source term. However, whenever sugar was present in the melter feed, off-gas concentrations of NO were significantly reduced. This subject will be discussed in greater detail in Section 7.1.

The processing characteristics documented for all feed batches prepared during RSM testing are summarized in Figure 6.1. Because of the manner in which the melter test conditions were staged (see Section 2), essentially all feed-batch changes were conducted without interrupting melter processing. The only exception to this statement occurred during the transition between processing conditions 3 and 4 (see Section 2) when a high reductant loading condition was being replaced with a low one. Since the high condition required sugar addition and the low condition contained no sugar, the melter's feed tank had to be emptied before the new batch could be prepared. Consequently, the feed-processing history is composed of two quasi-continuous curves with amazingly similar slopes. Figure 6.2 and Figure 6.3 summarize the processing characteristic during each of the two feeding campaigns.

The average feed-processing rates displayed in the latter two figures represent both active feed-processing periods and non-feeding time intervals, and as such are conservative estimates. The fact that noticeable gaps are not apparent in these processing-campaign curves indicate that down times associated primarily





Figure 6.1. Process History of RSM Feed Batches and Average Overall Processing Rate



—— Cmpgn #1 — – Linear (Cmpgn #1)

Figure 6.2. Processing History of the First Quasi-Continuous Processing Campaign



Figure 6.3. Processing History of the Second Quasi-Continuous Processing Campaign

As is apparent from Figure 6.2 and Figure 6.3, individual batch-processing rates varied somewhat from the composite average curve. Table 6.1 summarizes the processing rates and derived glass-production rates for all individual test conditions evaluated during RSM testing. The relationship between feed-processing and glass-production rates used in this table was derived from empirical feed-property data discussed in Table 5.7.

Referring to the Table 6.1 data, the highest processing rate achieved was when the melter-system operations were being aligned. In fact, this rate represents an over-feeding condition that ultimately produced a complete cold-cap-coverage condition when the feed pile surrounding the bubbler vents collapsed. Although, at first glance, Test Condition #3 results appear to suggest that processing the C-104/AY-101 feed formulation with a sugar additive is not advantageous; this conclusion is not, however, substantiated by Condition #7 results.

Also included in Table 6.1 are specific glass-production rates derived from the experimentally determined total oxide loading of the feed (see Section 5.3), the glass pool surface area (6-in. dia), and the corresponding melter feeding rates. These calculated values comfortably exceed the minimum acceptable rate of 0.4 MT/h/m² and closely approach the maximum expectation value (0.8 MT/h/m²). The derived overall average-production-rate value, however, does not agree completely with the actual glass-accumulation information (0.54 vs 0.49 kg/h) that was derived from manually recorded data throughout the duration of the test. These glass-accumulation data are summarized in Figure 6.4 and in Table 6.2.

Test	t Condition	Test I	Period	Duration	on Feed		Glass ^(a)		s ^(a)
#	Hg/Cl/Red	Start	Stop	(h)	kg	kg/h ^(b)	kg	kg/h ^(b)	$MT/d/m^{2(b)}$
0	ShakeDwn	07/28/03 09:25	07/28/03 14:58	5.6	11.1	2.00	4.28	0.77	1.01
1	Lo/Lo/Lo	07/28/03 16:33	07/29/03 11:22	18.8	28.1	1.49	10.8	0.58	0.76
2	Lo/Hi/Lo	07/29/03 11:22	07/30/03 05:47	18.4	26.7	1.45	10.3	0.56	0.73
3	Lo/Hi/Hi	07/30/03 05:47	07/30/03 13:26	7.6	10.4	1.36	4.02	0.52	0.69
Idle	Transition	07/30/03 13:26	07/30/03 16:16	2.8					
4	Hi/Lo/Lo	07/30/03 16:16	07/31/03 10:55	18.7	27.3	1.47	10.5	0.56	0.74
5	Hi/Hi/Lo	07/31/03 10:55	08/01/03 05:26	18.5	26.1	1.41	10.0	0.54	0.71
6	Mx/Hi/Lo	08/01/03 05:26	08/01/03 22:20	16.9	27.0	1.60	10.4	0.61	0.81
7	Mx/Hi/Hi	08/01/03 22:20	08/02/03 05:11	6.8	10.9	1.59	4.19	0.61	0.80
			Total/Average	114.2	168	1.48	64.5	0.57	0.75
			Sample/OG Corrections		160	1.40	61.5	0.54	0.71
(a) C	Calculated valu	e.							
(b) A	Average exclud	es Cond# 0.							

Table 6.1. C-104/AY-101 Feed Processing Rates and Derived Glass-Production Rates

Although the tabular and graphically derived average glass rates are nominally equivalent, the variability in test-segments glass rates appears to be due to the highly discrete nature of the manually collected data, as interpolation between recorded values was not conducted. Since data recording is not synchronized to process events, maximum glass-canister weights are generally not recorded, as it would require an unlikely coincidence of events. As a result, the total glass estimate derived from these data is lower than the value established from post-test measurements (54.6 vs 55.7 kg).

The larger glass-production estimate derived from feed-processing data (61.5 vs 55.7 kg) is believed to be predominantly due to unaccounted-for feed material encrusting the inside surfaces of the melter's plenum. The cold-cap turbulence created by the melter's bubblers in conjunction with low plenum temperatures (~400°C) created ideal conditions to accumulate unfused plenum deposits. However, there was no reasonable way to measure this internal residue at the conclusion of testing.

Although the $\sim 10\%$ difference between feed projections and glass could also have resulted from variations of the oxide loading of the batched baseline feed, no dilution of the master or melter-feed tank contents occurred during melter testing, and all batch-feed transfers were extracted from a tank bottom-to-top recirculation loop. Since measurements of the physical properties of feed-batch were, as described in Section 5.3, remarkably consistent; unaccounted for feed losses must be responsible for the variance between projected and measured glass results.

6.2 Operating Parameters

The primary functional indicators of the RSM processing system are temperatures (glass melt, melter plenum space, and post-film cooler off-gas stream), pressure (plenum, off-gas), and melter electrical

values (electrode current and voltage). The process data relating to these operating parameters will now be discussed.



Figure 6.4. C-104/AY-101 Glass-Accumulation Data

|--|

Te	st Condition	Test	Period	Duration		Glass Pro	duced
#	Hg/Cl/Red	Start	Stop	(h)	kg	kg/h ^(a)	MT/d/m ^{2*}
0	ShakeDwn	7/28/03 9:25	7/28/03 14:58	5.6	1.99	0.36	0.47
1	Lo/Lo/Lo	7/28/03 16:33	7/29/03 11:22	18.8	10.5	0.56	0.74
2	Lo/Hi/Lo	7/29/03 11:22	7/30/03 5:47	18.4	9.21	0.50	0.66
3	Lo/Hi/Hi	7/30/03 5:47	7/30/03 13:26	7.6	2.50	0.33	0.43
Idle	Transition	7/30/03 13:26	7/30/03 16:16	2.8			
4	Hi/Lo/Lo	7/30/03 16:16	7/31/03 10:55	18.7	9.85	0.53	0.69
5	Hi/Hi/Lo	7/31/03 10:55	8/1/03 5:26	18.5	8.95	0.48	0.64
6	Mx/Hi/Lo	8/1/03 5:26	8/1/03 22:20	16.9	9.11	0.54	0.71
7	Mx/Hi/Hi	8/1/03 22:20	8/2/03 5:11	6.8	2.50	0.36	0.48
			Total/Average	114.2	54.6	0.47	0.62
			-	Post-Test, Derived Values	55.7	0.49	0.64
(a) Av	verage excludes Co	ond# 0					

6.2.1 Process Temperatures

During RSM testing, the temperatures of the following process items were routinely logged:

- Glass Melt
- Melter Plenum
- Post Film-Cooler Off-Gas
- Melter Off-Gas (horizontal run)
- Post EVS Off-Gas
- Post HEME Off-Gas
- EVS Spray
- EVS Condensate.

Table 6.3 summarizes the maximum, minimum, and average temperatures of the melter's glass, plenum, and off-gas stream throughout the entire duration of C-104/AY-101 melter-flowsheet evaluation. Also presented in this table are the standard deviations associated with the temperatures listed. The magnitude of temperature variations (StDev) about the mean is indicative of overall process parameter stability. Table 6.4 provides similar daily tabular data for the melter's kiln, overflow spout, and canister oven.

Test compliance with process operational-temperature target values described in Section 2 of this report are clearly demonstrated by the summary data presented in Table 6.3, although the averaging also included the shakedown and idling phases of the test. Temporal melter and off-gas process-temperature data collected during the C-104/AY-101 flowsheet evaluations are also graphically presented in Appendix C on a daily basis. These graphical data clearly show the influence of the extremely high ambient temperatures accompanying the test (Mon 108°F, Tues 108°F, Wed 108°C, Th 106°F, Fri 104°F). The diurnal, cyclic-temperature variations of the HEME exhaust most clearly document the influence of ambient-temperature conditions upon the process parameters that were not actively controlled. Increased bubbler-induced melter-vessel heat losses obviated the need to use kiln power to achieve the melter tank's external temperature setpoint. More will be said about this condition in Section 6.2.3.

	Temperature (°C)								
Date/Item	Glass 1 ^(a)	Glass 2	Plenum	FCExh	MOGLine	PostEVS	PstHEME	EVSpray	EVSTnk
7/28/2003									
Average	1150.0	1140.0	456.0	202.0	221.0	34.1	34.6	37.5	33.6
Std Dev	21.1	24.3	59.0	28.5	20.7	1.0	2.0	3.4	1.3
Min	1050.0	1020.0	307.0	149.0	159.0	30.0	28.5	31.0	29.0
Max	1160.0	1160.0	607.0	307.0	269.0	38.0	37.0	42.0	36.0
7/29/2003									
Average	1150.0	1150.0	421.0	223.0	232.0	33.7	33.7	34.6	33.3
Std Dev	2.2	5.6	36.8	16.0	13.4	1.1	2.3	6.0	1.0
Min	1140.0	1130.0	312.0	180.0	208.0	31.0	29.4	22.0	31.0
Max	1170.0	1170.0	551.0	276.0	263.0	40.5	38.0	43.6	36.0
7/30/2003									
Average	1150.0	1150.0	433.0	235.0	240.0	34.2	34.1	35.4	33.9
Std Dev	2.4	7.4	72.5	8.8	8.2	1.3	2.5	7.0	1.2
Min	1140.0	1130.0	343.0	216.0	215.0	31.6	30.0	24.0	31.0
Max	1160.0	1170.0	658.0	268.0	252.0	44.0	38.5	47.5	37.0
7/31/2003									
Average	1150.0	1140.0	409.0	219.0	220.0	34.1	34.0	36.0	33.8
Std Dev	2.4	6.3	23.1	14.3	8.2	0.6	1.7	5.4	0.4
Min	1140.0	1130.0	332.0	184.0	199.0	31.9	30.5	26.0	31.9
Max	1160.0	1160.0	500.0	256.0	244.0	41.1	37.0	44.5	35.0
8/1/2003									
Average	1150.0	1140.0	409.0	182.0	204.0	33.3	33.5	36.1	32.9
Std Dev	2.2	5.6	18.2	12.0	9.8	0.7	1.9	5.1	0.6
Min	1140.0	1110.0	343.0	153.0	182.0	31.6	30.0	25.5	31.6
Max	1160.0	1150.0	509.0	213.0	227.0	37.0	37.0	44.0	34.5
8/2/2003									
Average	1150.0	1130.0	407.0	165.0	198.0	32.3	32.1	32.1	32.1
Std Dev	2.5	4.0	24.8	7.5	3.6	0.4	0.9	1.6	0.4
Min	1140.0	1120.0	371.0	148.0	190.0	31.0	30.5	29.4	31.0
Max	1160.0	1150.0	476.0	178.0	204.0	33.5	34.1	36.0	33.0
7/28 - 8/2									
Average	1150.0	1140.0	422.0	211.0	222.0	33.8	33.9	35.6	33.4
Std Dev	7.9	12.4	47.5	26.4	17.9	1.1	2.1	5.6	1.0
Min	1050.0	1020.0	307.0	148.0	159.0	30.0	28.5	22.0	29.0
Max	1170.0	1170.0	658.0	307.0	269.0	44.0	38.5	47.5	37.0
Target									
Range/Value	1150		400-450		200-250				
(a) Control V	/alue.								

 Table 6.3. RSM's Operating Temperature Characteristics

	Temperature (°C)							
		Kiln		Pour	Can			
Date/Statistic	Bot	Mid	Тор	Spout	Oven			
7/28/02								
Average	807.00	855.00	767.00	1060.00	685.00			
Std Dev	26.60	12.30	30.90	21.50	160.00			
Minimum	766.00	844.00	724.00	1030.00	125.00			
Maximum	909.00	901.00	887.00	1110.00	775.00			
7/29/02								
Average	845.00	878.00	805.00	1130.00	748.00			
Std Dev	13.80	10.60	15.60	23.30	18.00			
Minimum	820.00	861.00	779.00	1080.00	492.00			
Maximum	872.00	893.00	835.00	1150.00	781.00			
7/30/02								
Average	844.00	877.00	807.00	1090.00	749.00			
Std Dev	32.50	24.80	32.90	111.00	13.40			
Minimum	752.00	811.00	711.00	769.00	484.00			
Maximum	874.00	898.00	837.00	1180.00	777.00			
7/31/02								
Average	838.00	873.00	800.00	1090.00	822.00			
Std Dev	20.00	14.50	20.50	80.30	41.90			
Minimum	757.00	818.00	716.00	801.00	587.00			
Maximum	857.00	886.00	820.00	1180.00	883.00			
8/1/02								
Average	863.00	890.00	825.00	1100.00	869.00			
Std Dev	27.20	21.10	27.90	81.60	11.20			
Minimum	760.00	821.00	717.00	789.00	641.00			
Maximum	884.00	910.00	847.00	1140.00	881.00			
8/2/03								
Average	876.00	904.00	836.00	1070.00	870.00			
Std Dev	23.50	10.60	24.70	96.70	1.34			
Minimum	823.00	867.00	782.00	877.00	862.00			
Maximum	903.00	910.00	863.00	1140.00	876.00			
7/28 - 8/2								
Average	844.00	877.00	805.00	1100.00	786.00			
Std Dev	30.20	21.00	31.50	79.20	87.10			
Minimum	752.00	811.00	711.00	769.00	125.00			
Maximum	909.00	910.00	887.00	1180.00	883.00			

Table 6.4. Melter Kiln, Pour Spout, and Canister Oven Temperatures

6.2.2 Process Pressures

Melter and differential off-gas system pressures were recorded throughout the duration of the RSM test. Specifically, the process pressures recorded were

- plenum gauge pressure
- film-cooler pressure drop
- EVS pressure drop
- HEME pressure drop.

A graphical summary of these operating parameters extracted from the electronic process log is presented in Figure 6.5. The pressure-drop data for the EVS is not displayed because it did not produce any. With the low off-gas flow rate (~5 scfm) used during the current test, the EVS, apart from all other off-gas system devices, assisted the blower in maintaining the melter-plenum vacuum. Specifically, the operation of this device created aspiration conditions that help pump gas from the melter's plenum. Indeed, the EVS was so effective in pumping gas that it was fully capable of maintaining an ~2 wc" vacuum without any assistance from the blower.

The low off-gas flow rate also made it impossible for the automated off-gas valve to control the melter vacuum. Recognizing that automatic off-gas control based on plenum pressure was inoperable throughout the entire duration of RSM testing, it is apparent that vacuum control of the melter and its off-gas system was easily maintained throughout all phases of testing. Buildup of debris within the film-cooler posed the only challenge to the manual control system. However, the debris was easily removed by inserting a rod through the off-gas sampling port (see Figure 4.14). As is clear from the stable and nominally invariant film-cooler pressure-drop data, no permanent off-gas obstructions formed within this device during C-104/AY-101 melter flowsheet testing.

6.2.3 Melter Electrical Data

The RSM's electrodes, kiln, discharge, and pour-spout heating loads are all controlled by phase angle, silicon-controlled rectifiers (SCRs). The SCRs control the voltage going to the load and are capable of adjustments anywhere from zero to the full line voltage (120 V and 208 V).

Figure 6.6 graphically summarizes the time-dependent behavior of the electrode electrical parameters. Due to the glass bubblers that provided an additional heat-loss mechanism and a required 400°C plenum temperature that required inleakage cooling, much greater electrode amperage and associated power was required to maintain the glass pool at 1150°C than is normally observed in the absence of bubbling and higher plenum temperatures. The spikes in amperage and power that occurred periodically throughout the test resulted from the need to compensate for decreased overflow heating power that was used to increase the glass level in the melter. Bubbler action eliminated the RSM's normal cyclic pouring behavior; by freezing the overflow channel and building up a glass head, high volumetric glass flows could be stimulated. Electrode parameter plots that provide much more time-resolved detail are displayed in Appendix C for the interested reader.



Figure 6.5. Melter Vacuum and Off-Gas Pressure Drop Data



Figure 6.6. RSM Electrode Operating Characteristics

Table 6.5 summarizes the maximum, minimum, and average electrode operating parameters and their standard deviations during all processing periods (see Table 6.1). The average Joule-heating requirements of each batch in this table were used with corresponding average batch-feeding rates (see Table 6.1) to derive specific Joule-energy processing requirements for C-104/AY-101 feeds. These data are summarized in Table 6.6.

Typical specific energy requirements for unagitated slurry-fed Joule-heated ceramic melters range from 2 to 4 kW•h/kg of glass produced (Perez et al. 2001). The average energy requirements under all phases of RSM testing for vitrifying the nominally invariant C-104/AY-101 feed were found, without exception, to exceed this range. The use of bubblers is, of course, the reason for this result. Although increased glass-production rates result from the agitation created by the bubblers, there is also a very significant concomitant thermal energy loss rate to the melter plenum and off-gas system that has to be compensated for by increased Joule-heating requirements.

As mentioned earlier, the melter was significantly overfed during the initial shakedown testing phase, and since the glass rates are derived from feeding-rate data, the associated specific energy result is artificially low, totally misleading, and should be disregarded. Steady-state processing conditions were, however, achieved during all remaining testing phases as the similarity in their specific energy requirements attest. As a group, the first five test conditions are the most similar in energy requirements, exhibiting only a 2% variability. The final two test conditions, which involved maximum mercury concentration, exhibited consistently lower specific energy results within two discrete test-condition groups, the overall variability amongst all seven test conditions is only \sim 8%.

It should also be noted that in addition to the electrical power delivered to the melt pool, combustion energy was also being provided by the reductant feed-stream component. Although sugar was the most significant source of combustion energy, at an ~ 1 L/h feeding rate and a 5 g/L concentration, the combustion contributions to the vitrification energy requirements are seen to be trivially small. Although the RSM construction and design is hardly representative of ceramic-lined production melters, the energy expended to vitrify the C-104/AY-101 feeds is, nevertheless, consistent with generalized LFCM operating expectations, if bubbler energy loss rates are considered.

Like the melter electrodes, the RSM's kiln and overflow heater circuits were similarly characterized. Because of the relatively invariant nature of the electrical loads involved, these data do not contain much structure or embedded information and are therefore summarized on a daily basis in Table 6.7. As mentioned earlier, the bubblers delivered so much heat energy to the melter vessel that kiln power was not always required to maintain the melter canister's external temperature setpoint. The minimum overflow heater condition summarized in Table 6.7, as discussed earlier, resulted from attempts to generate large volumetric pours by freezing the overflow port in order to build up a glass inventory (head) in the melter.

		RSM Electrod	e Parameter	•
Date/Item	Volt	Amp	kVA	Ohm
Shake Down		1	<u>.</u>	<u>L</u>
Average	33.20	83.90	2.79	0.41
Std Dev	3.10	12.20	0.58	0.09
Min	29.00	49.90	1.68	0.35
Max	64.20	119.00	7.63	0.73
Lo/Lo/Lo				
Average	32.10	91.10	2.92	0.35
Std Dev	1.25	3.57	0.16	0.02
Min	28.30	71.90	2.05	0.32
Max	38.10	99.80	3.39	0.43
Lo/Hi/Lo				
Average	31.80	89.10	2.84	0.36
Std Dev	1.09	2.27	0.14	0.01
Min	28.40	80.40	2.37	0.34
Max	35.50	96.20	3.31	0.40
Lo/Hi/Hi				
Average	30.80	90.00	2.77	0.34
Std Dev	0.81	2.01	0.10	0.01
Min	28.00	83.80	2.38	0.32
Max	32.70	96.50	3.12	0.37
Hi/Lo/Lo				
Average	32.20	87.80	2.83	0.37
Std Dev	1.57	5.50	0.30	0.01
Min	28.00	74.80	2.18	0.31
Max	38.80	114.00	4.10	0.43
Hi/Hi/Lo				
Average	31.20	90.60	2.83	0.35
Std Dev	1.21	5.29	0.26	0.01
Min	28.10	76.80	2.18	0.31
Max	35.00	105.00	3.64	0.38
Mx/Hi/Lo				
Average	29.30	90.60	2.66	0.32
Std Dev	1.20	5.12	0.25	0.01
Min	25.20	80.00	2.09	0.29
Max	34.60	108.00	3.67	0.35
Mx/Hi/Hi				
Average	28.00	95.90	2.68	0.29
Std Dev	1.06	7.53	0.27	0.02
Min	21.50	83.30	1.96	0.24
Max	30.20	111.00	3.28	0.33
All Condition				
Average	31.20	90.10	2.81	0.35
Std Dev	1.93	5.98	0.28	0.03
Min	21.50	49.90	1.68	0.24
Max	64.20	119.00	7.63	0.73

 Table 6.5. RSM Electrode Circuit Operating Characteristics

		CI	D ((a)	G	Power			
	Test Condition	Gla	ss Rate ^w	Sucrose	(KW)		Spec Engy (kW	/-h/kg)
#	Hg/Cl/Red	(kg/h)	MT/d/m ²	$(g/L)_{Fd}$	Joule	Combust	Joule	Total
0	ShakeDwn	0.67	1.01	0.00	2.79	0.00	4.20	4.20
1	Lo/Lo/Lo	0.50	0.76	0.00	2.92	0.00	5.88	5.88
2	Lo/Hi/Lo	0.48	0.73	0.00	2.84	0.00	5.90	5.90
3	Lo/Hi/Hi	0.45	0.69	5.00	2.77	0.02	6.11	6.16
4	Hi/Lo/Lo	0.49	0.74	0.00	2.83	0.00	5.81	5.81
5	Hi/Hi/Lo	0.47	0.71	0.00	2.83	0.00	6.05	6.05
6	Mx/Hi/Lo	0.53	0.81	0.00	2.66	0.00	5.01	5.01
7	Mx/Hi/Hi	0.53	0.80	5.00	2.68	0.02	5.08	5.12
(8	a) Feed projection scale		Average ^(b) =	5.71				
(1	b) Condition 0 excluded	d.					Std Dev ^(b) =	7.9%

Table 6.6. Specific Process Energy Requirements for C-104/AY-101 Feeds

6.2.4 Bubbler Glass Agitation

Two ¹/₄-inch Inconel[®] tubes were inserted through the melter lid and into the glass pool and were used with a compressed air source to create forced convective mixing within the melter's glass crucible. The tubes, separated by 3 inches, are off centerline but collinear with the electrodes that are separated by ~6 inches. Because of the active-convective process created by this bubbler configuration, the minimum acceptable compressed-gas flow rate was used throughout all phases of testing. Figure 6.7 summarizes the operational characteristics of the bubbler system during RSM testing. The average rate achieved compares well with the minimum of the target flow range (1 to 2 scfm/m²).

6.2.5 Melter Off-Gas Temperature and Residence Time Characteristics

In order to create off-gas chemistry conditions that are representative of the WTP, the unquenched off-gas temperature and melter-to-quencher residence time had to be controlled. Off-gas line insulation and external heating were the primary parameters used to provide temperature control. The film-cooler airinjection rate was used to establish the proper flow-rate/residence-time conditions. While active off-gas temperature measurements were used to make appropriate off-gas line-heating adjustments, heliumdilution techniques were employed to measure off-gas flow rate and associated off-gas residence time in the melter's off-gas jumper. To determine the RSM's off-gas flow rate, a helium tracer was injected at a fixed flow rate (1 SL/min) into the RSM's film-cooler injection stream, and the resultant helium concentration was continuously sampled and measured downstream of the film cooler, after complete mixing with melter exhaust had occurred, using an on-line gas chromatograph. Since the chromatograph provides concentration results on a dry basis, these data were used with melter feeding rates and the physical properties of the feed (see Table 5.4) to establish the total steam-laden flow-rate and resultant residence-time value. An operational summary of unquenched melter off-gas temperatures and residence times occurring during all phases of RSM testing is graphically portrayed in Figure 6.8. A statistical summary of these and related flow-rate parameters for each of the melter operating test conditions is also detailed in Table 6.8.

	Kiln Ele	ctrical Par	ameter	OverFlow Heater Param		rameter
Date/Item	Volt	Amp	kVA	Volt	Amp	kVA
7/28/2003		•	•	<u></u>	•	-
Average	27.60	1.91	0.13	149.00	13.50	2.01
Std Dev	30.10	2.85	0.32	10.60	0.76	0.26
Min	2.50	0.00	0.00	129.00	11.90	1.54
Max	123.00	14.20	1.74	176.00	15.50	2.72
7/29/2003		•				
Average	2.84	0.00	0.00	170.00	14.80	2.52
Std Dev	0.31	0.00	0.00	11.50	0.73	0.29
Min	2.25	0.00	0.00	149.00	13.30	1.98
Max	3.50	0.00	0.00	181.00	15.60	2.84
7/30/2003		•		•	,	
Average	9.57	0.40	0.03	158.00	13.50	2.40
Std Dev	22.20	1.33	0.12	55.40	4.72	0.93
Min	2.25	0.00	0.00	3.00	0.00	0.00
Max	112.00	6.50	0.73	189.00	17.50	3.16
7/31/2003		•				
Average	3.14	0.01	0.00	158.00	13.60	2.35
Std Dev	1.85	0.12	0.00	48.60	4.21	0.76
Min	2.50	0.00	0.00	3.00	0.00	0.00
Max	23.30	1.34	0.03	181.00	17.80	3.22
8/1/2003						
Average	2.94	0.00	0.00	167.00	14.10	2.54
Std Dev	0.40	0.02	0.00	45.30	3.83	0.73
Min	2.25	0.00	0.00	2.75	0.00	0.00
Max	7.50	0.36	0.00	182.00	17.30	3.05
8/2/2003						
Average	31.30	1.90	0.22	154.00	13.20	2.22
Std Dev	48.50	3.35	0.45	49.30	3.84	0.94
Min	2.50	0.00	0.00	2.75	0.00	0.00
Max	161.00	11.70	1.88	181.00	16.90	3.06
7/28 - 8/02						
Average	8.77	0.42	0.03	161.00	13.90	2.39
Std Dev	20.40	1.55	0.17	41.90	3.53	0.71
Min	2.25	0.00	0.00	2.75	0.00	0.00
Max	161.00	14.20	1.88	189.00	17.80	3.22

 Table 6.7. Operational Characteristics of Melter Kiln and Overflow Heaters

	Melter Off-Gas Characteristics					
Test Condition		Flow Rate		Transit	Temp	
Hg/Cl/Red Conc	Dry, scfm	Wet, scfm	Wet, acfm	Time (s)	°C	
Shake Down						
Average	5.27	5.86	10.90	0.966	235	
Std. Dev.	1.28	1.28	2.21	0.196	7.9	
Minimum	4.09	4.68	8.83	0.786	227	
Maximum	6.45	7.04	12.90	1.150	243	
Melter Idle #1		Ł		ł		
Average	4.33	4.92	9.32	1.100	243	
Std. Dev.	0.51	0.51	1.02	0.120	10.4	
Minimum	3.50	4.09	7.81	0.890	232	
Maximum	5.20	5.79	11.40	1.300	268	
Low/Low/Low		2		•		
Average	5.50	6.09	11.00	0.927	221	
Std. Dev.	0.48	0.48	0.84	0.077	10.8	
Minimum	3.56	4.15	7.37	0.783	208	
Maximum	6.58	7.17	13.00	1.380	269	
Low/High/Low		2		•		
Average	5.48	6.07	11.50	0.884	244	
Std. Dev.	0.17	0.17	0.36	0.027	5.9	
Minimum	5.04	5.63	10.50	0.701	232	
Maximum	6.97	7.56	14.50	0.967	263	
Low/High/High		•		•		
Average	5.20	5.79	10.90	0.931	242	
Std. Dev.	0.24	0.24	0.49	0.043	5.4	
Minimum	4.45	5.05	9.33	0.852	231	
Maximum	5.67	6.26	11.90	1.090	252	
Melter Idle #2		-				
Average	4.72	4.74	8.79	1.150	234	
Std. Dev.	0.06	0.09	0.19	0.023	3.0	
Minimum	4.56	4.56	8.48	1.040	229	
Maximum	4.86	5.23	9.74	1.200	243	
High/Lo/Lo						
Average	5.09	5.68	10.40	0.981	228	
Std. Dev.	0.50	0.50	1.04	0.099	10.8	
Minimum	2.86	3.46	6.17	0.701	201	
Maximum	7.02	7.62	14.50	1.650	251	

 Table 6.8. Off-Gas Flow Rate, Residence Time, and Temperature Test-Condition Statistics
	Melter Off-Gas Characteristics							
Test Condition		Flow Rate		Transit	Temp			
Hg/Cl/Red Conc	Dry, scfm	Wet, scfm	Wet, acfm	Time (s)	°C			
High/High/Lo		-		-				
Average	4.96	5.55	10.00	1.010	219			
Std. Dev.	0.18	0.18	0.33	0.034	5.3			
Minimum	4.17	4.77	8.70	0.910	205			
Maximum	5.51	6.11	11.10	1.170	227			
Max/High/Lo								
Average	5.02	5.61	9.72	1.040	200			
Std. Dev.	0.12	0.12	0.26	0.028	7.0			
Minimum	4.47	5.06	8.90	0.929	182			
Maximum	5.84	6.43	10.90	1.140	213			
Max/High/High								
Average	5.01	5.61	9.68	1.050	199			
Std. Dev.	0.13	0.13	0.24	0.025	3.3			
Minimum	4.76	5.35	9.12	0.954	192			
Maximum	5.58	6.17	10.60	1.110	204			
All Conditions								
Average	5.18	5.76	10.50	0.976	224			
Std. Dev.	0.42	0.44	0.95	0.090	17.2			
Minimum	2.86	3.46	6.17	0.701	182			
Maximum	7.02	7.62	14.50	1.650	269			

Table 6.8 (Contd)

As the graphical data show, good control over both these experiment parameters was achieved throughout the melter testing campaign. Indeed, the target values for these parameters are well represented by the average values achieved.

6.2.6 EVS Condensate Tank, Film Cooler Injection Air

As described earlier, the EVS acts to both quench the melter exhaust stream and remove entrained debris generated by the melter source. As seen in Table 6.3, the EVS's scrubbing-liquor temperature appeared to remain fairly constant (~33°C), on the average, throughout RSM testing. However, as Figure 6.9 shows, the condensate temperature was measurably affected by diurnal ambient temperature variation throughout the test, despite the relatively constant temperature of the heat exchanger. Because of the relatively high temperature of this fluid, EVS steam-collection rates were quite low. Based on the average feeding rate of the melter and the feed properties described in Table 5.7, the 0.18 L/h represents 22% of the overall average rate to which water was fed to the melter.



Figure 6.7. Flow Rate Behavior of Glass Agitation Bubblers



Figure 6.8. Unquenched Off-Gas Temperatures and Residence Times During RSM Testing



Figure 6.9. EVS Condensate Tank Temperature and Volume

7.0 Melter Off-Gas Emission Characterization

Off-gas effluent studies were conducted during C-104/AY-101 flowsheet testing to characterize the melter-effluent source. As described in Section 4.1.4, the off-gas sampling network assembled in support of this objective was designed to determine the composition of the melter exhaust with regard to non-condensable (25°C) as well as condensable effluents.

The composition of melter-generated, non-condensable effluent emissions was established using the gas analyzers described in Section 4 of this report. The instruments used were designed to continuously (or quasi-continuously) monitor and record process-exhaust concentrations of H_2 , He, N_2 , CO, CO₂, NO, NO₂, O₂, SO₂, and THC. In addition to these conventionally applied analytical instruments, a continuousemission monitor for volatile forms of mercury was also integrated into the RSM's off-gas monitoring network. Discrete sampling for gaseous emissions, including semi-volatiles such as mercury as well as condensed-phase effluents, was also conducted as described below.

To characterize the melter source of gaseous and condensed-phase effluents, manual (40 CFR 60, Appendix A, Method 29) sampling trains composed of a high-efficiency (99.95% efficient for 0.3-µm aerosols) filter, condenser, and a series of chemically specific gas scrubbers were employed (see Section 4). The manner in which any given element is distributed across the various discrete sampling stages of this device allows the physical/chemical state or states assumed by this effluent species to be inferred. The Method 29 sampling train used in this study is designed specifically to quantify mercury emissions and to provide limited speciation information.

For reference, Table 7.1 summarizes the masses of test-parameter chemical additives processed during each of the seven test conditions evaluated using averaged test-segment operational conditions described in the previous Section of this report. Since mercury incorporation in glass has been shown to be negligibly small (Table 5.11), the test segment-dependent off-gas concentrations of mercury in all its forms (vapor and condensed) have been projected and are also included in Table 7.1.

The operational data and experimental results obtained from the melter off-gas studies conducted in support of the C-104/AY-101 flowsheet evaluations are discussed below.

									Wt%	, D	Mass	Pro	cessed		
Т	est Condition	Duration	Fee	ed	Glass ^(a)		(02	xide ba	asis)	112000	(g)		Mel	ter Off-Gas	
#	Hg/Cl/Red	(h)	kg	kg/h	kg	kg/h	MT/d/m ²	HgO	Cl	Sugar	Hg	Cl	Sugar	scfm	Hg(mg/m ³)
1	Lo/Lo/Lo	18.8	27.3	1.45	10.5	0.56	0.73	0.05	0.009	0.00	4.9	0.9	0.0	6.09	25.0
2	Lo/Hi/Lo	18.4	26.0	1.41	10.0	0.54	0.72	0.05	0.06	0.00	4.6	6.0	0.0	6.07	24.4
3	Lo/Hi/Hi	7.6	10.1	1.32	3.90	0.51	0.67	0.05	0.06	0.90	1.8	2.3	35.0	5.79	24.0
4	Hi/Lo/Lo	18.7	26.6	1.42	10.2	0.55	0.72	0.15	0.01	0.00	14.2	0.9	0.0	5.68	78.9
5	Hi/Hi/Lo	18.5	25.5	1.38	9.8	0.53	0.70	0.15	0.06	0.00	13.6	5.9	0.0	5.55	78.1
6	Mx/Hi/Lo	16.9	26.3	1.55	10.1	0.60	0.79	0.95	0.06	0.00	89.1	6.1	0.0	5.61	553.0
7	Mx/Hi/Hi	6.8	10.5	1.53	4.02	0.59	0.77	0.95	0.06	0.90	35.5	2.4	36.2	5.61	543.0
	Total/Average	105.8	152.2	1.44	58.6	0.55	0.73				163.7	24.6	71.2	5.77	
(a	() Calculated v	alues.													

 Table 7.1. Test-Segment Trace-Additive Summary

7.1 Flue-Gas Effluent

For the surrogate C-104/AY-101 melter feed used during the July/August 2003 test, CO_2 and NO_x (specifically NO) were the major non-condensable (~25°C) gases produced by the vitrification process. Table 7.2 summarizes the maximum, minimum, and average concentrations (by volume) of melter-generated gaseous effluents during the active processing periods for each discrete melter operating condition (see Table 6.1). Also presented in this table are the standard deviations associated with the concentrations listed.

For a non-condensable off-gas flow rate of ~5.5 scfm and a steady-state feeding rate of 1 L/h, the melter off-gas (MOG) concentrations of the major effluent gases (CO₂ and NO) were found to be, nominally, 0.27% and 0.01%, respectively. The combustible gas CO was barely detectable (<5 ppm), except when sugar was added to the feed, and H₂ was not detectable (<10 ppm) under any of the test processing conditions. Throughout all phases of processing, the CO concentration averaged only 2.7 ppm, and the highest sustained concentration recorded, 0.0023%, occurred during processing of feed containing 5 g of sugar/L of feed. These concentrations are well below the lower flammability limits of this combustible gas, 15.5% for CO.

Test Condition	1	Melter Ex	chaust	Gas Ce	oncentra	ation (p	(a)	
Hg/Cl/Red Conc	O ₂	CO ₂	CO	SO ₂	NO	NO ₂	THC	NO _x
Pretest Idle								
Average	211,000	1.79	0.00	1.19	0.01	0.00	0.07	0.00
Std. Dev.	171	10.90	0.00	0.61	0.10	0.00	0.17	0.00
Minimum	211,000	0.00	0.00	0.00	0.00	0.00	0.01	0.00
Maximum	211,000	79.40	0.00	2.56	2.04	0.00	0.97	0.00
Process Shake Down								
Average	209,000	3,860	1.00	0.03	164.0	9.0	0.12	172.0
Std. Dev.	888	1,520	1.41	0.18	42.9	7.8	0.08	45.1
Minimum	202,000	0.00	0.00	0.00	1.2	0.00	0.01	0.00
Maximum	211,000	11,800	3.54	1.48	270.0	30.2	0.86	293.0
Melter Idle #1								
Average	210,000	2,670	0.50	0.00	93.8	9.4	0.12	103.0
Std. Dev.	376	2,070	0.81	0.00	90.2	8.7	0.10	98.5
Minimum	209,000	273.00	0.00	0.00	5.6	0.00	0.00	5.6
Maximum	210,000	7,310	1.92	0.00	357.0	34.8	0.40	385.0
Low/Low/Low								
Average	210,000	2,420	0.59	0.38	102.0	10.1	0.09	112.0
Std. Dev.	255	1,060	1.28	0.69	35.7	3.8	0.17	38.5
Minimum	208,000	0.00	0.00	0.00	1.53	0.00	0.00	2.02
Maximum	210,000	5,230	5.27	3.41	215.0	48.3	1.38	239.0
(a) Dry concentration	except for	THC.					· · · · ·	

Table 7.2. Unquenched Melter Off-Gas Composition

Test Condition Melter Exhaust Gas Concentration (ppm)^(a) Hg/Cl/Red Conc NO \mathbf{O}_2 CO_2 CO SO₂ NO_2 THC NO_v Low/High/Low 210,000 3,850 1.32 0.14 111.0 9.6 0.19 121.0 Average Std. Dev. 574 1,310 1.80 0.38 35.5 3.3 0.25 38.0 209,000 348.00 0.00 Minimum 0.00 2.54 0.0 0.00 2.76 Maximum 211,000 6,040 7.61 2.06 278.0 19.7 1.55 293.0 Low/High/High 210,000 2,860 13.20 2.12 23.0 0.98 23.6 Average 0.7 790 Std. Dev. 232 2.83 1.16 11.5 1.2 0.34 12.5 209,000 1.25 Minimum 0.00 1.41 0.00 1.25 0.00 0.01 Maximum 211,000 5,240 23.20 4.05 107.0 9.9 2.71 115.0 Melter Idle #2 210,000 979 0.88 0.00 3.4 0.1 0.08 3.4 Average 815 0.00 0.2 0.37 747 3.42 6.8 6.7 Std. Dev. 209,000 0.00 0.00 0.90 Minimum 555.00 0.00 0.00 1.07 211,000 5,200 20.10 0.00 48.1 1.2 2.38 47.4 Maximum High/Lo/Lo Average 210,000 2,980 2.26 2.27 125.0 9.7 0.02 134.0 Std. Dev. 680 1,750 2.30 1.88 43.2 3.9 0.10 45.6 Minimum 209,000 0.00 0.00 0.00 1.33 0.00 0.00 1.37 6,590 Maximum 211,000 8.39 5.21 257.0 29.6 3.18 276.0 High/High/Lo 206,000 2,500 0.99 124.0 Average 3.37 10.7 0.04 135.0 Std. Dev. 3,040 940 2.30 1.12 34.9 3.5 0.46 37.0 192,000 0.00 2.1 0.00 3.2 Minimum 0 0.00 0.0 Maximum 212,000 4,880 23.30 3.60 275.0 27.0 14.40 291.0 Max/High/Lo 209,000 3,430 210.0 Average 0.62 1.10 16.6 0.00 226.0 Std. Dev. 265 927 0.68 1.25 53.7 4.9 0.01 56.4 Minimum 209,000 0.00 0.00 7.76 0.00 0.00 10.8 0 Maximum 210,000 8,050 4.34 3.80 476.0 49.3 0.01 489.0 Max/High/High 209,000 3,440 1.09 0.25 10.10 67.4 1.5 Average 68.4 Std. Dev. 233 940 3.30 0.37 26.2 2.9 0.19 28.3 Minimum 208,000 0 0.33 0.16 3.00 0.00 0.00 0.81 Maximum 210,000 5,700 14.90 1.88 228.0 30.4 1.17 239.0 All Conditions 209,000 2,800 2.70 1.00 110.0 0.14 119.0 Average 8.8 1,820 1,540 1.31 68.8 6.3 0.33 74.1 Std. Dev. 4.07

Table 7.2 (Contd)

0.00

23.30

0.00

5.21

0.00

476.0

0.00

49.3

0.00

14.40

0.00

489.0

0.00

11,800

192,000

212,000

(a) Dry concentration except for THA.

Minimum

Maximum

The time-dependent behavior of gaseous process effluent emissions was recorded at nominally 1-min intervals throughout the melter-processing campaign. Because steady-state feeding conditions were maintained throughout most phases of testing, the average process exhaust concentrations of these off-gas effluents remained relatively invariant. They were, however, perturbed by scheduled feed-batch preparations and changes in film-cooler injection rates as well as during feed sampling and/or feed system repair. Figure 7.1 graphically presents the temporal behavior of melter off-gas effluents on a daily basis.

The only distinctive feature in this graphical gaseous-effluent data that has not already been discussed is the apparent effectiveness of even small amounts of sugar upon the melter's NO_x source term. Only during the processing of feeds containing no sugar is NO a significant byproduct of melter nitrate destruction. These data suggest that the sugar-feed additive is effectively reducing the nitrate feed component to N₂. Throughout RSM testing, NO₂ (~ 8 ppm) was found to be an unimportant contributor to the melter's overall nitrogen oxide source term. Appendix C provides higher time-resolved plots of these off-gas data.

Table 7.3 compares actual NO_x and CO_x off-gas concentrations with calculated values based upon nitrogen oxy-anion and reductant feed composition, respectively, and off-gas flow-rate data. The impact of sugar upon apparent NO₂⁻/NO₃⁻ destruction is clearly shown by this NO_x comparative data. Nominally 21% destruction of the oxy-anions of nitrogen was observed in the absence of sugar while ~81% reduction to nitrogen is achieved when this reductant at a 5g/L loading is used.



Figure 7.1. Temporal Behavior of Major Process Effluent Gases

Test	Feeding Rate		Off-Gas	NO _x Conc.	(ppm)	Reduction	CO _x Conc.	(ppm)
Hg/Cl/Red	kg/h	L/h	Flw (scfm)	Calculated	Actual	To N_2 (%)	Calculated	Actual
ShakeDwn	2.00	1.39	5.27	199	172	14	64.9	3,860
Lo/Lo/Lo	1.49	1.03	5.50	149	112	25	46.4	2,420
Lo/Hi/Lo	1.45	1.00	5.48	145	121	16	45.1	3,850
Lo/Hi/Hi	1.36	0.94	5.20	144	24	84	464.0	2,880
Hi/Lo/Lo	1.47	1.01	5.09	171	134	22	49.1	2,980
Hi/Hi/Lo	1.41	0.98	4.96	169	135	20	48.5	2,510
Mx/Hi/Lo	1.60	1.11	5.02	309	226	27	54.3	3,430
Mx/Hi/Hi	1.59	1.10	5.01	308	68	78	561.0	3,450

Table 7.3. Actual Vs. Calculated CO_x and NO_x Process Off-Gas Concentration

The comparative CO_x data in Table 7.3 reveal an off-gas source of carbon oxides that is an order of magnitude greater than projections based upon both oxalate- and sugar-feed sources. The pretest monitoring data clearly show that an analyzer zero bias is not the source of this apparent anomaly. Although the two short RSM idling periods demonstrate a relationship between the unexpectedly high CO_x levels and processing, the impact of relatively low concentrations of sugar upon nitrogen oxy-anion destruction suggests a carbon source external to the feed stream. That is, the observed impact of sugar at 5 g/L-Fd upon both nitrogen-oxide reduction and glass-oxidation state is clearly inconsistent with a high extraneous source of carbon in the feed stream. New refractory material in the melter lid could be the source of carbonaceous material responsible for the abnormally high CO_x production. Assuming that process steam and carbonaceous lid materials were producing water-gas reaction products, then it must also be assumed that highly efficient plenum combustion of these reaction products also occurred as no hydrogen (<10 ppm) and very low concentrations of CO were present in the melter's process off-gas stream.

7.2 Mercury-Vapor Emissions

As described in Section 4.1.1.4, quasi-continuous off-gas monitoring of vapor-state mercury compounds was conducted in the exhaust stream of the melter, EVS, and HEME. Since melter glass exhibits no appreciable capacity for mercury, it was expected that nominally all mercury fed to the melter will be lost to the melter's off-gas system. Although mercury compounds are unstable at melter-glass temperatures (~1150°C), reactions in the plenum (~400°C) and unquenched (225°C) off-gas line can convert the thermally decomposed elemental mercury source to a variety of (oxidized mercury) compounds. Since some of these oxidized forms (compounds) of mercury (e.g., HgCl₂) exhibit vapor pressures similar to that of elemental mercury, characterizing the melter's mercury effluent source requires measuring volatile (elemental and oxidized) as well as non-volatile (chemically combined) species in the process exhaust stream. Nonvolatile mercury partitioning results will be separately discussed with condensed-phase effluents.

7.2.1 Thermodynamic Model of Mercury Speciation

In a previous, but similar, melter study (Goles et al. 2002), efforts to shed light on the mercury speciation in a typical melter exhaust stream produced the thermodynamic equilibrium model predictions shown in Figure 7.2. These model predictions were found to be largely invariant over a rather large range of oxidizing and generic off-gas conditions. Since the model presumes an equilibrium condition, it serves only to indicate what mercury speciation is possible and which species are most probable, independent of kinetic considerations. Specifically, the model results indicate that at typical unquenched off-gas processing conditions, the predominant mercury species predicted, under favorable kinetic conditions, is HgCl₂. In support of earlier statements, elemental mercury is also found to be the predominate species predicted above ~500°C.

7.2.2 CEM Mercury Measurements

Two separate sample-stream conditioning systems were used to support process off-gas monitoring objectives. The primary system exclusively sampled the melter source while the secondary system was used to selectively monitor the EVS or HEME exhaust sampling sites (see Figure 4.9). Since both systems shared a common detection module, simultaneous results could not be obtained from the primary and secondary sampling sites. Table 7.4 summarizes the monitoring results obtained for each of the test conditions evaluated during the C-104/AY-101 vitrification flowsheet study.

This table presents the average total and elemental gaseous concentrations of mercury in the melter, EVS, and HEME exhausts as a function of test constituents. Also presented, for each testing condition, are the corresponding standard deviation and the minimum and maximum values of the concentrations recorded. The number of observations used to derive these numbers, which are also detailed, clearly show that an emphasis was placed on collecting mercury-species information from the primary melter-exhaust sampling site.



Figure 7.2. Thermodynamic Predictions of Speciation of Mercury Compounds

The standard deviations associated with these tabular data suggest a very large spread in recorded results, especially for the low-mercury-concentration conditions. The nature of these highly dispersed data is graphically summarized in Appendix C where all test-condition data are appropriately grouped and displayed. Subsequent discussions of these data will reveal that instrumental difficulties compromised the quantitative value of much of the mercury CEM data collected. As a result, average sample site concentration results are somewhat inconsistent and should not be directly compared. Indeed, comparison of Table 7.4 average concentration data with processing expectation detailed in Table 7.1 suggests a very significant quantitative problem, unless significant condensed-phase mercury effluent is being generated, which is not likely as will be shown in Section 7.3. In order to further clarify the functional dependence of the mercury off-gas source term upon test conditions, the average (relative) mercury concentration data for the melter, EVS, and HEME sampling sites have been graphically combined and summarized in Figure 7.3 through Figure 7.5, respectively.

The melter exhaust data in Figure 7.3 suggest that under low mercury feed-concentration conditions, chlorine-feed loading may significantly increase partitioning to oxidized forms of mercury (i.e., HgCl₂). When the high-feed concentration of mercury was employed, overall mercury-vapor concentrations increased in a manner consistent with the corresponding feed-concentration change (\sim 3×). As in the low-mercury-feed case, an increase in feed halogen content was found to enhance the relative yield of the oxidized mercury, which was further increased when the mercury-feed content was maximized at fixed chlorine content. However, there are strong suggestions that this latter result may not be reliable, as no significant increase in total mercury-vapor concentrations accompanied the \sim 6-x change in feed composition. In addition, unreasonable elemental to total-mercury ratios (Hg°/Hg>1) occurred with increasing frequency as is typified by the results of the seventh test condition.

Unlike the melter source term, the mercury vapor in the exhaust of the EVS was found, with the exception of the Hi/Lo/Lo condition, to be dominated by the unoxidized form of the element. Although the magnitude of mercury-vapor concentration increased significantly when the mercury-feed-stream concentration was maximized, meaningless elemental to total-mercury ratios (>1) were also consistently recorded.

Significantly fewer mercury-vapor measurements were made downstream of the HEME than occurred at the Melter and EVS sampling sites. No observations were, in fact, conducted at the HEME sampling site under the low-mercury-feed condition. However, like the EVS result, a systematic increase in mercury-vapor concentration accompanied the Hi to Mx mercury-feed-composition change, and absurd elemental-to-total-mercury ratios were also observed under maximum mercury-processing conditions. The greater than unity average elemental-to-total-mercury ratio observed under the Hi mercury-feed conditions is in part a consequence of the small number of observations averaged.

It should be pointed out that the concentrations in Table 7.4 and in Figure 7.3 through Figure 7.5 are average results of, in general, noncontemporaneous measurements. As a result, meaningful comparisons of averaged elemental and oxidized concentrations assume stable operating and concomitant off-gas processing conditions, which, for the current test, is a valid supposition. However, it also has to be assumed that the mercury off-gas behavior is functionally and reproducibly related to the controlled process parameters.

	Merc	Mercury Species Exhaust Concentration (mg/m ³)								
	Melt	ter	EV	S	HEN	ЛE				
Date/Item	Element	Total	Element	Total	Element	Total				
ShakeDwn										
Average	2.60	2.70	0.20	0.14						
Std Dev	0.36	0.45	0.09	0.01						
Min	2.40	2.40	0.13	0.13						
Max	2.90	3.20	0.26	0.15						
#Obs.	2	3	2	2						
Idle #1										
Average	0.50		0.11	0.07						
Std Dev	0.61		0.12	0.12						
Min										
Max	1.20		0.24	0.20						
#Obs.	4	2	3	3						
Lo/Lo/Lo										
Average	0.46	0.45	0.35	0.11						
Std Dev	0.89	0.93	0.35	0.23						
Min										
Max	5.40	5.50	1.20	0.84						
#Obs.	50	55	25	25						
Lo/Hi/Lo										
Average	0.68	3.10	0.05	0.04						
Std Dev	1.70	2.20	0.06	0.04						
Min			0.00							
Max	14.00	17.00	0.21	0.19						
#Obs.	130	120	19	22						
Lo/Hi/Hi										
Average	2.20	3.70	0.92	0.97						
Std Dev	1.20	1.00	0.53	0.60						
Min		1.30		0.03						
Max	7.70	7.90	1.40	1.50						
#Obs.	66	61	6	8						
Idle #2										
Average	0.55	2.70								
Std Dev	0.59	0.95								
Min	0.23	2.00								
Max	3.20	6.20								
#Obs.	24	28								
Hi/Lo/Lo										
Average	9.60	10.00	3.10	5.20	5.00	6.10				
Std Dev	1.90	1.90	2.40	2.60	0.59	2.30				
Min	3.00	5.70	0.51	0.48	4.60	1.60				
Max	15.00	15.00	7.40	9.20	6.40	7.40				
#Obs.	130	130	20	33	8	6				

 Table 7.4. Volatile Mercury Concentration During RSM Test Conditions

	Merc	cury Speci	ies Exhaust	Concent	ration (mg/	m ³)
	Melt	ter	EV	S	HEN	1E
Date/Item	Element	Total	Element	Total	Element	Total
Hi/Hi/Lo		-		-	-	
Average	8.10	8.90	4.70	4.70	7.80	4.00
Std Dev	1.70	1.10	2.10	2.30	0.80	3.30
Min	3.60	6.00	1.20	1.10	7.00	0.95
Max	15.00	12.00	10.00	7.10	8.60	8.70
#Obs.	140	140	24	10	3	13
Mx/Hi/Lo			• •			
Average	6.60	9.60	21.00	15.00	14.00	14.00
Std Dev	2.70	2.80		2.10	4.40	1.90
Min	0.37	0.17	21.00	9.50	7.30	11.00
Max	22.00	16.00	21.00	18.00	20.00	18.00
#Obs.	110	110	1	54	15	10
Mx/Hi/Hi			•			-
Average	8.20	5.70	14.00	8.00	12.00	10.00
Std Dev	4.00	2.00	2.50	4.10	0.48	1.30
Min	0.12	0.10	7.70	0.08	11.00	9.40
Max	16.00	8.70	19.00	15.00	13.00	12.00
#Obs.	29	29	23	14	8	3
All Cond.			• •			
Average	5.30	6.60	4.30	6.70	11.00	8.20
Std Dev	4.20	4.00	5.40	6.50	4.60	5.20
Min					4.60	0.95
Max	22.00	17.00	21.00	18.00	20.00	18.00
#Obs.	680	670	120	170	34	32

 Table 7.4 (Contd)

In an attempt to further characterize the occurrence and frequency of incongruous instrument responses (e.g., $Hg^{\circ}/Hg>1$), quasi-contemporaneous data pairs for each test condition at each sampling location were examined. The results of this evaluation are summarized in Table 7.5, assuming a 20% tolerance, because the measurements, although closely spaced, are nevertheless sequential. For the melter sampling site, the highest frequency of errant data generation occurred during the first (Lo/Lo/Lo) and seventh (Mx/Hi/Hi) test conditions, suggesting that unreliable information was being generated during these testing phases. The conclusion regarding the quality of EVS and HEME data based on this errant frequency criteria is less clear cut as the populations of these time-paired data are too small to be reliably used.

Nevertheless, these tabular results in combination with the high temporal variability of the recorded concentration values suggest that significant instrument-stability problems may be affecting the quality of the data generated. The time-dependent data collected from the melter sampling site shown in Figure 7.6 may help illustrate this point.



Figure 7.3. Mercury-Vapor Concentration in the Unquenched Melter Exhaust



Figure 7.4. Mercury-Vapor Concentration in the EVS Exhaust



Figure 7.5. Mercury-Vapor Concentration in the HEME Exhaust

Specifically, the instrument response accompanying a step change in reductant feed concentration at ~06:00 on 7/30/03 appears to be a definite cause-and-affect correlation. Indeed, the instrument also appeared to respond appropriately to a feeding interruption at ~09:00. However, after feeding was resumed and steady-state conditions were re-established, the instrument responded differently than before the feed outage. On the basis of systematic arguments, this suggests nonreproducible results, although there could conceivably be unknown factors responsible for the observed changes in process chemistry.

	Melter: (l	Hg°/Hg)	EVS: (H	(g°/Hg)	HEME: ()	Hg°/Hg)
Condition	Observ #	% >1.2	Observ #	% >1.2	Observ #	% >1.2
Lo/Lo/Lo	23	65	14	64	0	
Lo/Hi/Lo	42	2	3	33	0	
Lo/Hi/Hi	20	0	0		0	
Hi/Lo/Lo	43	2	1	100	1	100
Hi/Hi/Lo	45	4	2	50	0	
Mx/Hi/Lo	62	13	1	100	8	0
Mx/Hi/Hi	28	71	1	100	1	0

Table 7.5. Paired Elemental and Total-Mercury-Vapor Measurements



Figure 7.6. Mercury CEM Response to a Step Change in Feed Composition

The time-dependent variability (Std. Dev) in the data collected during each test condition summarized in Table 7.4 and graphically portrayed in Appendix C further erodes confidence in the quantitative validity of much of the CEM data collected since stable processing conditions existed throughout essentially all phases of testing. The only exceptions to the above characterization occurred during the Hi/Lo/Lo and Hi/Hi/Lo test segments where instrument responses were long-term stable and reproducible but were nevertheless significantly below engineering expectations (see Table 7.1). Although certain aspects of Hg CEM data compare favorably with off-gas sampling and secondary-waste-stream analytical results to be subsequently discussed, these latter results, as will be shown, undercut the overall quantitative validity of the quasi real-time mercury-monitoring data.

As will be further discussed in Section 8.5, inadequate sample-stream dilution is likely responsible for instrument condensation losses and resultant low concentration-related responses. This unfortunate circumstance, if true, should not, however, appreciably affect Hg:HgCl₂ ratio measurements, as these species exhibit very similar vapor-pressure characteristics.

7.3 Condensed-Phase Effluents

The effluents entering the MOG system that require long-term environmental isolation are primarily condensed-phase matter, i.e., aerosols. Many feed components are volatilized to some extent within the melter; however, rapid condensation in the melter plenum transforms most of these effluent vapors to airborne aerosols before they can be carried into the off-gas system. Feed and/or glass matter can also be physically ejected into the melter-plenum volume by cold-cap and/or glass-surface turbulence. Once in the plenum, this debris can become entrained in gas currents and exhausted from the melter as entrained

particulate matter. Both of these loss mechanisms produce off-gas system aerosols; however, the physical characteristics and chemical composition of these two types of airborne matter are markedly different.

Entrained aerosols typically have a mass-median diameter of » 1 µm and are compositionally similar to the feed. Consequently, entrainment losses, to first approximation, will influence all feed components in the same way. Feed constituents that fume at melter vitrification temperatures, e.g., alkali halides, quickly form condensation aerosols, which are predominantly submicron and are chemically dissimilar to the bulk feed. The importance of the volatilization/condensation loss mechanism is totally dependent upon the physical and chemical properties of the feed components and the range of compounds they can form. Consequently, melter aerosol loss rates will be exacerbated by the presence of semi-volatile feed components, and effluent emission rates of elements capable of forming semi-volatile compounds will always be greater than those elements only capable of forming refractory compounds. Effluent loss rates are traditionally expressed in terms of equipment DFs. A device DF value for a particular feed component is derived by taking the ratio of the rate at which the component enters the device to the rate at which it exits. Aerosol DFs are partial DFs that relate to only one off-gas effluent form: aerosols.

7.3.1 Aerosol Mass DFs

The melter's aerosol mass DFs, as measured by the filter catches of the Method 29, differential sampler previously described, are tabulated in Table 7.6 for each distinct sample taken during RSM testing. Sampling data were taken during four of the seven test conditions previously discussed in Section 2 of this report. The four conditions sampled were considered to be most likely to reveal any differences in off-gas mercury chemistry related to the test's independent parameters.

These melter aerosol loss data summarized in Table 7.6 represent a 0.6% to 2.5% mass partitioning to the off-gas system, which is entirely consistent with previous small-scale melter flowsheet tests. Also listed in this table are related melter DFs and off-gas aerosol loading values. Since the processing data presented in Section 6.1 clearly demonstrated that uniform feeding and concomitant stable, steady-state cold-cap conditions were achieved throughout RSM testing, the variability of these condensed-phase melter loss data is indicative of the significant range over which stable melter processing conditions can be achieved.

It should be noted that quasi isokinetic sampling conditions were achieved during RSM particulate sampling campaigns. Consequently, a fairly representative sample of off-gas aerosols should have been collected by the sampling filters. With the caveat that sampling was conducted in a small 2-in.-OD pipe, 96% isokinetic conditions, on the average, were maintained throughout all sampling campaigns. The relative proportions of fuming to entrained effluents will be examined in the following section.

Date/	Date/Time Proc. Cond.		Sampling		MOG Flw Aeros		sol Catch		lelter	
Start	Stop	#	Hg/Cl/Red	Time (m)	Flw (slpm)	(scfm)	Mass (g)	Con (mg/L)	DF	Loss%
07/30/03 00:04	07/30/03 01:59	2	Lo/Hi/Lo	115	15.7	6.2	1.020	0.562	100	1.00
07/30/03 23:19	07/31/03 01:19	4	Hi/Lo/Lo	120	17.4	5.8	1.480	0.710	87	1.15
07/31/03 23:32	08/01/03 01:32	5	Hi/Hi/Lo	120	15.9	5.5	2.760	1.450	40	2.51
08/02/03 00:42	08/02/03 02:42	7	Mx/Hi/Hi	120	15.8	5.6	0.808	0.427	161	0.62

Table 7.6. Gross Melter Aerosol Emission Characteristics

7.3.2 Aerosol Elemental DFs

Individual melter-aerosol DFs have been calculated for all melter-feed components using the compositional data derived from off-gas filter samples. A comparison of these filter compositional data with the oxide-feed target values is shown in Table 7.7. It must be pointed out that the silica content of the Method 29 aerosol samples does not include likely sizable contributions from the filter material, as the quartz filter substrate was digested with the aerosol material it contained. As a result, the wt% values of all effluent constituents are artificially elevated relative to the corresponding feed data also appearing in Table 7.7. To establish a clearer basis for comparing the aerosol samples with the baseline feed composition, the sample data in Table 7.7 have been normalized to iron—a classic nonvolatile feed constituent. These results, detailed in Table 7.8, show that the concentration of the classic semivolatiles (e.g., B, Cs, Cr) are modestly enriched over their nominal feed-composition values. As expected, the non-fuming components, such as Ca, La, Mn, and Zn, are seen to be present at much lower to nominally equivalent feed concentrations. The high aluminum ratio in these samples, as discussed previously, is due to its higher than target concentration in the feed. The general compositional characteristics of the effluents listed in Table 7.8 are in total conformity with generalized LFCM effluent-emission expectations developed from past melter-source-term characterization studies.

Using target feed-composition and physical-property information provided in Section 5, melter DFs associated with aerosol loss for individual elements can be calculated for the constituents listed in Table 7.7. These derived DF values are tabulated in Table 7.9. These tabular results reinforce the previous discussion that predicted low DFs for feed constituents that are volatile or can form volatile or semivolatile compounds at melter-processing temperatures and higher DFs for those that cannot.

Although enhanced relative to refractory, non volatile components, the Cs losses recorded, as predicted earlier from glass-composition results, are significantly lower than nominal expectations (DF=10). On the other hand, the vanishingly low glass concentrations of mercury and its relatively high aerosol DF suggest high volatility losses for this element, especially for the Mx/Hi/Hi test condition.

The reproducibility of melter-feed component DFs is, overall, quite good. Furthermore, the magnitudes of the DF values listed in Table 7.9 are well within expectations and are quite representative of average melter-performance behavior.

7.3.3 Volatile Partitioning and Total Elemental DFs

Since only a very few feed components are lost to the off-gas processing system in the gaseous state, essentially all the aerosol performance values listed in Table 7.9 also represent total melter DFs for these elements. Notable exceptions to this statement include B, Cl, F, Hg, P, and S, whose volatility usually dominates melter off-gas system losses and determines their melter DFs. However, because the methodology and chemical solutions employed in Method 29 sampling precluded halide analyses in the various sample fractions, only B, Hg, P, and any volatilized metals (e.g., Cr) could be effectively tracked. Table 7.10 summarizes the overall cumulative elemental mass collected by the Method 29's impinger train for effluents penetrating the upstream aerosol filter of the differential sampling system previously described (see Section 4.1.4.2). With the exception of B, Hg, and possibly Cr, these data show little evidence of other gas-phase effluents. The other cations in these impinger solutions are an artifact of reagent blank correction uncertainties.

		Hg/Cl/Red Test Condition Aerosol Compositions (Wt%)										
	Lo/Hi/Lo Hi/Lo/Lo Hi/Hi/Lo							li/Hi				
Oxide	Aerosol	Feed	Aerosol	Feed	Aerosol	Feed	Aerosol	Feed				
Al ₂ O ₃	8.140	3.580	7.840	3.580	8.100	3.570	7.800	3.540				
B_2O_3	27.400	10.800	27.000	10.800	24.900	10.800	22.100	10.700				
CaO	0.499	0.480	0.485	0.479	0.457	0.479	0.505	0.475				
Cl		0.060		0.009		0.060		0.060				
Cr ₂ O ₃	0.145	0.060	0.155	0.060	0.138	0.060	0.127	0.059				
Cs ₂ O	0.191	0.050	0.195	0.050	0.164	0.050	0.198	0.050				
CuO 0.035 0.030 0.034 0.030 0.032 0.030 0.051 0.030												
F		0.120		0.120		0.120		0.119				
Fe ₂ O ₃	13.900	9.530	13.500	9.530	16.000	9.520	13.900	9.450				
HgO	0.257	0.050	0.331	0.150	0.170	0.150	1.300	0.950				
Ι		0.100		0.100		0.100		0.099				
La ₂ O ₃	0.253	0.150	0.235	0.150	0.247	0.150	0.273	0.149				
Li ₂ O	2.950	3.310	4.430	3.310	4.510	3.300	4.300	3.280				
MnO	1.650	1.520	1.810	1.520	2.550	1.520	1.660	1.500				
Na ₂ O	26.600	11.500	24.900	11.500	24.300	11.500	22.800	11.400				
Nd ₂ O ₃		0.110		0.110		0.110		0.109				
NiO	0.657	0.470	0.762	0.469	0.757	0.469	0.702	0.465				
P ₂ O ₅	0.222	0.040	0.237	0.040	0.240	0.040	0.144	0.040				
PbO	0.255	0.120	0.206	0.120	0.196	0.120	0.281	0.119				
SiO ₂ ^(a)	5.100	46.300	3.080	46.300	2.390	46.300	4.410	45.900				
SrO	0.003		0.004		0.005		0.005					
TiO ₂	0.094	0.020	0.114	0.020	0.138	0.020	0.098	0.020				
ZnO	3.720	2.160	3.440	2.160	3.460	2.160	3.350	2.140				
ZrO ₂	ZrO ₂ 7.950 9.420 11.200 9.420 11.300 9.410 16.000 9.340											
(a) Gre	ater/equal v	value				•						

 Table 7.7. Oxide Composition of Melter-Generated Aerosols and Melter Feed

	Aerosol	Element Ra	atio For Hg	/Cl/Red Fd E	Batches	Baseline
Element	Lo/Hi/Lo	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Hi	Average	Feed
Al	0.443	0.439	0.384	0.424	0.422	0.284
В	0.873	0.888	0.693	0.704	0.789	0.503
Ca	0.037	0.037	0.029	0.037	0.035	0.051
Cl						
Cr	0.010	0.011	0.008	0.009	0.010	0.006
Cs	0.019	0.020	0.014	0.019	0.018	0.007
Cu	0.003	0.003	0.002	0.004	0.003	0.004
F						0.018
Fe	1.000	1.000	1.000	1.000	1.000	1.000
Hg	0.025	0.032	0.014	0.124	0.049	
Ι						0.015
La	0.022	0.021	0.019	0.024	0.022	0.019
Li	0.141	0.218	0.188	0.205	0.188	0.230
Mn	0.132	0.149	0.177	0.132	0.147	0.176
Na	2.030	1.960	1.610	1.740	1.830	1.280
Nd						0.014
Ni	0.053	0.063	0.053	0.057	0.057	0.055
Pb	0.021	0.023	0.020	0.014	0.020	0.017
Р	0.012	0.010	0.008	0.013	0.010	0.003
Si	0.245	0.152	0.100	0.212	0.177	3.250
Sr	0.0002	0.0003	0.0004	0.0005	0.0004	
Ti	0.006	0.007	0.007	0.006	0.007	0.002
Zn	0.307	0.292	0.249	0.277	0.281	0.260
Zr	0.605	0.876	0.749	1.220	0.861	1.050

 Table 7.8. Normalized Oxide Composition of Melter-Generated Aerosols

As was done for particulate matter, volatile melter DFs, which are partial DFs relating to only volatile melter losses, can be derived by combining the meaningful condensate data with actual feed-compositional and physical-property information previously discussed in Section 5. These volatile DFs are summarized in Table 7.11.

As discussed above, the only constituents with any significant volatility in this list are Hg, B, and possibly Cr. The low volatile DFs determined for Hg were expected, and this behavior was quite reproducible, except possibly for the Lo/Hi/Lo sampling period during which time a significant proportion (11 wt%) of effluent mercury was associated with condensed phase (particulate) matter. The volatile DFs determined for boron were found to be consistent throughout all phases of testing, and their magnitudes were quite comparable with previous melter results.

By combining the aerosol-filter and corresponding impinger sampler fractions for condensed and condensable effluents, total melter DFs for all feed constituents detected in the process exhaust can be derived. Table 7.12 summarizes total elemental melter DFs measured during the processing of C-104/AY-101 waste-feed simulant. Comparison of these total values with corresponding particulate DFs previously discussed (Table 7.9) clearly illustrate the dominant role played by the aerosol loss mechanism. Volatility contributions to the melter-effluent source term were only significant for boron

and mercury, although halogen behavior, as discussed earlier, could not be determined; all other total DFs are nominally equivalent to their particulate values.

	Melter Aerosol DF For Hg/Cl/Red Feed Batch										
Element	Lo/Hi/Lo	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Hi	Ave ^(a)						
Al	70	70	31	139	59						
В	63	61	31	149	56						
Са	153	150	74	289	132						
Cl											
Cr	66	59	31	143	56						
Cs	42	39	21	77	37						
Cu	137	133	65	179	112						
F											
Fe	109	107	42	208	85						
Hg	31	69	62	224	59						
Ι											
La	94	97	43	167	80						
Li	178	114	52		89						
Mn	146	127	42	209	92						
Na	69	70	33		51						
Nd				2							
Ni	114	94	44	203	85						
Pb	86	77	35	254	70						
Р	25	30	14	43	24						
Si											
Sr											
Ti	34	27	10	62	22						
Zn	92	96	44	196	81						
Zr 188 128 59 179 112											
(a) From	average parti	itioning valu	es.								

Table 7.9. Elemental Melter DFs Associated With Aerosol Emissions

These element-specific, total DFs, like their corresponding particulate values, are reasonably close to general expectations and are with few exceptions consistent with previous RSM melter-testing results. Mercury, however, is an exception to this statement. Although fairly low DFs (<2) were recorded for all but the Lo/Hi/Lo test condition, the general expectation as validated by the glass data presented in Section 5.4.2, is that a DF of ~1 should have been obtained. At melter and off-gas operating temperatures, no significant loss mechanism between the melter and the EVS can be identified that might be responsible for the greater-than-unity DFs measured. Analytical losses, unrepresentative sampling, and/or analytical uncertainties are no doubt factors responsible for the higher-than-expected DF results obtained.

	Imping	ger Train Ir	ntegral Cato	ch (mg)	
Element	Lo/Hi/Lo	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Hi	Average
Al	0.195	0.243		0.200	0.211
В	12.300	11.000	14.600	18.700	13.600
Ca	0.074	0.199	0.237	0.064	0.104
Cr	0.046				0.046
Cu				0.015	0.015
Fe	0.078	0.144	0.072	0.610	0.114
Hg	10.200	98.500	72.700	539.000	32.400
Li				16.600	16.600
Na	2.480	2.140	2.380	0.362	0.987
Si		0.095	0.020	0.160	0.046
Zn		0.069		0.025	0.037

Table 7.10. Off-Gas Sampler Impinger Solution Composition

Table 7.11. Volatile Melter DFs from Impinger Solution Data

	Mel	ter Volatile	DF For Hg/	'Cl/Red Fd I	Batch
Element	Lo/Hi/Lo	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Hi	Average ^(a)
Al	8,500	8,640		10,500	8,570
В	239	338	217	198	238
Са	4,080	1,910	1,370	5,890	2,400
Cr	781				781
Fe	74,700	51,400	87,200	12,100	31,500
Hg	4	2	2	2	2
Na	3,010	4,420	3,390	26,000	4,480
Si		252,000	1,000,000	150,000	258,000
Zn		27,800		76,700	40,800
(a) From	average part	itioning valu	ies.		

Apart from generating volatility DF information, the Method 29 sampling train, as previously described (Section 4.1.4.2), is also capable of providing mercury-speciation information as well. Table 7.13 summarizes the distribution of mercury effluent across the filter and sequential H_2O_2 and KMnO₄ chemically selective gas-scrubber traps. These data show that the majority of the total Hg (94% on average) was captured by the initial H_2O_2/HNO_3 gas-scrubber impingers, suggesting, at first glance, that most of the total mercury existed as a chemically combined volatile (e.g., HgCl₂) at the film-cooler outlet. However, this is only a correct interpretation if the elemental mercury-vapor dew point of the influent stream is below the gas scrubber's bath temperature (0°C). For the current test, this condition was never satisfied. Table 7.14 projects the mercury distribution across the Method 29 sampling components if the mercury influent source was totally in its elemental state. Since the model used assumes equilibrated vapor-pressure conditions and neglects condensed phase-carryover effects, the KMnO₄ projections represent minimum expectation values.

		Total Melter DF								
Element	Lo/Hi/Lo	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Hi	Average ^(a)					
Al	69.3	68.9	31.1	138.0	58.5					
В	49.7	51.6	26.8	85.0	45.1					
Са	147.0	139.0	70.1	275.0	125.0					
Cl										
Cr	60.6	58.9	30.6	143.0	54.7					
Cs	41.6	38.9	21.4	76.6	36.5					
Cu	137.0	133.0	65.4	163.0	110.0					
F										
Fe	109.0	107.0	42.0	205.0	84.7					
Hg	3.5	1.5	1.8	1.8	1.9					
Ι										
La	94.3	97.1	42.8	167.0	79.6					
Li	178.0	114.0	51.6	71.2	83.6					
Mn	53.5	37.0	13.1	98.2	30.3					
Na	67.0	68.9	32.9	152.0	60.3					
Nd										
Ni	114.0	93.8	43.6	203.0	84.6					
Pb	85.8	76.8	35.2	254.0	70.2					
Р	24.9	29.5	14.3	43.2	24.0					
Si		2,270.0	1,360.0	3,130.0	2,010.0					
Sr										
Ti	33.7	26.7	10.2	62.1	22.0					
Zn	92.3	95.1	43.9	195.0	81.2					
Zr	188.0	128.0	58.7	179.0	112.0					
(a) From	average parti	tioning value	es.							

 Table 7.12. Total Individual Elemental Melter DF Values

Comparing Table 7.13 results with the Table 7.14 projections, the absence of significant mercury in the KMnO₄ catch during the Lo/Hi/Lo test condition suggests a predominantly chemically-combined mercury effluent source, which is in total conformity with the mercury CEM results discussed earlier (see Section 7.2.2). For the Hi-Hg/Lo-Cl test condition, the results suggest a mercury-effluent source dominated by the elemental form, which, again, is in total conformity with the previously discussed CEM data. On the other hand, a mixed volatile mercury source is suggested during the Hi-Hg/Hi-Cl test condition, as the proportion of mercury trapped in the final KMnO₄ impinger solutions was below the minimum projected value for a pure Hg° source. The Hg CEM data taken during this test period suggests, on the average, that 90% of the volatile mercury was in its elemental state. The sample-train distribution of mercury under the Mx-Hg/Hi-Cl test condition appears to have been influenced by breakthrough and/or Hg-mist carryover effects and as a consequence provides no useful speciation insights.

	% Collected By Device							
Test Cond	Filter	H_2O_2	KMnO ₄					
Lo/Hi/Lo	11.400	88.600	0.003					
Hi/Lo/Lo	2.220	94.800	2.950					
Hi/Hi/Lo	2.830	96.200	0.990					
Mx/Hi/Hi	0.806	98.000	1.160					

 Table 7.13. Distribution of Mercury Effluent Across Sampler Components

Table 7.14	Projected	Elemental	Mercury	Distribution A	Across Sample	Components
-------------------	-----------	-----------	---------	-----------------------	---------------	-------------------

	Hg° Ten	Hg° Temp Partitioning ^(a) (%)							
Test Cond	Filter	H_2O_2	KMnO ₄						
Lo/Hi/Lo	0.0	91.2	8.8						
Hi/Lo/Lo	0.0	97.1	2.9						
Hi/Hi/Lo	0.0	97.1	2.9						
Mx/Hi/Hi	0.0	99.5	0.5						
(a) Filter 15	(a) Filter 150°C; Impingers 0°C								

Additional effluent speciation insights can sometimes be inferred from the off-gas behavior and resultant distribution of effluents in successive off-gas emission abatement devices. Indeed, estimates of total melter, elemental DFs can also be extracted from the off-gas system and secondary-waste-stream-composition data, provided the volatiles are efficiently captured by emission-abatement equipment and that the resultant equipment effluent steams can be representatively sampled. These off-gas-related insights and source-emission estimates will be discussed in the following Section.

7.4 Process-Waste-Stream Composition

The RSM's off-gas system, which is composed of an EVS quench scrubber, a HEME (deep-bed regenerable filter), and a HEPA filter, acts like a multi-component sampler for process aerosols and condensable and/or reactive effluent gases (see Section 4.1.3). The EVS function is to quench and condition the hot melter-exhaust stream for subsequent cleanup stages. As such, it removes entrained (large-diameter) debris and condenses steam and other chemically reactive and/or condensable gases that are generated during LFCM processing. Boron, chlorine, fluorine, phosphorus, and sulfur feed components are all volatilized to some extent during LFCM processing, and some of the volatile species are efficiently removed (physically and/or chemically) by the aqueous off-gas system quencher—in this case, the EVS. The HEME serves to efficiently demist the EVS's exhaust stream and to remove small-diameter aerosols (<1 μ m) that successfully penetrate the upstream EVS. However, because of its high surface area and the moisture content of the influent stream, it can also serve as a fairly efficient aqueous contactor for reactive/condensable gases. The HEPA is a polishing filter that removes any remaining off-gas aerosols before the off-gas is released to the environment.

An aqueous secondary waste stream is generated by both the EVS and HEME off-gas processing devices, although the extent of HEME runoff can be significantly influenced by temperature and dew-point considerations. In the case of the EVS, both soluble and insoluble matter that accumulates within this

device's condensate tank is readily collectable. The waste stream generated by the HEME, on the other hand, is composed primarily of soluble-effluent species that dissolve in, usually, a highly acidic oxidizing aqueous media. Because of its extremely high surface area, the HEME provides efficient contacting between acid gasses (e.g., NO_x , SO_x , and halogens) and the mist-generated aqueous run-off. Insoluble solids collected on the HEME are not directly recoverable, as is the case with the HEPA filter, which is itself a secondary waste. Consequently, overall melter DFs can be approximated from the compositions and flow rates of the secondary waste streams generated by the EVS and HEME.

7.4.1 EVS Condensate Composition

Although UDS were not collected from the EVS's condensate tank until the RSM test was complete, EVS supernatant was sampled throughout the duration of melter flowsheet testing. The time-dependent cation composition of this EVS scrubbing liquor during RSM testing is summarized in Table 7.15 while condensate physical properties (pH and density) and anion composition are detailed in Table 7.16. The fact that only universally soluble species (e.g., halogens, nitrates, and alkali-earths boron) exhibit a strongly increasing concentration while insolubles (e.g., Fe, Mg, Mn, and Si) do not is a reflection of the presence of these latter constituents in the UDS. The time-dependent, monotonically increasing soluble anions concentrations are graphically shown in Figure 7.7 where the dramatic increases in time are amplified by a low-steam-condensing efficiency (22%) of the EVS due to high operating temperatures (~33°C) as discussed in Section 6.2.6.

Also presented in this graphical display is the time- and test-dependent behavior of soluble mercury. Clearly, the most dramatic aspect of the mercury data presented is the step change in soluble mercury concentration that was associated with the 6-h, high-reductant-processing campaign—the Lo/Hi/Hi test condition. Unfortunately, the magnitude of this accumulation is totally inconsistent with the quantity of mercury processed during this 7.5-h test period. The only possible, nontrivial explanation for such a short-term injection of the significant quantities of mercury required to produce the observed concentration change is to assume that a long-term mercury-accumulation source in the EVS inlet port suddenly collapsed and was transferred to the condensate tank by the EVS water jet.

Since the EVS' inlet port provides a 200°C to 30°C temperature transition region that can influence condensable gases, selective condensation and resultant accumulation of volatile, chemically-combined mercury compounds is a completely plausible circumstance. Indeed, the composition of pipe deposits to be subsequently discussed corroborates that vapor deposition at the EVS' inlet is an observable fact. Assuming that the mercury injection mechanism as described above is true, the EVS condensate result suggests that chemically combined mercury dominated the melter mercury source term during the initial three test conditions. Furthermore, the declining mercury concentrations for all subsequent samples suggest an EVS process chemistry that slowly converts soluble mercury into insoluble sludge, and a melter source that, under all but Lo-Hg test conditions, is generating predominantly insoluble forms of mercury effluent. Although the spike result could also be explained by analytical contamination and/or error, the systematic and dramatic change in the mercury content of condensate samples collected before and after the large injection cannot be adequately explained without the presence of such an event.

		EVS Cond	ensate Conc	entration a	t Test-Cond	ition Conclu	usion (mg/L)	
Element	ShkDwn	Lo/Lo/Lo	Lo/Hi/Lo	Lo/Hi/Hi	Hi/Lo/Lo	Hi/Hi/Lo	Mx/Hi/Lo	Mx/Hi/Hi
В	31.00	120.00	190.00	210.00	270.00	320.00	370.00	375.00
Ca	18.00	6.20	1.40	0.90	0.41	0.18	0.46	0.44
Cl	8.77	16.80	31.70	38.60	49.70	64.80	78.10	86.40
Cr	0.30	0.93	1.20	1.40	1.50	1.70	1.60	1.00
Cs		1.40	1.90	2.30	2.70	2.90	3.10	3.30
F	20.60	55.40	82.50	95.50	126.00	154.00	176.00	183.00
Fe		0.06	1.70	0.26	0.38	0.45	0.60	0.40
Hg	0.01	6.00	5.50	175.00	110.00	49.00	31.00	35.00
Ι	11.50	13.80	23.00	31.30	34.50	33.30	5.75	5.75
Li	6.40	26.00	40.00	45.00	56.00	73.00	79.00	80.00
Na	62.00	220.00	330.00	370.00	470.00	600.00	640.00	660.00
Ni							0.07	
Si	5.10	6.20	6.40	6.80	7.00	6.90	7.50	6.70
Sr	0.80	0.18	0.03	0.03				
Zn	0.07	0.06	0.09	0.12	0.15	0.19	0.30	0.26
Zr		0.06			0.16	0.20	0.28	0.27
Vol (L)	90	95	95	97	100	106	110	113

 Table 7.15. EVS Condensate Soluble Effluent Composition

Table 7.16. EVS Condensate Soluble-Anion Concentrations

				Concentration (mg/L)								
Sample ID	Hg/Cl/Red	pН	SpG	Cl ⁻¹	NO2 ⁻¹	Br ⁻¹	NO3 ⁻¹	PO ₄ ⁻³	SO_4^{-2}	$C_2O_4^{-2}$	I ⁻¹	F ⁻¹
RSM-Hg-9	ShkDwn	7.9	1.00	8.77	3.91	< 1	26.4	< 7	19.7	< 2.4	< 7	20.5
RSM-Hg-25	Lo/Lo/Lo	8.1	1.00	16.80	16.20	< 0.5	23.6	< 3.5	36.8	0.783	< 3.5	55.3
RSM-Hg-39	Lo/Hi/Lo	8.3	1.00	31.70	25.00	< 0.5	24.8	< 3.5	47.4	0.602	< 3.5	82.2
RSM-Hg-44	Lo/Hi/Hi	8.3	1.00	38.60	24.00	< 0.5	24.6	< 3.5	56.1	0.783	< 3.5	95.2
RSM-Hg-59	Hi/Lo/Lo	8.4	1.00	49.70	33.60	< 1	26.6	< 7	68.6	1.200	< 7	126.0
RSM-Hg-74	Hi/Hi/Lo	8.4	1.00	64.80	39.80	< 1	27.2	< 7	75.5	1.810	< 7	154.0
RSM-Hg-86A	Mx/Hi/Lo	8.4	1.00	78.10	47.10	< 1	29.0	< 7	79.4	1.200	< 7	176.0
RSM-Hg-89	Mx/Hi/Hi	8.4	1.00	86.40	63.00	< 2	32.1	< 14	86.1	1.450	< 14	183.0
RSM-Hg-100	#N/A	8.4	1.00	41.00	29.40	< 1	20.6	< 7	43.1	< 2.5	< 7	75.7

EVS UDS, as mentioned above, were harvested from the condensate tank at the conclusion of testing. These solids were concentrated, sampled, dried, weighed, and homogenized before being sent to the analytical laboratory. The compositional results obtained from two independent UDS samples are summarized in Table 7.17. The results reveal a very significant inventory of accumulated mercury, as well as classically insoluble feed constituents, e.g., Fe, Si, and Zr. Table 7.18 summarizes the elemental distributions of all detected EVS effluent species between the soluble aqueous phase and the UDS. In agreement with the inferences of graphical data previously discussed, the EVS' soluble inventory of mercury at the conclusion of testing represents only ~9% of the total mercury collected by this device.



Figure 7.7. EVS Condensate Anion Concentrations at the Conclusion of Sequential Test Conditions

7.4.2 Overall Melter DF

As mentioned earlier, total melter, elemental DFs can also be estimated from the off-gas system and secondary-waste-stream-composition data, provided the volatiles are efficiently captured by emission-abatement equipment and that the resultant equipment effluent steams can be representatively sampled. Using the EVS waste-stream composition, volumes and/or masses with melter feed composition, and physical properties in combination with processing-history information, overall average melter DFs have been derived for the C-104/AY-101 flowsheet test. Table 7.19 presents these DF approximations and compares them to the reference average values derived from the off-gas sampling campaigns.

These data show that both off-gas sampling and secondary-waste-stream derived DF values exhibit the same relative trends for related groups of elements and are of remarkably comparable magnitudes for most corresponding waste constituents. Not surprisingly, the greatest differences between these data sets are associated with volatile constituents that present unique and significant EVS collection difficulties. In particular, the $\sim 2\times$ difference in sampling and EVS mercury DFs suggest a likely significant accumulation of mercury in the downstream HEME, which will be subsequently discussed. The general agreement between multiple off-gas sampling results, taken during ~ 2 -h periods, and cumulative overall test DF results further corroborates that stable operating conditions prevailed throughout most phases of this melter-flowsheet-evaluation study. Furthermore, the overall element-specific DFs recorded using both approaches are all reasonably close to general expectations.

	EVS UI	DS Wt%	Average			
Element	RSM-96A	RSM-96B	(Wt%)	(g)		
Al	3.520	3.410	3.470	26.200		
В	0.712	0.703	0.707	5.350		
Ca	0.637	0.717	0.677	5.120		
Cl						
Cr	0.086	0.080	0.083	0.627		
Cs	0.042	0.042	0.042	0.320		
Cu	0.039	0.038	0.039	0.292		
F						
Fe	9.610	9.190	9.400	71.100		
Hg	4.910	5.930	5.420	41.000		
Ι						
La	0.121	0.122	0.122	0.921		
Li	0.177	0.530	0.353	2.670		
Mn	0.918	0.917	0.917	6.940		
Na	1.500	1.720	1.610	12.200		
Nd						
Ni	0.527	0.445	0.486	3.670		
Pb	0.142	0.147	0.145	1.090		
Р	0.101	0.089	0.095	0.718		
Si	15.000	14.900	15.000	113.000		
Sr	0.063	0.109	0.086	0.648		
Ti	0.095	0.095	0.095	0.717		
Zn	1.540	1.530	1.530	11.600		
Zr	9.620	10.600	10.100	76.300		
]	Fotal EVS U	DS Mass	756.000		

Table 7.17. Composition of EVS Undissolved Solids

7.4.3 HEME Effluent Catch

Although a HEME aqueous runoff stream is normally expected, ambient temperatures during the test period created atypical high operating temperatures that precluded aqueous accumulations in the HEME's sump drain. However, the HEME filter element and its containment were soaked in a single 5 M nitric acid bath over a period of weeks in order to dissolve and subsequently account for any accumulations of mercury. Since the HEME had been exposed to a similar, nonmercury-containing effluent source previous to the current test, only mercury that is unique to the current test could be defensibly quantified. However, sampling of the HEME leachate solution was complex, as the filter media retained a significant mass of leachate (sponge) solution that was significantly higher in mercury concentration than the free-acid solution surrounding it. Assuming that a representative sample of the sponge leachate solution contained within the filter was successfully collected, the leachate solution analyses indicated that a very significant fraction of the mercury processed during RSM testing penetrated the EVS and accumulated within the HEME. Specifically, ~70% of the mercury fed to the melter was found to reside in the HEME. The overall distribution of mercury in the RSM's secondary waste streams will be further discussed in Section 8.2.

	EVS F	Effluent M	ass (g)	
Element	UDS	Soluble	Total	UDS %
Al	26.20		26.20	
В	5.35	41.20	46.60	11.50
Ca	5.12	0.05	5.17	99.10
Cl		9.50	9.50	
Cr	0.63	0.11	0.74	85.10
Cs	0.32	0.36	0.68	46.80
Cu	0.29		0.29	
F		20.10	20.10	
Fe	71.10	0.04	71.10	99.90
Hg	41.00	3.85	44.80	91.40
Ι		0.63	0.63	
La	0.92		0.92	
Li	2.67	8.80	11.50	23.30
Mn	6.94		6.94	
Na	12.20	72.60	84.70	14.40
Nd				
Ni	3.67		3.67	
Pb	1.09		1.09	
Р	0.72		0.72	
Si	113.00	0.74	114.00	99.40
Sr	0.65		0.65	
Ti	0.72		0.72	
Zn	11.60	0.03	11.60	99.80
Zr	76.30	0.03	76.40	100.00

Table 7.18.EVS Waste-Stream EffluentDistribution

Table 7.19. Off-Gas Sampler and Waste-StreamTotal Melter DF Values

	Me	ter DF
Element	EVS	Sampler
Al	46.5	58.5
В	46.4	45.1
Ca	42.7	125.0
Cl	2.7	
Cr	35.8	54.7
Cs	44.5	36.5
Cu	52.8	110.0
F	3.8	
Fe	60.4	84.7
Hg	3.8	2.0
Ι	102.0	
La	89.3	79.6
Li	86.2	88.3
Mn	109.0	
Na	64.7	63.9
Nd		2.2
Ni	64.7	84.6
Pb	65.5	70.2
Р	15.6	24.0
Si	122.0	
Sr		
Ti	10.8	22.0
Zn	95.9	81.2
Zr	58.8	112.0

7.4.4 Off-Gas Line Effluent Accumulations

To complete the off-gas characterization, off-gas solid debris had to be accounted for. To accomplish this, the off-gas system was disassembled, and all pipe deposits were collected, ball-mill homogenized, sampled, and analyzed, as discussed in Section 7.4.1. Significant pipe deposits were only found in pipe sections between the melter and the EVS quench scrubber. The location and masses of these deposits are detailed in Section 8.1. The elemental compositions of the above-described UDS and pipe deposits are summarized in Table 7.20.

The mercury distribution in these samples demonstrates the effect of mercury-vapor deposition in pipe segments undergoing a significant temperature transition with the coolest location having the highest pipe-solids mercury concentration. Unlike mercury volatiles that remain gaseous at off-gas temperatures (~200°C), fuming alkali salts like CsCl will condense and/accrete on all available surfaces. Not surprisingly then, the pipe segments closest to the melter source possess the highest concentration of these semivolatiles.

Comparing the average compositions of the EVS's UDS and the pipe deposits also summarized in Table 7.20, it is fairly clear that the off-gas deposits are representative of material ordinarily collected by the EVS. Since the pipe deposits possess higher concentrations of nominally soluble constituents, the overall wt% of insoluble constituents appears to be biased low as a result. For completeness, the soluble anionic constituents of the pipe deposits are summarized in Table 7.21. Comparison to similar feed data described in Table 5.9 clearly suggests a much higher oxide content in the pipe-deposit materials. Furthermore, a very low soluble $F^-:CI^-$ ratio in these solids when compared to the condensate (see Table 7.16) suggest that the melter's fluorine effluent source term may be predominantly gaseous, i.e., HF and/or F₂. The properties and magnitudes of the secondary waste stream discussed above are summarized in Table 7.22.

		Off-Gas	Pipe Deposits	(Wt%)		Average		
Element	Flm-Cool	Elbow	Horizontal	Reducer	EVS	Pipe Dep	UDS	
Al	2.510	2.010	2.440	2.540	2.720	2.450	3.470	
В	2.680	2.640	3.260	4.100	3.010	3.140	0.707	
Ca	0.279	0.241	0.272	0.277	0.492	0.312	0.677	
Cl	2.190	0.325	0.296	0.633	0.586	0.806		
Cr	0.347	0.044	0.053	0.062	0.065	0.114	0.083	
Cs	0.103	0.020	0.050	0.038	0.033	0.049	0.042	
Cu	0.028	0.015	0.023	0.026	0.026	0.024	0.039	
F	0.001	0.000	0.000	0.001	0.001	0.001		
Fe	7.130	4.890	5.980	5.410	5.660	5.810	9.400	
Hg	0.090	0.140	0.091	0.426	1.650	0.479	5.420	
Ι								
La	0.100	0.097	0.112	0.114	0.112	0.107	0.122	
Li	0.851	0.710	0.808	0.826	0.920	0.823	0.353	
Mn	1.330	1.510	1.610	1.160	1.230	1.370	0.917	
Na	6.950	6.350	7.110	8.640	5.720	6.960	1.610	
Nd								
Ni	0.969	0.264	0.363	0.351	0.361	0.462	0.486	
Pb	0.301	0.127	0.182	0.147	0.149	0.181	0.145	
Р	0.109	0.118	0.083	0.110	0.083	0.100	0.095	
Si	17.600	17.700	19.900	15.300	15.600	17.200	15.000	
Sr	0.013	0.019	0.010	0.050	0.047	0.028	0.086	
Ti	0.076	0.059	0.073	0.073	0.082	0.073	0.095	
Zn	1.100	0.971	1.180	1.120	1.170	1.110	1.530	
Zr	8.570	7.950	8.950	9.570	10.100	9.030	10.100	

Table 7.20. RSM Off-Gas Pipe Deposit Composition

		Solids Anion Concentration (mg/g)								
Sample ID	OG Location	Cl ⁻¹	NO2 ⁻¹	Br ⁻¹	NO3 ⁻¹	PO ₄ -3	SO4 ⁻²	$C_2O_4^{-2}$	I ⁻¹	F ⁻¹
RSM-Hg-94A	FC Inlet	21.90	0.221	< 0.04	0.710	< 0.3	28.60	0.120	< 0.3	0.0058
RSM-Hg-94B	90° Sweep	3.25	0.609	0.00	3.380	< 0.3	0.0	0.153	< 0.3	0.0040
RSM-Hg-94C	Horiz. Pipe	2.96	0.537	0.00	2.230	< 0.3	0.0	0.224	< 0.3	0.0039
RSM-Hg-94D	2-in. to 3-in. Trans.	6.33	1.620	0.00	2.540	< 0.3	0.0	0.314	< 0.3	0.0055
RSM-Hg-94E	EVS Inlet	5.86	2.370	0.00	1.890	< 0.3	0.0	0.199	< 0.3	0.0102

 Table 7.21. Pipe Solids Soluble Anion Composition

Waste Stream	Unit	Quantity
Condensate	L	110
UDS	g	756
Pipe Deposits	g	489

8.0 Byproducts, Residuals, Mass Balance, and Volume Reduction

Beyond the analytical characterization of feed, glass, off-gas effluents, and process-system waste streams previously discussed, a physical accounting of off-gas pipe deposits, process products, off-gas generated waste streams and mercury partitioning during condensate evaporation has also been conducted. The results of these evaluations as well as an overall process summary assessment are discussed below.

8.1 Off-Gas Line Deposits

At the conclusion of RSM testing, the off-gas line from the melter to the EVS was disassembled, inspected, and sampled. Figure 8.1 is a schematic of this segmented off-gas jumper that may be useful in providing perspective for understanding the interrelationship of these pipe segments and the samples obtained from them. Due to the low film-cooler injection rates during the current test, the cumulative accumulations within this off-gas jumper were found to be significantly (~25 x) higher than had been previously observed under similar feed-processing conditions (Goles et al. 2002). The manner in which the deposits were distributed within this jumper is summarized in Table 8.1.

Although these data suggest that film-cooler accumulations were minimal, in fact, the inlet section of this device had to be manually cleaned (rodded out) several times throughout testing. As a result, the reported mass of deposited material within this device actually represents only about 12 hours of accumulation. If one ignores film-cooler accretions, the greatest accumulations of deposited material occurred in the 90° elbow where physical impaction enhanced overall losses of entrained debris and in the 8-ft-long horizontal off-gas line segment where settling created the largest overall source of off-gas pipe deposits. All other pipe-accumulation sites accounted for the remaining 25% of collected deposits. Photographs of accumulated debris within the pipe segments identified in Figure 8.1 are displayed in Appendix D. As is clear from these photographs, the deposits that accumulated did not significantly influence the internal geometry of the pipe sections, except for the film-cooler, as discussed earlier. In total, about a half a kilogram of material accumulated within the entire run of this off-gas line jumper over the ~114-h processing period. This represents about 65% of the UDS collected in the EVS' condensate tank.

As mentioned earlier, all harvested pipe deposits were individually analyzed. The resultant analytical data were summarized and previously discussed in Section 7.4.4.

	Deposit Mass		
Pipe Segment	(g)	(%)	
Film-Cooler	5.73	1.17	
Elbow	157.00	32.10	
Horizontal	201.00	41.20	
Reducer	90.90	18.60	
EVS Throat	33.50	6.86	
Total	489.00		

 Table 8.1. Distribution of Melter Off-Gas Line Deposits



Figure 8.1. Melter Off-Gas Jumper Configuration and Sample Site Locations

8.2 Process Mass Balance

The compositional data of process streams previously discussed and summarized were combined in an attempt to fully characterize the C-104/AY-101 vitrification process flowsheet. Average analytical feed/ glass/off-gas compositional data were exclusively used for this purpose. The mass-balance results for the overall test are summarized in Table 8.2. Recognizing the limitations imposed by analytical uncertainties, very reasonable mass closures are demonstrated for most of the feed constituents for which complete analytical data exist. The partial data existing for the volatile halogen feed constituents suggests a low overall melter DF (\sim 1.1) for F, as discussed in Section 7.4, and an apparent accounting problem for Cl. Since the feed compositional data for chlorine represent only the soluble fraction of this element, the apparent over recovery of this element suggested by Table 8.2 is a reflection of the lower bound feed-concentration value used in this evaluation.

		EVS Condensate				
Element	Glass	Soluble	UDS	Pipe Dep.	HEME	Total
Al	96.3		1.8	0.8		98.9
В	87.4	2.0	0.3	0.8		90.4
Ca	109.0	0.0	2.5	0.7		112.0
Cl		110.0		23.1		133.0
Cr	140.0	0.4	2.2	1.0		143.0
Cs	87.0	1.8	1.6	0.9		91.3
Cu	62.8		1.3	0.5		64.5
F		92.2		0.0		92.2
Fe	91.0	0.0	1.8	0.7		93.5
Hg	0.0	2.3	24.7	0.8	70.3	98.2
Ι						0.0
La	89.2		1.4	0.8		91.4
Li	77.3	0.8	0.3	0.4		78.7
Mn	82.2		0.9	0.9		84.0
Na	83.6	1.3	0.2	0.6		85.7
Nd						0.0
Ni	97.8		1.6	0.7		100.0
Pb	95.3		1.6	1.1		97.9
Р	106.0		1.2	0.8		108.0
Si	91.1	0.0	0.9	0.7		92.6
Sr	612.0		17.5	3.0		633.0
Ti	101.0		1.7	0.8		104.0
Zn	89.6	0.0	1.6	0.7		92.0
Zr	87.0	0.0	1.6	0.9		89.5

 Table 8.2. Stream-Dependent %-Partitioning of C-104/AY-101 Melter-Feed Constituents

The other elements exhibiting poor mass closure but possessing complete analytical results include Cr, Cu, Li, and Sr. Declared analytical difficulties are responsible for the poor Li mass-balance results, while the results associated with the trace constituents Cr, Cu, and Sr were limited by instrumental analytical-detection sensitivities. Recorded mass deficits, especially for the alkali elements, are, to some extent, a result of the fact that the HEME waste-stream contributions to the process mass balance, as discussed in Section 7.4.3, could only be characterized for mercury.

The mass-balance result recorded for mercury is essentially identical to that obtained during a previous melter flowsheet test involving another simulated U.S. Department of Energy (DOE) tank waste (Goles et al. 2002). Specifically, 70% of the mercury processed penetrated the off-gas system quench scrubber and accumulated within the HEME's deep-bed filter. Of the 27% that accumulated in the quench-scrubber's condensate tank, only about 9% of the quench-scrubber's inventory was soluble.

8.3 Vitrification Process Statistics

During the July 2003 melter-flowsheet evaluation studies, 116 L (31 gal) of 30 wt% C-104/AY-101 simulated melter feed, having a total mass of 168 kg, were processed by the RSM, producing 22 L (5.7 gal) of glass having a total mass of 56 kg. Although vitrification results in both mass and volume waste reductions, only the volume-reduction parameter is meaningful since a major mass contributor to the waste (H₂O) is a nonvitrifiable, volatile effluent. On the other hand, since most of the hazardous and

rad-waste components can be incorporated and immobilized in the melter's vitreous product and tankwaste volumes are a physical reality, volume reduction has important waste-disposal implications. During the current RSM test, an overall C-104/AY-101 feed-volume reduction factor of 4.7 was achieved. If one assumes a ~15% waste-to-feed-volume expansion factor, a 4.1 waste-volume reduction factor is suggested.

8.4 Mercury Partitioning Under Vacuum-Evaporator Concentration of Melter-Generated Condensate

Under WTP operations, aqueous secondary waste from HLW processing will be transferred to the pretreatment facility where it will be concentrated. Since the species of mercury present in this waste stream may influence the manner in which mercury partitions to HLW, LAW, LERF/ETF, and off-gas streams, quench-scrubber condensate generated during RSM testing was concentrated using a laboratory-scale evaporator, described in Section 4.3.

Evaporator operating parameters were chosen conservatively from acceptable design ranges (Table 4.3) that apply to the Hanford Site's 242-A-Evaporator and by extension to the WTP evaporator design specifications. However, because the WTP's pretreatment evaporators are expected to operate at a higher pH range, duplicate evaporator tests were conducted at pH=8.4 and pH=13.0 that are consistent with current Site and projected WTP evaporator operating conditions, respectively. Beyond pH constraints, the target temperature, pressure, and flow parameters adopted for the current evaporator test are summarized in Table 8.3 along with the actual values achieved during both testing campaigns.

8.4.1 Experimental Observations and Results

The laboratory-scale evaporator, described in Section 4.3 and pictorially depicted in Appendix A, was used to concentrate a representative sample of the melter-generated, mercury-containing, quench-scrubber condensate at initial feed pH conditions of 8.4 and 13.

8.4.1.1 Testing at pH=8.4

During the initial evaporator test conducted at reference 242-A Evaporator conditions, a 1.7-L sample of melter-generated condensate containing UDS was concentrated by a factor of 1.67 (40.2% volume reduction) over an ~17-h period using a vacuum evaporator under nominally prototypic Table 8.3 conditions. Except for the concentration factor, all major operating test parameters were met and maintained throughout the evaporator experiment. Although a 2× concentration factor was sought, based upon optimum operation of site evaporators, a 40% boil down is well within the concentration design operation range of 20 to 80% for the evaporators.

The test overall was very successful. As mentioned above, 40.2% of the original feed volume was evaporated. Of this, 98.4% was collected in the condensate collection vessel, and 99.0% of the evaporated material was collected before the chemical scrubber solutions. Only \sim 1% of the material evaporated could not be accounted for in the condensate fractions collected at the conclusion of testing. A mass-balance of the pH=8.4 evaporator test is summarized in Table 8.4.

		Testing		
Stream	Basis Data	Target	Achieved	
Feed Temperature	18 – 49°C	49°C	pH=8.4	pH=13.0
Slurry Temperature (242-A Evaporator)	18 – 66°C	± 10°C	47–49°C	48–49°C
Reflux Condenser 1 Simulates de-entrainment pads (242-A Evaporator)	Same nominal temperature as evaporator pot – 49°C	49°C ± 10°C	47–49°C	49°C
Condensers #2 and #3 Simulates primary condenser (242-A Evaporator)	Raw Water – Skin Temp 2 – 24°C Material out usually 38°C	30°C	30°C	30°C
Purge Flow (Scc/min)	Air in leakage (scaled from SRTC small scale evaporation testing)	8–9	8.75	8.50
Vacuum (242-A Evaporator)	As close to 60 torr (current operation) as can be achieved with equipment available	60–100 Torr	71–90 Torr mid 70s to lower 80s typical	67–85 Torr mid 70s to lower 80s typical

 Table 8.3. Evaporator Design and Test Conditions

By far the most significant potential mass loss was material adhering to the surface areas of the condensers and all the connecting glassware and tubing. There was visible moisture on some parts at breakdown, even though the condenser temperatures were adjusted for the last 30 minutes, as part of the shutdown procedure, to try and dry these out. None of this equipment was tared independently, and therefore this material cannot be accounted for in the overall mass balance.

Based on the material balance, it appears that most of the moisture collected in the Drierite columns came from the last chemical impinger.

Vessel	Tare	Material Added g	Starting Wts g	Final Wts g	Gained/Lost g
4-L Reaction	1401	1752.9	3153.9	2454	-699.88
Condensate Collection	299.7	0	299.7	988.5	688.80
Method 29 Train Vessel 1	523	0	523	526.8	3.80
Method 29 Train Vessel 2	545.3	215.8	761.1	760.2	-0.90
Method 29 Train Vessel 3	510.2	236.1	746.3	744.5	-1.80
Method 29 Train Vessel 4	507.6	0	507.6	507.7	0.10
Method 29 Train Vessel 5	438.1	230	668.1	666.3	-1.80
Method 29 Train Vessel 6	442.4	228.7	671.1	640.1	-31.00
Drierite Desiccant #1			913.7	936.4	22.70
Drierite Desiccant #2			1108.1	1117.9	9.80
		Material evaporated (g)			699.88
		% collected at 30°C			98.42
Material collected before chemical impingers (g)			692.60		
% collected before chemical impingers			98.96		
Impinger #2 - #6 accumulation (g)			-7.28		
Water mass balance (g)			-10.18		

Table 8.4. Mercury Partitioning Study, Mass Balance (pH=8.4)

8.4.1.2 Testing at pH=13

During the evaporator test conducted at projected WTP pre-treatment evaporation conditions, a 2.2-L sample of melter-generated condensate containing UDS was concentrated by a factor of 2.7 over an ~4-h period using a vacuum evaporator under the nominally prototypic conditions summarized in Table 8.3. All major operating test parameters were met and maintained throughout the evaporator experiment. The significantly reduced time required to perform the second evaporation campaign resulted from the much better vacuum conditions achieved during the second test. The operating pressure of the evaporator system during the second test was dominated by the partial pressure of water vapor, which apparently was not true during the first campaign.

The test overall was very successful. As mentioned above, 62% of the original feed volume was evaporated. Of this, 81% was collected in the condensate collection vessel, and 94% of the evaporated material was collected before the chemical scrubber solutions. Less than 0.1% of the system's solution mass could not be accounted for at the conclusion of testing. A mass-balance of the pH=13 evaporator test is summarized in Table 8.5.

The lower collection efficiency of the primary condensate collection vessel observed during the second evaporator test was a reflection of the much greater evaporation rates achieved during this test relative to the first campaign, as previously discussed. This higher evaporation rate led to significantly higher collected condensate temperatures that averaged 37°C, even though the condenser's heat exchange fluid was maintained at 30°C. Although higher than the 30°C temperatures observed during the initial pH=8.4 test, the average condensate temperature achieved during the second pH=13 test was quite prototypic of the reference 242-A Evaporator condensate condition (38°C) discussed in Section 4.3.2.
		Μ	lass in Grams		
Vessel Description	Tare Wt.	Material Added	Starting Wt.	Final Wt	Gained/Lost
4-liter reaction	2953.9	2176.4	5130.3	3775	-1355.3
Condensate collection	603.0	0	603	1701.2	1098.2
method 29 Impinger 1	335.6	0	399.7	571.9	172.2
method 29 Impinger 2	341.7	107.8	513.5	568.8	55.3
method 29 Impinger 3	335.6	107.9	507.2	519.2	12
method 29 Impinger 4	333.4	0	397.9	401.1	3.2
method 29 Impinger 5	337.7	113.9	515.7	513.1	-2.6
method 29 Impinger 6	329.5	112.7	505.6	501.4	-4.2
Drierite Desiccant #1			944.7	964.6	19.9
Drierite Desiccant #2			1112.5	1114.8	2.3
		Material evaporated	(g)		1355.30
		% collected at 30°C			81.03
		Material collected b	efore chemical in	npingers (g)	1270.40
		% collected before a	chemical impinge	rs	93.74
		Impinger #2 - #6 ac	cumulation (g)		84.90
		Water mass balance	(g)		1.00

 Table 8.5. Mercury Partitioning Study, Mass Balance (pH=13)

8.4.2 Mercury Partitioning

All aqueous solutions making up the vacuum-evaporator/off-gas-sampler system described in Section 4.3 were analyzed to characterize the fate and behavior of the mercury in the melter's quench-scrubber aqueous waste under evaporator conditions that are representative of projected WTP pretreatment operations. As discussed above, two identical tests were conducted under different initial feed pH conditions. The mercury partitioning results obtained will now be discussed.

8.4.2.1 Melter Quench-Scrubber Evaporation At pH of 8.4

The initial evaporator test conducted on RSM quench scrubber aqueous waste did not require pH adjustment as its pH was within the 7 to 10 operating range of the reference 242-A Evaporator. The results obtained from the analyses of the condensate and off-gas sampling components summarized in Table 8.6 suggest that significant measurable mercury partitioning to both the condensate and vessel-vent off-gas streams does indeed occur. Specifically, up to 1.4% of the initial mercury inventory partitioned to the evaporator's condensate product. Moreover, the concentration of the condensate's nonvolatile constituents, e.g. Ca, Na, and Si, suggest little evidence for a gross entrainment transfer mechanism. The 3-fold increase in aqueous mercury concentration in the evaporator bottoms also suggests an increased suspension of insoluble mercury or a changing chemistry that is allowing mercury-sludge components to be solubilized. Ignoring off-gas system losses to be discussed next, the observed mercury in the evaporator's condensate fraction suggests that mercury partitioning to the evaporator's exhaust is over an order of magnitude greater than the current WTP design value for the contract maximum mercury-feed-rate condition.

				Met	hod 29 (Gas Pha	se Hg S	ampling '	Train			
	Boi	ler ^(a)	Evap.			Imp	oinger			R	eagent Blanks	
												Nitric
Element	Initial	Final	Cond.	#1	#2	#3	#4	#5	#6	H ₂ O ₂ /HNO ₃	HNO ₃ /KMnO ₄	Wash
Al	2.700	3.100							0.900			
В	370.000	600.000	1.600	0.280	0.170	0.150	1.800	1.100	1.800	0.130	1.300	0.100
Ca	1.500	2.700	0.370	0.770	0.430	0.320	1.200	0.320	0.410	0.500	0.420	0.800
Cr	2.000	3.200										
Cu				0.200	0.200		0.450					
Fe	5.700	3.000		0.220	0.100	0.070	0.150			0.300		
Hg	13.000	42.000	14.000	0.730	0.250	0.005	0.059					
Li	84.000	140.000					0.380					
Mn	0.200	0.090			0.100		0.460	570.000	370.000		170.000	
Na	690.000	1200.000	2.500	3.100	1.200	1.100	7.700			0.730	27.000	4.400
Ni	0.200	0.200										
Si	14.000	14.000	1.600	0.880	0.500	1.000	1.500			0.200		0.800
Sr	0.040	0.045										
Zn	2.700	1.200		0.090	0.030	0.020	0.110	0.050	0.050		0.040	0.035
Zr	1.100											
Hg	677	668	9.64	0.0174	0.0506	0.0011	0.0007					
$(mg)^{(b)}$												
% of	100	98.6	1.42	0.0026	0.0075	0.0002	0.0001					
Input												
(a) Solub	le concer	ntrations										
(b) Boiler	value in	cludes slu	dge con	tribution	s.							

 Table 8.6. Evaporator Component Composition for pH 8.4 Test (mg/L)

The presence of mercury in the gas-scrubber impingers, although small in comparison to the condensate fraction, further suggests a persistent, noncondensable mercury vapor source (HgCl₂) that is readily removed without the use of strong oxidizing agents, i.e., KMnO₄. The concentration of the evaporator's noncondensable mercury vapor source in the simulated inleakage flow that is projected to partition to the Pretreatment facility's vessel vent system is 8 mg/m³. However, this off-gas concentration represents an upper-bound value, as unintended inleakage of undetermined magnitude was likely significant (see pressure comments in 8.4.1.2).

Beyond the LERF/ETF and vessel-vent partitioning projections discussed above, the composition of the condensate concentrate in the evaporator bottoms will establish the manner in which mercury is distributed to the HLW and LAW streams. The current results indicate that ~93% of the mercury in the condensate concentrate is associated with UDS and will, as a result, be recycled to the HLW melter, while the remaining soluble fraction (~7%) will flow into the LAW stream.

Since the target pH range of the evaporator feed used in the current test (7 to 10) apparently was not representative of that planned for use during WTP processing, a duplicate evaporator test was subsequently conducted at pH 13. The results of this subsequent test are discussed below.

8.4.2.2 Melter Quench-Scrubber Evaporation at pH of 13

As discussed in Section 8.4.1.2, the second evaporation campaign was, with the exception of pH, conducted identically to the first test. To adjust the melter's quench scrubber's slurry to pH=13, sodium hydroxide was added. Significant cation precipitation accompanied this pH adjustment operation, which directly affected the physical properties of the slurry by significantly increasing settling rates. The

Appendix A photographs of post-test evaporator vessels graphically illustrate the differences in UDS suspension concentrations present in these evaporator bottoms. As a result of the precipitation reactions discussed above, a 23% excess of NaOH had to be used to achieve the required slurry pH value of 13.

Although constructed and operated identically to that of the first test, the laboratory-scale evaporator used for the second test achieved a much higher no-load system vacuum than was achieved during the first test. As a result, significantly improved evaporation throughput rates were achieved that appreciably reduced the required testing period duration. However, the higher throughput rates also affected the average condensate temperature, even though condenser cooling fluid was maintained at the reference 30°C temperature. This affected condensate collection efficiency as discussed in Section 8.4.1.2, and also influenced mercury off-gas system losses as is discussed below.

The mercury partitioning results obtained from the analyses of laboratory-scale evaporator samples summarized in Table 8.7 suggest that, like the previous evaporator test, significant mercury partitioning to both the condensate and vessel-vent off-gas streams does occur. However, unlike the first test, mercury partitioning to the condensate is significantly less (\sim 17×) than to the off-gas stream, and the overall evaporator mercury partitioning is appreciably greater (\sim 3×).

	Bo	iler	Evap.	Method	29 Gas P	hase Hg	Sampling	g Train Imj	pinger ^(a)
Element	Initial	Final ^(b)	Cond. ^(c)	#1	#2	#3	#4	#5	#6
В	646.000	640.000	(0.110)						
Ca	69.000	56.200	(0.270)						
Li	142.000	146.000	(0.010)						
Na	6920.000	6750.000	(2.200)						
Hg	380.000	326.000	0.920	0.095	0.814	3.550	0.017	11.600	0.003
% of Input	100.000	85.700	0.242	0.025	0.214	0.933	0.005	3.060	0.001
(a) Not mea	asured:								
(b) Slurry r	esults only;	does not inc	clude evapo	orator vess	sel deposi	ts.			
(c) Values	in parenthes	es are below	v analytical	l reportabl	le limits.				

Table 8.7. Evaporator Component Composition for pH 13 Test (mg)

Specifically during the current test, 0.24% of the mercury present in the evaporator feed partitioned to the condensate, and 4.2% partitioned to the off-gas stream. Of the uncondensable (30°C condenser) mercury partitioning to the off-gas stream, 0.7% was condensed at 0°C, and ~27% was collected by the nonoxidizing traps while the remaining ~72% was collected by the oxidizing KMnO₄/H₂SO₄ gas scrubber solutions. This suggests that the mixed mercury vapor source was dominated by the elemental form (~75%), but it contained a significant proportion (~25%) of one or more volatile oxidized mercury compounds.

If pure physical capture processes, e.g., condensation, were responsible for the mercury sample train collection, a distribution similar to that of moisture collection illustrated in Table 8.5 would be also expected for mercury. Since it is not, selective chemical processes are no doubt responsible for the distribution of collected mercury as explained above. Although the distribution of mercury across the sequential H_2O_2/HNO_3 gas-scrubbers appears unintuitive, chemical dilution affects resulting from

significantly different moisture accumulations within these traps may be partly responsible for the differing mercury collection yields in these sequential traps (see Table 8.5).

As in the previous, pH=8.4 evaporator test, the estimates of nonvolatile cationic species in the evaporator condensate suggest little evidence for aqueous aerosol carryover during the evaporator feed concentration process. The concentration increase in soluble mercury in the evaporator's feed concentrate (3.6) was also found to be greater than the overall test concentration factor (2.7), and most of the mercury mass in the evaporator's concentrate (86%) was associated with UDS in total agreement with previous testing results.

It should be noted that only 90% of the mercury is accounted for in the mass balance data summarized in Table 8.7. Since mercury surface deposits on all components down-stream of the evaporator's reflux condenser were collected and accounted for, the unaccounted for mercury is presumed to be associated with unremoveable evaporator vessel residues.

8.4.2.3 Evaporator Test Discussion

In contrast to the results of the second test in which only 0.24% of evaporated mercury was found in the primary condensate vessel, the analytical data from the initial evaporator test indicated that 99% of the partitioned mercury was condensable and resided in the evaporator's condensate. Moreover, the distribution of mercury trapped by chemical gas scrubbers during this first test suggested that the remaining mercury carried into the off-gas stream was primarily associated with one or more volatile oxidized forms of mercury whereas a mixed, 1:3, oxide/elemental volatile source of mercury was observed during the second test. For the same reason that the higher partial pressure of atmospheric gases severely limited moisture evaporation rates during the first test, overall mercury loss was also found to be significantly lower (3×) than that observed during the second test. Moreover, the lower condensate and off-gas temperatures observed during the first evaporation test that resulted from the reduced heat load to the evaporator system's condenser are no doubt responsible for the higher mercury condensate accumulation and lower mercury off-gas concentration observed during that test.

Of the two tests performed, overall operational conditions of the second test (pH=13) appear to be most representative of projected WTP evaporator operations due to the unreasonably low evaporation throughput rates that characterized the first test. Consequently, apart from pH considerations, the mercury partitioning behavior recorded during the second test is also more likely to be characteristic of expected Plant operations than is the behavior observed during the first test.

Although test results differed, the mercury partitioning observed during the current laboratory evaporator tests are in stark contrast to previous work^(a) that found no detectable mercury contamination in

242-A Evaporator Documented Safety Analysis, Chapter 2 – Facility Description. Waste Management Project, Hanford Site, Rev. 0, Richland, WA (April 10, 2003).

Waste Feed Evaporation: Physical Properties and Solubility Determination (U), Savannah River Technology Center, SRT-RPP-2003-00094, Rev 0. Westinghouse Savannah River Company, Savannah River Site, Aiken,

⁽a) Engineering Specifications for the Forced Circulation Vacuum Evaporator System: DIM No. 24590-PTF-3PI-MEVV-00001, Rev A. Issued by the River Protection Project – Waste Treatment Plant Project, Richland, WA (June 4, 2002).

evaporator condensate or in the vessel vent-exhaust stream. These contradictory results may be due to chemical differences in the evaporator feed materials used, but the factors responsible for these differences need to be identified as they apparently have a major impact upon the partitioning behavior of mercury.

8.5 Process Summary

The mercury CEM results obtained during C-104/AY-101 melter-flowsheet evaluations were to have definitively elucidated any correlations existing between the parameters of the test (Hg, Cl, and reductant concentrations) and the melter-effluent characteristics of mercury. However, the performance of this continuous monitor was less than ideal during the RSM testing campaign. Specifically, the mercury CEM did not provide reliable off-gas vapor concentration values, due, presumably, to inadequate sample stream dilution and resultant condensation losses. However, since the major volatile species expected (Hg° and HgCl₂) exhibit very similar temperature-dependent vapor pressures, sample condensation losses would be expected to affect both species in a similar way; consequently, CEM speciation results should remain valid if: 1) condensation losses were responsible for the observed nonquantitative instrumental behavior and 2) HgCl₂ is the primary oxidized form of mercury being sampled. However, meaningless elemental to total mercury ratios were nevertheless observed, especially for the melter-exhaust sampling site during the first and last testing segments.

Comparing the valid CEM data over the first three test conditions with soluble mercury collected in the quench-scrubber condensate, both independent sets of data suggest a melter mercury-effluent source dominated by oxidized species (presumably HgCl₂). Conducting a similar comparison for the last four test segments, these data suggest a melter mercury source that is mostly in the elemental form. Overall, the mercury CEM did not demonstrate a consistent dependence on the test-parameter variables.

Discrete sampling data were also used to characterize the melter's mercury source term by examining the distribution of mercury across sampler components. The results obtained during the 2nd test condition suggested a mercury effluent source primarily in the oxidized state, which is in total agreement with both CEM and EVS data previously discussed. Unlike the 2nd test condition, the Method-29 sampler results suggested a predominantly elemental melter-mercury effluent source during the 4th test condition in total conformity with both CEM and EVS data. On the other hand, sampler results obtained during the 5th test condition suggested a mixed melter mercury source with a strong elemental component, which is similar to corresponding CEM data that indicated a 90% elemental source. Like the CEM data, the massive mercury concentrations present during the last test condition created unsuitable sampling conditions for speciation determinations. However, because of the high Hg:Cl molar ratio (2.6) created by the Mx condition for mercury, there is little doubt that the melter's volatile mercury source term was dominated by the elemental species, which is corroborated by the decrease in soluble mercury content of the EVS condensate during this test segment.

SC (May 13, 2003).

AN-107 (C) Simulant Bench-Scale Law Evaporation with Organic Regulatory Analysis, SRT-RPP-2000-00047, Savannah River Technology Center. Westinghouse Savannah River Company, Savannah River Site, Aiken, SC.

Just as the melter's mercury source term for the last test condition can be deduced using the existing Hg:Cl molar ratio, so can the contrasting results for the two halves of the melter test previously described be explained. For the initial three test segments for which the oxidized form of effluent mercury predominated, the Hg:Cl molar ratio was less than 1. On the other hand, during the second half of testing when the melter's mercury source was predominantly elemental, the Hg:Cl molar ratio was, with a single exception, significantly greater than 1. The 5th test condition, being the 2nd half exception, used feed with a Hg:Cl molar ratio of 0.4 and exhibited a mixed melter-mercury source according to the sampling data discussed above.

Although not dramatically apparent from any single source of data, the CEM, off-gas sampling, and waste-stream data, when taken together, suggest that when significant chlorine is present in a mercurycontaining feed stream that is vitrified under WTP processing conditions, the formation of HgCl₂ is both thermodynamically and kinetically favored. With a Hg:Cl molar ratio of 0.1 (test conditions #2 & #3), conversion to chemically combined mercury was essentially complete, whereas a 0.4 molar ratio (test condition #5) produced a mixed mercury effluent source with an appreciable elemental component. On the other hand, for test segments 4, 6, and 7, where the elemental form was the dominant melter mercury effluent species, Hg:Cl ratios were all in excess of 2.5. Thus, these results suggest that when mercury-containing melter feed with a very low Hg:Cl molar ratio is vitrified under WTP processing conditions, a chemically combined (HgCl₂) mercury effluent source results; otherwise, depending upon the magnitude of the Hg:Cl molar ratio, a mixed or elemental source will dominate the melter-mercury source term.

Because of the focused interest in characterizing the melter source term, most mercury CEM data associated with the exhausts of the EVS and HEME were conducted during the second half of testing, i.e., test conditions 4 through 7. Recognizing that these data were similarly influenced by the instrument-performance problems discussed above, the body of reasonable data collected indicate an off-gas mercury vapor source that is composed primarily of the elemental species. This is in agreement with the corresponding melter-source-term results previously discussed.

Post-test evaluation of the melter's secondary waste streams has shown that \sim 70% of the mercury fed to the melter penetrated the off-gas system's quench scrubber and accumulated in the HEME's deep-bed filter, suggesting a primarily elemental mercury-effluent source. Indeed, 76% of the total mercury processed during RSM testing occurred during test segments 6 and 7 (Mx conditions) when the melter's mercury-effluent source was predominantly in the elemental state. Of the 27% of the mercury collected in the quench-scrubber's condensate, only 9% of the quench-scrubber's mercury inventory was found to be soluble.

To project HLW/LAW, LERF/ETF, and VV Hg partitioning during pretreatment secondary-waste concentration, representative samples of the quench-scrubber's condensate/scrubbing liquor were vacuum evaporated and concentrated with and without initial pH adjustment. Of the two tests conducted, the results obtained from the pH-adjusted (pH = 13) evaluation are considered to be most representative of projected WTP evaporation conditions and as a result are discussed below. The analyses of evaporator condensate and off-gas sample solutions suggest that the mercury partitioning to the Pretreatment Plant's condensate collection and VV systems will be an order of magnitude greater than the current WTP design value for the contract maximum mercury feed rate (Cramer 2001). Specifically, the observed evaporator mercury DF (~22) for RSM condensate is significantly less than the reference feed concentrator DF of 997 for this element. Of the partitioned mercury, 5.4% was collected in the overhead condensate fraction

with the remaining 94.6% being carried off by the gaseous exhaust (vessel vent system). The chemical nature of the evaporator's mercury off-gas source term was evaluated by observing the mercury distribution across the off-gas chemical scrubbers employed during the laboratory-scale tests. The results obtained suggest a mixed volatile oxide/elemental mercury source term dominated by the elemental form ($3\times$). Post-test analyses of the mercury remaining in the evaporator concentrate after both pH=8.4 and pH=13 tests demonstrated that most (86% and 93%, respectively) of the mercury present was associated with UDS. If representative, this result suggests that 7% to 14% of the mercury present in WTP evaporator bottoms will be recycled to the HLW melter, with the remainder contributing to the LAW stream.

However, since the Pretreatment Plant's evaporator influent stream will be composed of more than just the secondary, aqueous waste generated by HLW melters, the mercury-partitioning results established by the above laboratory-scale evaporator tests may not be totally representative of actual plant operations. All factors that can affect the chemical composition of mercury in the evaporator's influent stream need to be considered before accurate Pretreatment Plant projections of mercury behavior can be established.

Melter partitioning of individual feed constituents, derived from both off-gas sampling and secondary waste-stream analysis, revealed that with the exception of boron, mercury, sulfur, and the halogens, essentially all feed constituents (excluding, for example, C, N, and H_2O) were found to be primarily in a condensed state downstream of the film cooler. Overall, the element-specific DFs recorded during RSM testing are reasonably close to general expectations and are generally consistent with previous RSM testing results. Indeed, the very reasonable mass closure demonstrated for most of the feed constituents for which complete analytical data exist suggests that the current melter test has successfully characterized the C-104/AY-101 vitrification flowsheet.

Representative glass samples generated under worst-case testing conditions (Mx/Hi/Lo and Mx/Hi/Hi) were subjected to EPA's TCLP test (TCLP 1992). The concentrations of all hazardous analytes, except for Ba and Cr (for which estimates are provided), were found to be below instrument detection limits, and all were below their respective UTS limits. The extremely low mercury concentrations found in the glass present no adverse leaching problems that would preclude the glass-vitrification product from conforming with all existing RCRA land-disposal limits (40 CFR 268).

9.0 QA Requirements

PNWD implements the RPP-WTP quality requirements by performing work in accordance with the PNWD Waste Treatment Plant Support Project quality assurance project plan (QAPjP) approved by the RPP-WTP Quality Assurance (QA) organization. This work was performed to the quality requirements of NQA-1-1989 Part I, Basic and Supplementary Requirements, and NQA-2a-1990, Part 2.7. These quality requirements are implemented through PNWD's *Waste Treatment Plant Support Project (WTPSP) Quality Assurance Requirements and Description Manual*. The analytical requirements are implemented through PNWD's *Work in Support of Regulatory Programs*.

PNWD addresses internal verification and validation activities by conducting an Independent Technical Review of the final data report in accordance with PNWD's procedure QA-RPP-WTP-604. This review verifies that the reported results are traceable, that inferences and conclusions are soundly based, and that the reported work satisfies the Test Plan objectives. This review procedure is part of PNWD's *WTPSP Quality Assurance Requirements and Description Manual*.

10.0 References

40 CFR 60, Appendix A. 2002. U.S. Environmental Protection Agency. "Standards of Performance for New Stationary Sources," *US Code of Federal Regulations*. Available at: http://www.epa.gov/ttn/emc/promgate.html.

40 CFR 268. 2003. U.S. Environmental Protection Agency. "Land Disposal Restrictions," US Code of Federal Regulations.

Cramer K. 2001. *Mercury Pathway and Treatment Assessment for the WTP*. 24590-WTP-RPT-PR-01-011, Rev. 0, Richland, WA.

Goles RW, and AJ Schmidt. 1992. Evaluation of Liquid-Fed Ceramic Melter Off-Gas System Technologies for the Hanford Waste Vitrification Plant. PNL-8109, Pacific Northwest Laboratory, Richland, WA.

Goles RW, JA Del Debbio, RJ Kirkham, BD MacIsaac, JA McCray, DD Siemer, and NR Soelberg. 2002. *Test Summary Report INEEL Sodium-Bearing Waste Vitrification Demonstration RSM-01-2*, PNNL-13869, Pacific Northwest National Laboratory, Richland, WA.

Office of River Protection (ORP). 2003. *River Protection Project System Plan: Retrieve and Treat Hanford's Tank Waste and Close the Tank Farms to Protect the Columbia River*. ORP-11242, Rev. 1, Draft E, Richland, WA.

Perez JM, DF Bickford, DE Day, DS Kim, SL Lambert, SL Marra, DK Peeler, DM Strachan, MB Triplett, JD Vienna, and RS Wittman. 2001. *High-Level Waste Melter Study Report*. PNNL-13582, Pacific Northwest National Laboratory, Richland, WA.

Toxicity Characteristic Leaching Procedure (TCLP). 1992. Test Methods for Evaluating Solid Waste, Volume 1C: Laboratory Manual Physical/Chemical Methods, SW-846, Method 1311, Rev. 2, U.S. Environmental Protection Agency, Office of Solid Waste and Emergency Response, Washington D.C.

Appendix A

Evaporator Hardware Photographs

Appendix A: Evaporator Hardware Photographs



Laboratory Scale Evaporator Hardware

Boiler, Distillation Columns, Condensate Pot, and Gas Scrubbers



Insulated Evaporator System



Column Temperature Control Recirculators, Foreground; Boiler Temperature Controller, Inleakage Flow Controller, Background



Mercury Vapor Chemical Scrubbers



 $\label{eq:concentrate} Evaporator \mbox{ Concentrate at the Conclusion} \\ of the \mbox{ } pH = 8.4 \mbox{ Test}$



Evaporator Concentrate at the Conclusion of the pH = 13 Test

Appendix B

Feed Batching and Data Sheets

Appendix B: Feed Batching and Data Sheets

RSM-HgBtch-	1			
Prepared By:	Harry Smith	۱		
Date:	7/28/2003		Time:	15:00
Melter Feed Tank Tare:	Yes	No X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	He Lo X X X	eel Hi	<u>Tan</u> Lo X X X	<u>get</u> <u>Hi</u>
Trial Tank Mass:	<u>He</u> <u>Mass (kg)</u> <u>47.22</u>	<u>eel</u> Vol (L) 0.00	<u>Tar</u> <u>Mass (kg)</u> <u>111.00</u>	<u>get</u> Vol (L) 44.13
Heel Adjustment:	<u>Maxi</u> Mass (kg) #N/A	<u>mum</u> Vol (L) <u>#N/A</u>	<u>Actual</u> <u>Mass (kg)</u> <u>47.22</u>	(<u>⊴Max)</u> Vol (L) 0.00
Final Tank Mass:	<u>He</u> <u>Mass (kg)</u> <u>47.22</u>	<u>eel</u> Vol (L) 0.00	<u>Post</u> <u>Mass (kg)</u> <u>111.00</u>	<u>Xfer</u> <u>Vol (L)</u> <u>44.13</u>
Additive Mass: Hg(NO₃)₂·H₂O:	<u>Target</u> 19.42	g	<u>Actual</u> {	g
NaCl:	<u>3,64</u>	g	3.51	g
Sugar:	0.00	g	0.00	g
	I	Balance ID	743886	
Completed By:	Na		t.	Date /
Reviewed By:	aule	a.		7/28/03
Feed Properties:	<u>kg/L</u> 1.45	<u>kg-Ox/L</u> 0.56		
*Feed Concentration:	Additive: Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

C-104/AY-101 FEED BATCHING CALCULATION SHEET: TP-RPP-WTP-252

RSM-HgBtch-	2			
Prepared By:	Harry Smith			
Date:	7/29/2003		Time:	10:29
Melter Feed Tank Tare:	<u>Yes</u>	No X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	Lo X X X X	<u>əi</u> <u>Hi</u>	<u>Tar</u> Lo X X	get Hi X
Trial Tank Mass:	Hee Mass (kg) 81.00	<u>əl</u> Vol (L) 23.37	<u>Tar</u> Mass (kg) <u>81.00</u>	r <u>get</u> <u>Vol (L)</u> 23.37
Heel Adjustment:	<u>Maxin</u> <u>Mass (kg)</u> #N/A	<u>um</u> Vol (L) #N/A	<u>Actual</u> Mass (kg) <u>81.00</u>	<u>(⊴Max)</u> Vol (L) 23.37
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> 81.00	<u>>l</u> Vol (L) 23.37	<u>Post</u> <u>Mass (kg)</u> <u>81.00</u>	<u>Xfer</u> <u>Vol (L)</u> 23.37
Additive Mass: Hg(NO ₃)₂·H₂O:	Target 0.00)	Actual	g
NaCI:	<u>10.93</u> g	3	10,93	g
Sugar:	<u>0.00</u> s)	0.00	g
	Bi	alance ID:	743886	
Completed By:	Nan Harrzi	le Said	te	Date, 7/29/09
Reviewed By:	au ha	4		7/29/03
Feed Properties:	<u>kg/L</u> 1.45	<u>kq-Ox/L</u> 0.56		
*Feed Concentration:	Additive: Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

Note: Shaded cells are not for electronic batch sheet input

RSM-HgBtch-	3			
Prepared By:	Ron Goles			
Date:	7/29/2003		Time:	23:45
Melter Feed Tank Tare:	<u>Yes</u>	<u>No</u> Х	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	<u>Hee</u> <u>Lo</u> 조 조	<u>н</u> і Х	<u>Tar</u> Lo X X	get Hi X
Trial Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> <u>62.60</u>	<u>əl</u> Vol (L) 10.64	<u>Tar</u> Mass (kg) <u>83.66</u>	rget Vol (L) 25.21
Heel Adjustment:	<u>Maxin</u> Mass (kg) #N/A	<u>num</u> Vol (L) #N/A	<u>Actual</u> <u>Mass (kg)</u> <u>62.60</u>	<u>(⊴Max)</u> Vol (L) 10.64
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> 62.60	<u>el</u> Vol (L) 10.64	<u>Post</u> <u>Mass (kg)</u> <u>83.66</u>	<u>Xfer</u> Vol (L) 25.21
Additive Mass: Hg(NO ₃) ₂ ·H ₂ O:	<u>Target</u> 6.41	9	Actual 6.49	9
NaCl:	<u>8.02</u> 9	g	8.08	g
Sugar:	<u>0.00</u> • •	g	0.00	g HDS T(1/0
	В	alance ID	743886	
Completed By	Nar	ne	l ,	Date 7/29/03
Reviewed By	Harry	12 Die	ti	8/7/03
Feed Properties:	<u>kg/L</u> 1.45	<u>kg-Ox/L</u> 0.56		
*Feed Concentration:	<u>Additive:</u> Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

RSM-HgBtch-	4			
Prepared By:	John Tixier			
Date:	7/30/2003		Time:	5:47 AM
Melter Feed Tank Tare:	Yes	<u>No</u> X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	Hee Lo X X	<u>əl</u> <u>Hi</u> X	<u>Tarr</u> <u>Lo</u> <u>X</u>	g <u>et</u> Hi X X
Trial Tank Mass:	Hee Mass (kg) 75.00	<u>el</u> Vol (L) <u>19.22</u>	<u>Tar</u> Mass (kg) 75.00	<u>get</u> Vol (L) <u>19.22</u>
Heel Adjustment:	<u>Maxin</u> Mass (kg) #N/A	<u>num</u> Vol (L) #N/A	<u>Actual (</u> <u>Mass (kg)</u> <u>75.00</u>	<u>′⊴Max)</u> Vol (L) 19.22
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> 75.00	<u>el</u> Vol (L) 19,22	<u>Post</u> <u>Mass (kg)</u> <u>75.00</u>	<u>Xfer</u> <u>Vol (L)</u> <u>19.22</u>
Additive Mass: Hg(NO ₃) ₂ ·H ₂ O:	<u>Target</u> 0.00	9	Actual	g
NaCi:	<u>0.00</u> 9	g	·	g
Sugar:	<u>96.11</u> (g	95.90	g
	B	alance ID:	743886	
Completed By:	f Nar	ne		Date 7/30/07
Reviewed By:	Qu. 1	Jal-		7/20/03
Feed Properties:	<u>kg/L</u> 1.45	<u>kg-Ox/L</u> 0.56	844-80002111	
*Feed Concentration:	Additive: Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

RSM-HgBtch-	5			
Prepared By:	Harry Smith			
Date:	7/30/2003		Time:	15:40
Melter Feed Tank Tare:	<u>Yes</u>	No X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	<u>Нее</u> <u>Lo</u> Х	Hi X X	<u>Tar</u> Lo X X	get Hi ≚
Trial Tank Mass:	Hee Mass (kg) 49.15	<u>Vol (L)</u> <u>1.34</u>	<u>Tar</u> <u>Mass (kg)</u> <u>115.28</u>	r <u>get</u> Voi (L) 47.09
Heel Adjustment:	<u>Maxin</u> <u>Mass (kg)</u> 47.22	<u>um</u> Vol (L) 0.00	<u>Actual</u> <u>Mass (kg)</u> <u>49.15</u>	(⊴Max) Vol (L) 1.34
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> <u>49.15</u>	<u>əl</u> Vol (L) <u>1.34</u>	Post Mass (kg) <u>115.28</u>	<u>Xfer</u> Vol (L) <u>47.09</u>
Additive Mass: Hg(NO ₃) ₂ ·H ₂ O:	<u>Target</u> 61.57	9	Actual	9
NaCI:	<u>3.15</u> 9	9	3.15	9
Sugar:	<u>-6.68</u>	9		g
	В	alance ID	<u>74388(</u>	
Completed By	Nar <u>Hoster</u>		t,	Date -7/14/03
Reviewed By	BULL	ele.		7/30/03
Feed Properties:	<u>kg/L</u> 1,45	<u>kg-Ox/L</u> 0.56	1 Yer	
*Feed Concentration:	<u>Additive:</u> Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

RSM-HgBtch-	6			
Prepared By:	Harry Smith			
Date:	7/31/2003		Time:	8:35
Melter Feed Tank Tare:	<u>Yes</u>	No X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	Hee Lo X X	<u>Hi</u> ⊻	<u>Tar</u> Lo X	r <u>get</u> Hi X X
Trial Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> <u>88.00</u>	<u>Vol (L)</u> 28.22	<u>Tai</u> <u>Mass (kg)</u> <u>88.00</u>	r <u>get</u> Vol (L) 28.22
Heel Adjustment:	<u>Maxim</u> Mass (kg) #N/A	<u>um</u> Vol (L) #N/A	<u>Actual</u> <u>Mass (kg)</u> <u>88.00</u>	<u>(⊴Max)</u> Vol (L) 28.22
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> 88.00	<u>Vol (L)</u> 28,22	<u>Posi</u> <u>Mass (kg)</u> <u>88.00</u>	<u>Xfer</u> <u>Vol (L)</u> 28.22
Additive Mass: Hg(NO ₃) ₂ ·H ₂ O:	<u>Target</u> 0.00 g)	Actual	g
NaCl:	<u>13,20</u> g)	13.2	g
Sugar:	<u>0.00</u> - c	J	6.00	g
	Ba	alance ID:	743886	
Completed By:	Nar Herrez de	ne Smil	*	Date ,
Reviewed By:	Rith	le.		7/31/03
Feed Properties:	<u>ka/L</u> 1,45	<u>kg-Ox/L</u> 0.56	Her	
*Feed Concentration:	Additive: Mercury: Chlorine: Sucrose:	<u>HI</u> 0.15 0.06 5	<u>Lo</u> 0.05 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd



Note: Shaded cells are not for electronic batch sheet input!

RSM-HgBtch-	8			,
Prepared By:	RW Goles			
Date:	8/1/2003		Time:	4:45
Melter Feed Tank Tare:	<u>Yes</u>	<u>No</u> X	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	He Lo X X	el Hi X	<u>Таг</u> <u>Lo</u> Х	get Hi X X
Trial Tank Mass:	<u>He</u> <u>Mass (kg)</u> <u>68.50</u>	<u>el</u> <u>Vol (L)</u> <u>14.72</u>	<u>Tar</u> <u>Mass (kg)</u> <u>100.00</u>	get Vol (L) 36.52
Heel Adjustment:	<u>Maxir</u> Mass (kg) #N/A	<u>mum</u> Vol (L) <u>#N/A</u>	<u>Actual</u> <u>Mass (kg)</u> <u>68.50</u>	(<u>⊴Max)</u> Vol (L) <u>14.72</u>
Final Tank Mass:	<u>He</u> <u>Mass (kg)</u> <u>68.50</u>	<u>el</u> Vol (L) 14.72	<u>Post</u> <u>Mass (kg)</u> <u>100.00</u>	<u>Xfer</u> Vol (L) 36:52
Additive Mass: Hg(NO ₃) ₂ ·H ₂ O:	<u>Target</u> 285,84	g	Actual 235,77	g
NaCl:	<u>11.99</u>	g	12.02	g
Sugar:	<u>0.00</u>	g		g
	E	alance ID:	743886	i
Completed By:	Nai RV S	ne	CT I I I I I I I I I I I I I I I I I I I	Date 7///03
Reviewed By:	forge	J.K		8/1/03
Feed Properties:	<u>kg/L</u> 1.45	<u>kg-Ox/L</u> 0.56		
*Feed Concentration:	<u>Additive:</u> Mercury: Chlorine: Sucrose:	<u>HI</u> 0.95 0.06 5	<u>Lo</u> 0.15 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

C-104/AY-101 FEED BATCHING CALCULATION SHEET: TP-RPP-WTP-252 (Test Condition 6 & 7 only! Maximum Mercury Concentration)

RSM-HgBtch-	9			
Prepared By:	Harry Smith			
Date:	8/1/2003		Time:	20:53
Melter Feed Tank Tare:	Yes	<u>No</u> Х	<u>TkWt (kg)</u> 47.22	
Feed Additive Condition (X)*: Mercury: Chlorine: Sucrose:	Lo Lo	l Hi X X	<u>Tar</u> Lo	get Hi X X X X
Trial Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> <u>74.00</u>	<u>l</u> Vol (L) 18,53	<u>Tar</u> <u>Mass (kg)</u> <u>74.00</u>	<u>get</u> Vol (L) 18.53
Heel Adjustment:	<u>Maxim</u> Mass (kg) #N/A	<u>um</u> Vol (L) #N/A	<u>Actual</u> <u>Mass (kg)</u> <u>74.00</u>	(⊴Max) Vol (L) 18.53
Final Tank Mass:	<u>Hee</u> <u>Mass (kg)</u> 74.00	<u>I</u> Vol (L) 18.53	<u>Post</u> <u>Mass (kg)</u> <u>74.00</u>	<u>Xfer</u> Vol (L) 18,53
Additive Mass: Hg(NO ₃)₂ [.] H₂O:	<u>Target</u> 0.00 g		Actual	g
NaCl:	<u>0.00</u> g		0.00	g
Sugar:	<u>92.65</u> g	I	92,4	g
	Ba	alance ID:	743886	
Completed By:	Nar Harza	ie Sz	it.	Date 8/1/03
Reviewed By:	RV A	la .		8/1/03
Feed Properties:	<u>ka/L</u> 1.45	<u>kg-Ox/L</u> 0.56		
*Feed Concentration:	Additive: Mercury: Chlorine: Sucrose:	<u>H1</u> 0.95 0.06 5	<u>Lo</u> 0.15 0.009 0	<u>Unit</u> FdOx Wt% FdOx Wt% g/L-Fd

C-104/AY-101 FEED BATCHING CALCULATION SHEET: TP-RPP-WTP-252 (Test Condition 6 & 7 only! Maximum Mercury Concentration)

Data Sheet #1: Priority and Electrical Data

WTPSP TSS# B-91

TI-RPP-WTP-264

	R R R R R R R R R R R R R R R R R R R
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	10000000000000000000000000000000000000
1120 1150 1150 1150 1150 1150 1150 1150	1120 1020 1020 1120 1120 1120 1120 1120
EL COLOC	1-1-20-20-20-20-20-20-20-20-20-20-20-20-20-
10000000000000000000000000000000000000	227 - 22 - 23 - 24 - 22 - 27 - 27 - 27 - 27 - 27 - 27
2000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	25 25 25 25 25 25 25 25 25 25
200 200 200 200 200 200 200 200 200 200	1000
1000 000 000 000 000 000 000 000 000 00	699 7 1 4 8 8 8 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	6 9 9 4 4 8 9 9 4 4 9 9 4 4 9 9 4 4 4 9 9 4 4 4 9 9 4 4 4 9 9 4 4 4 9 4 4 4 9 4 4 4 9 4
NA NA NA NA NA NA NA NA NA NA	6 3 9 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4
244 24 24 24 24 24 24 24 24 26 25 26 25 26 25 26 25 26 25 26 25 26 25 26 25 26 25 26 25 26 26 26 26 26 26 26 26 26 26 26 26 26	244444 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
рана в стана	244 43 323 32 34 1 1 1 1 2 2 3 3 2 3 3 3 3 3 3 3 3 3 3 3
N N N N Y N Y N Y 34.7 32.3 32.4 32.3 32.4 34.7 32.3 32.2 32.7 9.7 34.7 32.3 32.7 9.7 9.7 34.7 32.3 32.7 9.7 9.7 34.6 1.6 1.6 0.30 9.7 11.50 11.60 1.65 0.4 9.7 11.50 1.60 1.6 0.30 9.7 11.60 1.60 1.60 0.30 0.4 11.65 1.65 2.6 0.4 0.4 11.65 1.65 2.6 0.5 0.5 1.8 1.6 1.6 0.5 0.1 0.1 1.9 1.6 1.6 1.6 0.5 0.1	х х х х х х х х х х х х х х х х х х х
Y N Y N 34:7 32.3 32.4 25.1 34:7 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 24.4 42 24.4 24.4 44 42 44 44 44 9550 950 950 950 950 954 950 950 950 950 950 950 955 746 956 950 950 955 746 956 950 955 746 956 105 950 955 255 255 255 255 955 50 355 355 255 255 965 1.9 1.9 1.9 1.13 113 1.8 1.9 31 35 255	Y N Y N 34:1 32.3 32.4 25.1 34:1 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 32.3 32.4 24.4 34:1 32.1 32.4 24.4 34:1 4.3 4.4 4.4 41.4 4.3 4.4 4.4 41.4 4.3 4.4 4.4 41.4 11.6 11.5 11.6 84.8 84.6 84.6 85.6 84.8 84.6 84.6 84.6 4 A A A 4 A A A 4 A A A 4 A A A 4 A A A 5 5.5 2.5 2.5 6.3 1.4 A A
34:0 34:0 256 94:1 32.3 32.0 24.1 96:0 13.2 37.0 24.1 96:0 13.2 37.0 24.1 11.50 11.50 11.50 24.1 11.50 11.50 11.50 11.50 11.50 11.50 11.50 11.50 11.50 11.50 11.50 11.50 99:1 1.50 24.4 44 41 1.43 74.4 44 91:1 74.7 74.7 44 91:1 74.7 74.7 44 91:1 74.6 74.6 44 91:1 74.7 74.7 74.7 91:1 74.7 74.7 74.7 74.7 91:1 74.7 74.7 74.7 74.7 91:1 74.7 74.7 74.7 74.7 91:1 74.7 74.7 74.7 74.7 91:1 74.7 <td>94.0 32.3 32.0 25.0 94.7 32.3 32.3 34.0 24.1 87.0 87.0 87.0 24.1 24.1 87.0 87.0 27.0 24.1 24.1 11.80 11.80 11.80 24.1 24.1 21.1 21.1 21.1 24.1 24.1 21.1 21.1 21.1 24.1 24.1 21.1 21.1 21.1 21.1 24.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1</td>	94.0 32.3 32.0 25.0 94.7 32.3 32.3 34.0 24.1 87.0 87.0 87.0 24.1 24.1 87.0 87.0 27.0 24.1 24.1 11.80 11.80 11.80 24.1 24.1 21.1 21.1 21.1 24.1 24.1 21.1 21.1 21.1 24.1 24.1 21.1 21.1 21.1 21.1 24.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1 21.1
34:1 32.3 32.4 34.1 39:0 53.2 39.4 34.1 39:0 53.2 39.4 34.1 21:1 2.1 21.7 21.4 21.7 21:1 2.1 21.7 21.7 21.7 11:00 11:00 11:00 11.50 11:00 41.4 4.2 4.7 21.7 21.7 41.4 4.2 4.7 4.4 4.4 0 0.2 0.4 4.4 4.4 0 0.2 0.4 4.4 4.4 0 0.2 0.4 4.4 4.4 0 0.2 0.4 4.4 4.4 0 0.2 0.4 4.4 4.4 0 0.3 0.5 0.5 0.5 0.3 0.5 0.5 0.5 0.5 0.4 1.8 1.8 1.7 0.5 0.5 0.5 0.5 0.6 0.5 0.5 1.9 0.7 0.5 0.5 1.9 0.8 0.6 0.5 1.9	34:1 32.3 33.0 34.1 36:0 83.2 33.0 34.1 36:1 2.7 24.4 24 24:1 2.3 33.0 34.1 24:1 2.7 24.4 24 24:1 2.7 24.4 24 24:1 2.7 24.4 24 25:0 150 24.7 24 26:1 160 160 25 84:1 84:1 84:1 744 4 1 1 24 84:1 84:1 744 744 25:0 23.5 25 25 4 1 1 1 55:0 23.5 25 25 4 1 1.5 1.5 4 1.4 1.5 1.5 4 1.4 1.5 1.5 4 1.4 1.5 1.5 4 1.5 1.5 1.5 4 1.6 1.5 1.5 5 1.5 1.5 1.5 6.3 1.5 1.5 1.5 6.3 1.5 1.5 1.5 6.3 1.5 1.5 </td
39:1 32:2 24:1 32:2 24:1 24:1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.1 2.2 0 0.1 2.1 2.1 1.4 0.3 0.5 3.5 3.5 2.5 0.3 0.5 0.5 0.6 0.3 0.5 0.5 0.5 0.4 1.8 1.8 1.8 1.8 1.8 1.8 1.4	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
11.50 11.50 <th< td=""><td>11.50 <th< td=""></th<></td></th<>	11.50 11.50 <th< td=""></th<>
11.70 11.50 11.50 11.50 11.50 414 12 44 44 0 0 0.2 0.44 44 0 0 0.2 0.4 0 848 148 149 145 849 148 149 145 849 149 145 44 14 14 6.3 0.5 0.4 0.4 6.3 0.5 0.4 0.4 6.3 0.5 0.4 0.4 1.8 1.8 1.8 1.9 1.8 1.9 1.50 10.50 10.50 1.8 1.9 1.8 1.9	11.50 11.60 11.50
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	44 44 44 44 44 44 44 44 44 44 44 44 44
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	8/30 8/30 <th< td=""></th<>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Matter 741 741 744 746 Rot 785 785 785 405 Rot 846 847 746 746 A A A A 74 745 A A A A 74 745 A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A A B A A A A A B A
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Ret
8** 8** <td>44 9 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td>	44 9 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
A B A B A B A B	A A
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	50 25 25 25 0.3 0.5 0.6 0.6 0.6 49 66 0.4 150 750 750 1.4 1.4 1.8 150 150 160 1.4 1.4 1.8 170 150 160 1.9 1.4 1.8 17 1050 1050 69 0.5 71 17 17 12
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	6:3 0.5 0,6 0,6 0,6 48 785 750 750 49 56 60 59 1.7 1.9 1.8 1.9 1.8 1.9 1.8 1.9 1.9 1.3 1.1 12
281 785 750 750 49 55 60 57 1.1 1.1 1.2 1.9 1.9 1.20	150 150 150 150 150 150 150 150 150 150
49 55 61 57 1.8 1.9 1.8 1.9 1850 1851 1050 1030	49 55 60 55 1050 1050 1050 1050 1050 1050
1050 1050 1050 1050	1.8 1.1 1.1 1.8 1.7 1.050 1050 10.00 69 73 71 72
1630 10501 1500 1500 1692	(030 m51 1050 1030 69 73 71 72
	69 73 71 16

Sheet _____ of ____3

Date: 2/1/63

S Reviewed and Approved: \mathcal{K} \mathcal{W}

WTPSP TSS# B-91 مراحد[م] مراحد[م]

50)	iorit
(²)	a i
9	et #
Le.	a She
, ک	Data
0_	

																																								Sheet \mathcal{L} of \mathcal{L} 3	
	54	1129	0360		(150	1650	450	218	80	20	3	2	3/4	R	5-			31.9	90.9	3.0	0.347	1150	46	4	0	828	781	829	865	0	A	9	0,5	750	54	2.1	1100	っと	0324	た	
114	CH	7129	1000		1149	1143	399	216	90	20	Z	8	74	τ	7			31.5	92.7	8.2	142'	1150	46	∢	0	625	280	128	863	0	V	MIPNA	هر	129	2	کررا	100	77	0211	ercept whe Ver	
	641	129	5 0 10		9411	1157	385	216	40*	۰.	Z	7	_	z	de .2612			31.7	96.6	3,1	325	1150	49	*	0	825	108 to 5 781	512	86.3	0	4	10	<i>w.</i> 5	229	3	2.2	616	bl.	0110	x couched w	
¢	and the	12/22	1200		1150	וועב	399	214	90	22	N	2	3/4	z	Dries	-		31.5	91.5	2.8	-347	1150	46	Å	٥	825	7%6	822	863	٥	٩	0	٥	750	64	212	100	79	L200		
ata 🔿	A.	128	1 622		1149	1157	413	224	85	40	2	ZBubbe	3/44	2	Delps	-		<u>3</u> 2.5	929	3.0	245	1150	47	A	0	825	740	922	862	٥	4	٥	212	165	54	2.2	1100	74	2022		
stripg) Da	<i>i</i>	821-1	2012		1149	1154	405	רוצ	80	z	2-	142	3/4	z	Dsie	-		33.4	95.9	3.2	1348	1150	НŚ	РА	0	825	766	808	854	0	æ	9	0,5 1	22	54	2,2	1100	78	2106	 	
ty & Elec	AN.	21290	1007		11,50	1150	420	218	85	30	~	2+15	3/4	2	D; 05	"		32.5	904	2.5	1362	1150	46	A	0	825	755	797	546	0	Ð	0	9.0	2002	58	1.8	1050	ف	2011	16103	
1: Rriori	ark	12/1/	1067		6411	1149	2142	214	06	30	2	2 Bubbler	"-	Z	2			32.6	2:58	2.7	1363	1150	46	¥	0	<u>ړ د</u> ح	756	Stor	846	0	A	3 2	و ة	750	Sg	1 8	1050	92	2061	Date: 2	
Data Sheet #			Danco	Ivalige	1125 - 1175	1125-1175	400 - 460	200-250 🕅	> 85			Street capitolity										1150				850	aller to second the			and the second second				750			1150				
)	By:	Date: /	1 IME	01110	ပ	υ °	с °	ပ	%	%	y or n	#	inch	y or n	y or n			Volts	Amps	kW	a	ပ	%	A or M	kW	ပိ	ာ	ပိ	° C	%	A or M	% / NA	Ϋ́	ပ	%	kΜ	ပ	%	hh:mm		
		RSM, July 2003	DESCRIPTION	Driority Data	Melt Temperature (T1, control)	Melt Temperature (T2)	Plenum Temperature	Post FC Off-gas Temperature	Cold Cap Coverage	Slurry Pool Coverage of Cap	Cold Cap Flexibility Visible	Number of Vents	Cold Cap Thickness	Phase Separation (Note in LRB)	Glass Pouring		Electrical Data	Electrode Potential	Electrode Current	Electrode Power	Melt Resistance	Melt (Electrode) Setpoint Temp	(Electrode Power) Output	(Electrode) Control Mode	Kiln Power	Kiin Temp Setpoint	Kiin Actual (Upper) Temp	Kiln Actual (Middle) Temp	Kiln Actual (Lower) Temp	Kiin Power Output	Kiln Control Mode	Kiln Variac Setting (or NA)	Discharge Can Power	Discharge Can Temp Setpoint	Discharge Can Power Output	Overflow Heater Power	Overflow Heater Setpoint	Overflow Heater Power Output	Time	proved: Ruf Zo 2	

Reviewed and Approved:

TP-RPP-WTP-252

WTPSP TSS# B-91

Sheet Z of 12

ar g

150

500

2

0,6

0.6

0.7

ð

A or M % / NA

130

750

Ċ

Discharge Can Temp Setpoint

Kiln Variac Setting (or NA)

Kiln Control Mode

Discharge Can Power

Discharge Can Power Output

Overflow Heater Power

%

Ş

Ņ

₹ ပ

9

216

2

0 Q

Š

0¢

868

361

ပ

%

0

0

00

90

19

18

B

100

200

1150

%

Overflow Heater Setpoint Overflow Heater Power Output

2,0

1109

1031

0220 2020

001

CH57

Time hh:mm

Date: 3/6/63

في 42 Reviewed and Approved: 2

TP-RPP-WTP-252

WTPSP TSS# B-91

5年 21164-14 PUR 1905 8081 2081 211 31.7 Drips 8º 4 2159 es S 멅 25 22 2 5 d ЪС ধন ñ ŝ Drips 1804 1706 2005 415 853 20 50 200 12/29 152 1351 825 \$15 200 750 31.8 3 46 0 201 0 415 25 800 111 1 Depe 7125 125 255 89.0 252 150 31.7 SiO 20 814 t Se al 6 ISI Z 32.4 825 D-105 0.5 30 ŝ 05 -354 888 1152 222 ok õ G 3 50 234 0111 9.28 365 1153 NDS: \$25 ы č 1150 4/29 1150 57 Rz 83 8 0 õ 6 Data Sheet #1: Priority & Electrical Data (パワ めの か) フノス報路 712463 パンヨッジ (ろのう 1402 1616 1 2 Bubbles 254.5 500 رزئز 5 051 150 \$28 20 1161 1142 0 757 53 \$ Ň どし จี 3041 405 KO 80 N N Dubbled Dubler Ed al ŝ Seg S 6 1149 7 F ¢ 2 ζ 0.5 750 7 55 866 1148 1-12 oxe 12 3 2 đ Å 1125 - 1175 1125 - 1175 400 - 450 200-250 Range 1150 1150 850 > 85 750 A or M kW A or M % / NA 0 UNITS y or n y or n Amps inch Volts ç v or n ပ c ပ ပ ပ ပ္ပပ္ ₹ ů ₹ % ₹ % ပ % % %% a # By: Date: Time Meit Resistance Meit (Electrode) Serpoint Temp (Electrode Power) Output (Electrode) Control Mode Discharge Can Power Output Overflow Heater Power Overflow Heater Setpoint Overflow Heater Power Output Cold Cap Thickness Phase Separation (Note in LRB) Glass Pouring Melt Temperature (T1, control) Melt Temperature (T2) Plenum Temperature Post FC Off-gas Temperature Discharge Can Temp Setpoint Cold Cap Coverage Slurry Pool Coverage of Cap Cold Cap Flexibility Visible Kiin Actual (Upper) Temp Kiin Actual (Middle) Temp Kiin Actual (Lower) Temp Kiin Power Output Kiln Variac Setting (or NA) RSM, July 2003 **Electrical Data** DESCRIPTION **Priority Data** Discharge Can Power Kiln Temp Setpoint Electrode Potential Kiln Control Mode Electrode Current Number of Vents Electrode Power Kiln Power

TP-RPP-WTP-252

榷

Sheet 4 of 22

1619

Time hh:mm

Date: 3/6/6)

Gal.

Reviewed and Approved: 2 4

WTPSP TSS# B-91

los a 2255 2555 2555 122 122 122 123 لالكنام 3.4.2 11.55 243 NIA 1144 404 394 0.0 ße 077 ۲ T 5 31.1 d 0 z 246 8125 1861 101.0 11 50 NA 1160 55-1700 4.12 ÞĘ 11 21 80 0 ¢ 5 С 2 1150 0.354 0305 1150 Derps 815 864 864 32.3 89.4 417 250 0 H30 2 3/6/ 5 0 01 2.9 3.0 23.2 7/30 0218 Deres 384 441 149 825 869 869 885 20 114 0 RU2 50 0 9 244 825 292 NR115 00:10 1149 314 263 1154 0%/2 30,8 90.1 5 0 Ł ¢. Data Sheet #1: Priority & Electrical Data 1/30 237.57 Deips 0 0 825 825 814 823 210 1152 347 و 0 1149 70.3 2.8 1150 SL Re 13 393 33 44 0 ≰ 59.4 90.7 2.9 2.9 8 .344 2305 1120 62/5/ 0.20 0.6 1150 31.6 \$55 587 5711 3×211 232 থ 0⊄ 3ेव 0.348 e v 20.12 1151 384 240 2002 4.25 8(J) ON H 74-1 32 1125 - 1175 1125 - 1175 400 - 450 200-250 > 85 Range 1150 850 750 By: Date: Time UNITS A or M kW % A or M % / NA V OL N y or n y or n Volts Amps ပ္ပံုိ % % inch ပ Š ပ ပ္ ξ ပ ပ ပ ပ a % # Melt Resistance Melt (Electrode) Setpoint Temp (Electrode Power) Output (Electrode) Control Mode Kiln Power Cold Cap Thickness Phase Separation (Note in LRB) Glass Pouring Discharge Can Temp Setpoint Discharge Can Power Output Melt Temperature (T1, control) Melt Temperature (T2) Plenum Temperature Post FC Off-gas Temperature Slurry Pool Coverage of Cap Cold Cap Flexibility Visible Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp Kiln Actual (Lower) Temp Kiln Power Output Kiln Variac Setting (or NA) RSM, July 2003 Electrical Data DESCRIPTION **Priority Data** Discharge Can Power Cold Cap Coverage Kiln Temp Setpoint Electrode Potential Kiln Control Mode Electrode Current Number of Vents Electrode Power

TP-RPP-WTP-252

Date: 3/1/63

Sheet 5 of 13

200

11 58

2.8

2

40

0

ଞ୍ଚ

3

90

200

99

20

90

96

%

Overflow Heater Power Overflow Heater Setpoint Overflow Heater Power Output

1-cive

Time hh:mm

150

162

1150

1150

l₹ ပ

%

いい

よん

Reviewed and Approved: Rulface

WTPSP TSS# B-91

WTPSF

7/30 247.5 120 825 22.23 32.4 370 220 19/01 152 <u>ष्ठ</u> = 096 1099 1113 1109 1411 30 410 86.9 コレル B.-b 54 85 drip Δ S Ð ž 2 35L So YES 150 150/05/0 Duble led B. baler 825 235.5 366 60 7 953 180 1701 33.6 549 60 93.4 28 30 53 20 47 প্র Ř 1 652 5311 -e311 2000 12003 (001) Jan 1300 11903 72003 11303 -1300 11903 0300 0900 1035 1110 810 4.46 0.31 1150 05 9 0 5 D-11119 1150 246 244 246 244 Dutabled bubbler 1150 257 93.3 2.7 3.9 0.349 0.331 1.150 30.6 Se8 2466 £ ¢o ¢ ζ 0 No Pool 869 895 0 8 G 23 Data Sheet #1: Brigrity & Electrical Data Q mon Ø ð 80.0 2090 0212, 8005/2, 1354 1444.556 5110 202 31.6 828 E I X.9 С 1145 1144 \$83 Ş ŝ 203 20 837 مہ دو 20 665 868 5.5 Υ 2 ∢ 0 μų 225 825 389 246 80 89.5 11500 61.25 11 20 2 555 ع 126 1149 22 2.9 52 с С 3 2.8 2 4 23 0 X. 2 1125 - 1175 1125 - 1175 400 - 450 200-250 Range 1150 1150 850 > 85 750 By: Date: UNITS A or M % / NA kW y or n y or n % A or M y or n inch Volts Amps Time hh:mm ပ ပ %} ပပ ပ ŝ ပ ů ° S о ° Š ≷ ပ % % 88 # a Melt Resistance Melt (Electrode) Setpoint Temp (Electrode Power) Output (Electrode) Control Mode Phase Separation (Note in LRB) Melt Temperature (T1, control) Melt Temperature (T2) Plenum Temperature Post FC Off-gas Temperature Cold Cap Coverage Overflow Heater Setpoint Overflow Heater Power Output Discharge Can Temp Setpoint Discharge Can Power Output Slurry Pool Coverage of Cap Cold Cap Flexibility Visible Kiln Actual (Lower) Temp Kiln Power Output Kiln Control Mode Kiln Variac Setting (or NA) Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp RSM, July 2003 DESCRIPTION Priority Data Electrical Data **Overflow Heater Power** Discharge Can Power Cold Cap Thickness Kiln Temp Setpoint Electrode Potential Electrode Current Number of Vents Electrode Power Glass Pouring Kiln Power

TP-RPP-WTP-252

Sheet 6 of 23

Date: 1/1/63

Reviewed and Approved: Ru Ru

WTPSP TSS# B-91

TP-RPP-WTP-252

Dress 0209 1149 0.75 0206 86.5 1150 2000 90 224 86 2 [13] 2 31.1 2 865 7/3 40 С 052 8 55 J 0110 Trips 07/31 32.4 800 838 830 1130 404 ·340 .26 0,3% 828 150 2103 2207 2303 0021 2014 149 25 50 8 8 20 Ø 90.5 32.9 a'le 218 1367 147 8 0 es. B HS 125 7 <u>۲</u>۲ 0 ∢ X 36.4 2300 366 822 2294 21810 1150 UZIL 125 220 1145 5.5 728 825 768 1145 90 0 50 B N C 0.6 6 Z 4022 84.2 ,400 .366 1152 235 250 1138 USIT 0515 175 1150 22 \$74 82 31.3 スン Data Sheet #1: Priority & Electrical Data 101 3 5 \mathcal{S} 825 825 45 43 040 0 t 262.0 750 86.6 934 <u>5</u>2 3.5 2059 1137 1144 37.7 448 そみ \$<u>8</u> 3]41 50 <u>N</u> J. 弘在 đ Ş 22 0 021120 32,0 3.0 2.7 .358 .366 1906 2010 7.50 Delos 231.0 2 Bub 796 794 833 8222 1.62 45 2007 313 825 228 125 ้ง 0 255 330 ZUMEIL 2+2Bul 925 1125 - 1175 |157 1125 - 1175 |156 400 - 450 41 200-250 243 > 85 90 -LST 1125 1904 9211 A \$70 2.7 769<u>7</u> A.O 3 04 9 12 Range 1150 1150 850 750 By: Date: Time UNITS Amps kW % A or M % / NA v or n °C % A or M Time hh:mm y or n y or n Volts ξ kW ပ ပ ပ္ %≷ ပ ပံဂံဂံ×ိ× inch ပ ပ ပ္ဂံိန a # Cold Cap Thickness Phase Separation (Note in LRB) Melt (Electrode) Setpoint Temp Overflow Heater Power Overflow Heater Setpoint Overflow Heater Power Output Discharge Can Temp Setpoint Priority Data Melt Temperature (T1, control) Plenum Temperature Post FC Off-gas Temperature Discharge Can Power Output Slurry Pool Coverage of Cap Cold Cap Flexibility Visible Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp Kiln Actual (Lower) Temp Kiln Power Output Kiln Variac Setting (or NA) RSM, July 2003 Electrical Data (Electrode Power) Output (Electrode) Control Mode DESCRIPTION Aelt Temperature (T2) Discharge Can Power Cold Cap Coverage Kiln Temp Setpoint Electrode Potential Kiin Control Mode Electrode Current Number of Vents Electrode Power Welt Resistance Glass Pouring Kiln Power

Reviewed and Approved: 21/12

Date: 2/1633

Sheet $\overline{\mathcal{F}}$ of $\overline{\mathcal{F}}$

WTPSP TSS# B-91

Data Sheet #1: Priority & Electrical Data

		Bv:		20	C.01	54	54	54	553	CCC D	lever.
	RSM. July 2003	Date:		7/31/03	SO ISIL,	7/31	7614	2/31	7/31/03	7/3/103	7/31/03
		Time		0.306	OHIZ	0506	0000	2050	0820	HIEL	1305
	DESCRIPTION	UNITS	Range		•						
	Priority Data										
	Melt Temperature (T1. control)	ပ	1125 - 1175	1150	1132	1147	1150	1149	1149	6711	1150
	Melt Temperature (T2)	ပ	1125 - 1175	1133	02/1	1145	OHJI	0 H 11	1137	1143	1143
	Plenum Temperature	о.	400 - 450	343	HSR*	420	383	395	418	431	403
	Post FC Off-oas Temperature	0.	200-250	822	242	238	122	022	224	8 0 0	213-
	Cold Can Coverage	%	> 85	90	60 *	20	80	8 S	40	10	20
	Shirry Pool Coverage of Cap	%		25	*es	25	25	25	60	50	80
	Cold Can Flexibility Visible	v or n		2	,-	3~	2	6	N	N	el
	Number of Vents	#		2	К	5	2	2	bubbler	bubbler	bubbler
	Cold Can Thickness	inch		3/4	44	3/4	10/c	hle	3/4	34	/
	Phase Separation (Note in LRB)	v or n	A CONTRACTOR OF A CONTRACTOR A	R	I	2	R	3	<i>ک</i>	2	Z
	Glass Pouring	y or n		10.05	N	12.05	62.05	Devol	gup	and	d np
							•		-		-
	Electrical Data										
	Electrode Potential	Volts		31.6	32.6	32.6	31.3	31.7	31.1	31.9	31./
	Electrode Current	Amps		89.3	81.8	88.4	B2.9	85.6	84.4	616	82.8
	Electrode Power	Ň		2.6	2.7	2.9	2.6	2.2	3,6	8 8 8	2,8
	Melt Resistance	a		0.372	104.	0.360	0.379	0.373	0.362	0,3,6	0.351
	Melt (Electrode) Setpoint Temp	с °	1150	1150	11 50	1150	1150	1150	1150	1150	1150
	(Electrode Power) Output	%		43	M	5 A	12	43	43	97	45
	(Electrode) Control Mode	A or M		¥	Å	H	4	4	4	¢	\$
	Kiln Power	κ		0	0	0	0	0	0	þ	0
	Kiln Temp Setpoint	ပ	850	825	G 25	828	828	825	825	ex	825
	Kiln Actual (Upper) Temp	ပ့	10 Mar 1 Mar	908	617	811	806	806	<u>6</u> 0%	618	818
	Kiln Actual (Middle) Temp	0	Ac.	644	8<3	848	844	844	647	693	\$ <u>\$</u>
	Kiln Actual (Lower) Temp	ບ •		928	88(288	288	881	281	188	886
	Kiln Power Output	%		0	٥	0	Ø	0	0	0	0
	Kiln Control Mode	AorM		4	ヤ	A	¥	A	ŧ	Ŧ	=
	Kiln Variac Setting (or NA)	% / NA		9	NA	10	10	10	0	9	2
	Discharge Can Power	kW		0,8	0,8	0.7	0.7	0.3	Г. О	200	110
	Discharge Can Temp Setpoint	° C	750	800	202	80	80	88	999 	3	200
	Discharge Can Power Output	%	States - States - Same	63	6)	29	63	29	٥	3	كمح
	Overflow Heater Power	kW		2.7	R	2.5	2.5	2.6	310	1	4.10
	Overflow Heater Setpoint	ပ	1150	1125	i(Z)	1125	1125	125	Sel	6	cell
	Overflow Heater Power Output	%		84	89	θς	80	90	<u>a</u>	12	88
										4	1000
	Tim	e hh:mm		0309	0415	0509	000	0709	9600		107
	* That Findhed charging feed p	why tube									
Reviewed and P	pproved: Rul Q. L.		-	Date: 2	16/63						
				-							

TP-RPP-WTP-252

Sheet & of 13

WTPSP TSS# B-91

100 1122 EVIC 0102 8081 6081 1150 200 2.25 25.6 120 5.492 200 2005 2010 Ğ 90 ISIL. 394 2,0 8 50 00 3 2 3 ଷ୍ମ 80.00 Þ 3 R F 32.3 1990 12/2 1.101 CK 12 1137 E 122 0.7 H 2 ST IS 418 335 2008 7131 1133 30.5 1150 \$70 96.5 \$19 5584 555 <u>j</u> 0 ¢ ó 1906 870 215,0 204,0 00 7/3(1120 428 Dripping Drip 21 313 ₹ \$ 883 142 1136 33.7 7:86 202 x 2 M S ß \$2 ¢ 0 810 30.4 6.9 00 855 807 878 1758 5 399 20 935 935 45 7/3/ 149 0 3 Data Sheet #1: Priority A Electrical Data Deipping 2 Bubley 1001 343 904 219.0 1150 1658 806 137 7/3/ 31.9 92.5 3 ž 5 ઝું 149 32 Z 4151 4165 915 212.0 915 212.0 915 212.0 915 215.0 101001/2428 3,1 3,1 6,334 ,334 1160 1150 1160 1150 04010 010 870 870 870 870 0.0 0 805 825 797 799 838 838 838 849 1150 0110 801 Held H 22.7 31.3 96 96.1 ц Ч 131 50 2 ge 03 1125 - 1175 114 年 1125 - 1175 114 8 400 - 450 427 200-250 2-15 > 85 <u>6</u> 0 <u>,</u> 8 1510 1510 2 Range 1150 1150 750 850 A or M % / NA UNITS y or n Amps kW °C % A or M y or n y or n Volts inch ပ ç ₹ ပ ပ ပ္ပံုိ Š ပ ပ ပ % ₹ % % % # a By: Date: Time Discharge Can Power Output Overflow Heater Power Overflow Heater Setpoint Overflow Heater Power Output Cold Cap Thickness Phase Separation (Note in LRB) Glass Pouring Electrode Power Melt Resistance Melt (Electrode) Setpoint Temp Discharge Can Temp Setpoint Melt Temperature (T1, control) Melt Temperature (T2) Post FC Off-gas Temperature Slurry Pool Coverage of Cap Cold Cap Flexibility Visible Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp Kiln Actual (Lower) Temp Kiln Power Output Kiln Variac Setting (or NA) (Electrode Power) Output (Electrode) Control Mode RSM, July 2003 DESCRIPTION Priority Data Electrical Data Discharge Can Power Plenum Temperature Cold Cap Coverage Kiln Temp Setpoint Electrode Potential Kiln Control Mode Electrode Current Number of Vents Kiln Power

sheet 1 of 13

Date: 3/1/62

Reviewed and Approved: Rulled

519 11618

Time hh:mm

TP-RPP-WTP-252

B.19

WTPSP TSS# B-91

1050 0500 040 0526 0704 000000 0.348 150 50.7 90 000 040 1199 34 Ø. 2 6 319 2 3 à 2 0 6 94.8 , <u>30</u> 7 *"* 50 1150 90 30,6 0571 1453 665 1 20 N/4 203 ž 0 825 8.70 <u>رو</u> [-845 ß 200 đ 00 T 0 411 7 ₹ 0301 0465 149 381 30.7 0.343 1150 796 200 \$20 800 85 0.0 3/4 Sin 0 0 9 5 ₹ 81 0.356 1150 69 1150 805 623 394 83 949 2 8 5 ۵ 1/50 2020 41.105 2.6 31.0 90 825 836 897 897 0 3511 a i 0/1/0 2010 143 870 000 ð 0 23 Š, Data Sheet #1: Provity & Electrical Data 358 0.369 0.350 200 30,7 11 50 N 14 870 1147 2024 123 ŝ ß インや 820 μH 825 9 2 2 22 314 1151 9 Σ 32.2 201 860 1150 1141 0.0 5. S 392 00 60 Q 7 2 1125 - 1175 1142 / 1125 - 1175 1145 / 400 - 450 345 -200-250 204.0 26.2 1150 341 825 585 010 34:5 000 \$12 o≮ 2000 54 δ Ζ オンネ 2 Range 1150 1150 > 85 750 850 By: Date: UNITS ma/m³ A or M % / NA A or M kV ° C ၊ပူပ y or n y or n Volts Amps V OT U ° % inch Ċ ပ Ķ ç ပ ပ ပ Š ပ % ₹ 8 a % %% # Cold Cap Thickness Phase Separation (Note in LRB) Glass Pouring Melt Temperature (T1, control) Melt Temperature (T2) Plenum Temperature Post FC Off-gas Temperature Melt (Electrode) Setpoint Temp (Electrode Power) Output (Electrode) Control Mode Overflow Heater Power Overflow Heater Setpoint Overflow Heater Power Output Discharge Can Temp Setpoint Discharge Can Power Output Cold Cap Coverage Slurry Pool Coverage of Cap Hy cont in Creathough Kiln Variac Setting (or NA) Cold Cap Flexibility Visible Temp Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp RSM, July 2003 DESCRIPTION **Priority Data** Discharge Can Power Kiin Power Output Kiin Control Mode Kiln Temp Setpoint Kiln Actual (Lower) Electrode Potential Electrode Current Number of Vents Electrode Power **Welt Resistance** Kiln Power

Reviewed and Approved: \mathcal{N} ω

B.20

TP-RPP-WTP-252

Date: 2/6/63

0110

0003

2302

Time hh:mm

Sheet 10 of 13

WTPSP TSS# B-91

295. 45% 3 1153 30.3 1008 50 21 0 151 2711 000 182.4 175.4 1507 R2 Q M 2% 413 ŝ 3 0 Ś ñ C 0 5 2141 0302 1201 2111 12001 1306 1902 Ś P1150 22.20 202 0000 1410 1148 52,9 743 00 00 0 19 ¢, 406 39 194 1150 29.0 ۹ ۹ ۹ 000 250 1120 100 10 もよ 121 0 92 37, Ð 2 ₹₹ 164.0 1665 95 70 .326 220 2051 27.5 84.9 1154 408 \$28 836 281 و 0 \$ Q 90 3 40 0 3 űþ ¢ Ь コ q 8.9% 1150 \$70 24.7 1110 406 .332 ŝ 0000 825 1143 r o 2 8 0 Data Sheet #1: Priority & Electrical Data 163.0 90 .334 1150 36 865 200 20 Ē 1142 1005 000 5 5 5 권 0 സ് 90 90 29.9 1150 326 0859 1140 1151 000 \$25 900 80 Ż 241 84 413 5 40 0 0 20 0 2 3 185.5 90 90 1150 150 5 404 29.2 925 974 974 1150 000 2.5 C.O 868 45 8 0%0 **4**0 3 11/2 0 0 ∢ 2 1125 - 1175 1125 - 1175 400 - 450 200-250 Range 1150 1150 750 ~ 85 850 By: Date: UNITS A or M A or M AN / % Amps Time hh:mm 0 y or n V or n Volts y or n ы Ч ပ ပ ပ o ပ Š kΝ ပ ပ ပ ₹ % ξ ပ္ 8 a % % # 88 Hy concentration in Freedwarde Cold Cap Thickness Phase Separation (Note in LRB) Glass Pouring Meit Temperature (T1, control) Meit Temperature (T2) Plenum Temperature Post FC Off-gas Temperature Melt (Electrode) Setpoint Temp (Electrode Power) Output (Electrode) Control Mode Overflow Heater Setpoint Overflow Heater Power Output Discharge Can Temp Setpoint Discharge Can Power Output Slurry Pool Coverage of Cap Kiln Variac Setting (or NA) Cold Cap Flexibility Visible Kiln Actual (Upper) Temp Temp Kiln Actual (Middle) Temp RSM, July 2003 **Electrical Data Overflow Heater Power** DESCRIPTION Priority Data Discharge Can Power Kiln Actual (Lower) T Kiln Power Output Kiln Control Mode Cold Cap Coverage Kiln Temp Setpoint Electrode Potential Electrode Current Number of Vents Electrode Power **Welt Resistance** Kiln Power

836 Rug

Date: <u>Z/C/03</u>

Reviewed and Approved: R W Kale

Sheet 11 of 13

B.21

TP-RPP-WTP-252

WTPSP TSS# B-91

0405 , 306 010 90.0 033 1150 K14 4.0 <u></u> ۍ ن 600 000 40 854 392 80 11 54 ۲ 0 <{ 27 0.0 ~ 63 8263 21 0/ 2234 005 0000 405 048 00 149 124 0.31. 2 27. 68 22.30 5 0.0 392 118 19 88 e alle WCB 400 81103 81103 2010 2103 0.0 Ì 1908 3001 3.2 pubbler 1 5.86 0.304 882 HS 0 eg 7 g 909 5 ភ្ 5 J Data Sheet #1: Priority & Electrical Data 208* 201* while bubled 2 E 00 0 86,3 93,3 9 8,5 8,8 9 0,551 0,321 0 1710 1812 <u>8/1/03</u> 1808 3,00 206 130 1150 113 1130 113 1130 1130 1130 1130 1130 1130 N N N N N THE DE 8/1/03 N Sycie 829 876 876 1705 01,0 22 ¢ ф 1125 - 1175 1125 - 1175 400 - 450 200-250 > 85 Range 1150 1150 850 750 2 VII / Chi A or M A or M % / NA UNITS y or n Amps y or n ပ္၀ပ္ပ Volts Time hh:mm с Ъ kχ ပ္ပံုိ Š ပ inch kΝ a°% ပ ပ S % %≩ % % By: Date: Time # H contentration in Greenhunge Cold Cap Thickness Phase Separation (Note in LRB) Melt (Electrode) Setpoint Temp **Overflow Heater Power Output** Aelt Temperature (T1, control) Aelt Temperature (T2) Discharge Can Temp Setpoint Plenum Temperature Post FC Off-gas Temperature Discharge Can Power Output Overflow Heater Power Overflow Heater Setpoint Slurry Pool Coverage of Cap Kiin Actual (Upper) Temp Kiin Actual (Niddle) Temp Kiin Actual (Lower) Temp Kiin Power Output Cold Cap Flexibility Visible Kiln Variac Setting (or NA) Electrode Power) Output (Electrode) Control Mode RSM, July 2003 DESCRIPTION **Priority Data** Discharge Can Power Cold Cap Coverage Kiln Temp Setpoint Electrode Potential Kiln Control Mode Electrode Current Electrode Power Melt Resistance Vumber of Vents Glass Pouring Kiln Power

TP-RPP-WTP-252

* accelently weed off gas the time

Date: 0/1/03

Reviewed and Approved: NUL Park

Sheet 12 of 13

B.22

WTPSP TSS# B-91

.

5.1 3.1 6.246 12 60 prology 108.6 0210 0,000 00000 195 ž Š 1147 12511 200 **R**: 1 30 28.8 28.4 0.7 670 Š 20 20 2/3 В 0 3 ۲ 8 ≻ 4 Data Sheet #1: Priority & Flectrical Pata 107.2 *(* 20 841 ×03 1150 1127 396 2.5 000 54 205 441 5 350 0 217 eg g Z 2 900 ک T 8/1/03 0,000 1257 570 02010307 280 11 20 100 95.9 1(45) 117 102 48 M/A 30 (Jal) 20 0 1 4 2 95 Z 0.000 9:5 27.72 91.9°2 525 305 951 870 2020 346 205 505 11 50 HG 200 54 (67 314 90 6 11:50 6411 1124 417 0 5 20 30 2 ∢∣ 13/2/ Ô 2 1125 - 1175 1125 - 1175 400 - 450 200-250 > 85 Range 1150 1150 850 750 A or M By: Date: UNITS Hy concentration in Erechment has and y or n Amps kW % A or M % / NA y or n y or n Volts Time hh:mm inch aပ ပ ပ Š ပ ပ္ပပ္ပပ္ ပ ပ %₹ ပ 88 % # Plenum Temperature Post FC Off-gas Temperature Cold Cap Coverage Slurry Pool Coverage of Cap Phase Separation (Note in LRB) Glass Pouring Mett (Electrode) Setpoint Temp (Electrode Power) Output (Electrode) Control Mode Overflow Heater Setpoint Overflow Heater Power Output Discharge Can Temp Setpoint Meit Temperature (T1, control) Meit Temperature (T2) Discharge Can Power Output Cold Cap Flexibility Visible Kiln Variac Setting (or NA) Kiln Actual (Upper) Temp Kiln Actual (Middle) Temp Kiln Actual (Lower) Temp Kiln Power Output RSM, July 2003 **Overflow Heater Power** DESCRIPTION **Priority Data** Discharge Can Power Cold Cap Thickness Kiln Temp Setpoint Electrode Potential Electrode Current Electrode Power Kiln Control Mode Number of Vents Melt Resistance Kiln Power

Sheet 13 of 23

Date: 8/6/03

Reviewed and Approved: Rul 9

ТР-КРР-МТР-252
Date Sheet #2: Routine Status Sheets

WTPSP TSS# 91

TI-RPP-WTP-264

للمر

			Data Sh	eet #2: R	outine St	atus She	t l				
			By:	4b3	H.D.S.	H.D.S.	web.	3	N.	AL.	CRA-
	RSM, July 2003		Date:	7/28	7/28	7/28	7/28/03	7003	4/28	32/2	2/20
			Time:	10:00	11:06	12:03	1415	0101	18/1	1305	2010
	DESCRIPTION	units	range								
	Feed Nozzle Cooling Flow	gpm	.5 - 1(reg FNT)	21-0	0,75	0.8	0, X	<u>8</u> .9	0,4	6.8	98
	Fitm Cooler Air Supply	scfm	1-4 (reg OGT)		2.1	31	3.1	2.9	2.7	2.9	2.9
	Film Cooler Pressure	psig	20-50	2.5.	2	2	2.5	515	γ	2	2
	Melter Vacuum-Magnehelic	in. H2O	0.1 - 2.0	1:6-	3.4	2.5	214	3.1	311	2.0	3.
	EVS AP	in. H2O		Q	0	0	Q	0	٥	0	0
	Film Cooler AP	in. H2O		0.1	Q	0	0	6	٥	٥	0
	HEME <u>AP</u>	in. H2O		0.3	0.2	0.2	0,2	える	ہ ہ	2.0	0 U
	System AP	in. H2O		0	0	0	0	0	٥	٥	0
	EVS HX Cooling Flow	gpm	1-5 (reg SLT)	2.8	1.9	2.0	<i>a.</i> 0	2,1	2.1	210	6 9
	EVS Scrub Tank Volume	liters	60-220	06	90	90	40	92	32	ہ ح	6
	EVS Nozzle Pressure	psi	50 - 55	53	53	5-3	53	53	53	ß	2
	EVS Nozzle Flow	db		3.1	3.(3.2	<u>3</u> ,1	3,1	3, [3,1	2
	EVS Scrub Solution pH	На	500 184 1943	1	١	1		Ļ	1	-	ļ
	OG Control Valve position	%	ille office of the	80	0	-	0	0	0	0	0
	OG Control Valve mode	A or M	36	đ	ø	Å	H	Ч Ч	W	Ł	٤
	Feed Pump (tubing) Condition	Chg/Pull	des ab	J	ſ	1	oκ	¥	٥K	20	٩ م
	Recirc Pump Tubing (Master)	Chg/Pull		1	6	ι	¢Κ	P.C.	оκ	٥k	2.0
	Recirc Pump Tubing (DayTnk)	Chg/Pull		١	1	١	oκ	or the	, Хo	0 K	9Å
1-10	Day Tnk Agitator Setting	%	100 100 100 100 100 100 100 100 100 100	Ŋ	NC	NC	NC	r/C	Ц	£	2
201-0	Master Tank Agitator Setting	%		0		01	70	20	ЧL	20	40
)	Blower Cooling Flow	gpm	1-1.5		1.4	0.1	1.6	1.9	1,5	118	1.8
	Feed Nozzle Temp (FNT)	ပိ	< 40	29	28	27	38	38	2A	28	2J
	Melter Off-Gas Jumper Temp	ပိ	200 - 250	227	215	180	343	234	217	215	218
	Post EVS Off-Gas Temp	° C	< 50	32	34	34	ઝઠ	36	54	34	34
	Post HEME Off-Gas Temp	ပိ	100	30	32	33	34	36	36	36	30
	Hg-Bed Temperature (or NA)	° C / NA	100	NA	NА	Ч Ч	NA	XIX	1	-	
	Scrub Liquid Temp (SLT)	ပ	A	305	34.1	34.5	36	33.57	34	34	47.
	EVS HX Cooling Temp	ပိ	≤30	32	34.0	34	ŝ	34	34	34	34
	Feed Pump Setting	%		3	w	Ũ	6	07	9,10	0/26	9.0
	Feed Pump Control Mode	A or M		đ	£	đ	æ	¢.	Ø	A	
	Feed Pressure	psi	<1.0	ſ	(ł	١		1	ł	•
	Day Tank Wt	, Kg	崎 <83 Net	1(9.3	117.5	115.5	110rgK	107.92	106.17	105.0	10 7:35
	Master Feed Tank Wt	The state	<300 Net	46	410	400	100	40%	40%	408	202
	Overflow Temp	°C	1100 - 1150	1051	1048	1050	1051	1050	1050	1055	KOS0
	Discharge Can Temp	ပ	750 - 850	380	HUM	205	155	199	220	750	050
	Glass Scale	Ъ В	< 10	0	0,58	1.15	1.99	4405	4.26	4.51	5.00
	Alarm Condition	On/Off	off	0 H	РĢ	J	7	AL .	04	1	т 71 71
	time	աա։կկ		10:31			1440	0211,	1823		2018

TP-RPP-WTP-252

Date: 2/c/03

Reviewed and Approved: $\mathcal{IA}\mathcal{U}\mathcal{P}_{\mathcal{A}}\mathcal{L}_{\mathcal{A}}$

WTPSP TSS# 91

20 2 2 3.6 200 0 0.06 0.6 0 200 P 10 2/4 0412 05 90,0 0.0 R212 0313 24 200 0319 0.3 0.77 0.Z 2.4 ok 201 214 0 06 0 QK য 1220 5.5 21B 214 1.5 ষ ξ 0 0.2 0120 0 8 8 7/29 7.23 25.61 80% 0112 s S S 8 Routine Status Shee 7129 3 C ΦN 408 212 500 0 . いい 5 8 0 ۶ С ş 0 4 99,14 40% 7128 g 3 210 ふろ 5 8 223 Z 22 5 05 3 0 2 20 0 12:10 ŝ 217 749 -Bata Sheet #2: 3,2 2 22 to B NC 404 3 200 ₹ Z ð 0 39 5 C đ 5 range 5 .5 - 1(reg FNT) 1-4 (reg OGT) 1-5 (reg SLT) 60-220 By: AU Date 7125 Time: 2106 1100 - 1150 750 - 850 200 - 250 <300 Net 0.1 - 2.0 50 - 55 <83 Net < 10 < 20-50 1-1.5 <u>^</u>1.0 < 40 < 50 < Ř <u>8</u>8 ₽0 ff
 Film Cooler AP
 In. H2O

 HEME AP
 In. H2O

 System AP
 In. H2O

 System AP
 In. H2O

 EVS HX cooling Flow
 In. H2O

 EVS Strub Tank Volume
 Ines

 EVS Nozzle Pressure
 ppm

 EVS Nozzle Pressure
 ppm

 EVS Nozzle Flow
 ppm

 EVS Scrub Solution pH
 pH

 OG control Valve mode
 A or M

 Peed Pump (tubing) Condition PH
 pH

 OG Control Valve mode
 A or M

 Recirc Pump Tubing (Master)
 ChgPull

 Recirc Pump Tubing (DayTink)
 ChgPull

 Recirc Pump Tubing (DayTink)
 ChgPull

 Diower Cooling Flow
 gpm

 Feed Nozzle Temp (FNT)
 ° C

 Peed Nozzle Temp (FNT)
 ° C

 Post HEME Off-Gas Jumper Temp
 ° C

 psig in. H2O in. H2O in. H2O in. H2O On/Off scfm time hh:mm is p p ပ ပ Kg units Feed Nozzle Cooling Flow Film Cooler Air Supply Film Cooler Pressure Melter Vacuum-Magnehelic RSM, July 2003 Overflow Temp Discharge Can Temp Glass Scale DESCRIPTION Day Tank Wt Master Feed Tank Wt Alarm Condition EVS AP

TP-RPP-WTP-252

Sheet 2 of 13

Je Co

Date: 0/1

NC - No Change

Reviewed and Approved: Reviewed

WTPSP TSS# 91

TP-RPP-WTP-252

	were word an web.	50/57/ 62/12 60/60/L	1109 1157 1300 1406		0.8 0.840 0.83	24 2,4 2,3 2,5	4 ~ ~	3,8 3,8 3,6 80	000	0000	0.3 0.5 0.2	000	2.0 2.0 2.0	95 95 45	53 53 53 53	31 31 3.2 3.1		0 0 0	<u>۲</u> ۲	or or or or	of of of of	or or or	NC NC NC NC	NC NC NC	- 22 11 SS- 1-9	21, 21, 21 21	246 238 234 235	34 34 34 34	32 32 34 35	NA NA NA NA	34 50 37 34	84 34 33 34	0.0	T	1.1 1.1 E 1 1.0	P-76 10/82 12.64 1.18	400 408 408 408	133 1149 1151 1150	158 758 750 750	194 4.84 5.39 2.2	
et	220	2/20/6	1015		8	ais	4	ର ଜ	0	0	2.0	0	6.1	8 S	ŝ	3.6	1	٥	হ	ok	ØK	DK	Š	ž	1.8	9.1	229	32	32	۲	32	22	8.8	¢	-	22.6	408	1149	9 5 7	OLAH	
tatus She	axu	712.910	09/1		08	3.6	ě	3,4	0	0	0,3	0	18	95	53	6	1	0	N	٥k	۵K	0 X	N'K	NC	1.10	51	116	32	30	NA	30	32	9,5	А	7	83, 2	H04	0011	1750	SILE	2
outine	WXD.	7129105	0803		L'0	are	H	3,4	0	0	6.3	0	1.8	qS	53	312		0	M	NO VO	ž	DK	2	NC	91	36	313	32	30	NA	24	32	9.5	4	t. 8	84.8	409	0011	6h6	1.81	100
et #2: R	\$4	2/00/03	0207		0.7	2.6	2	3.2	0	0	2.0	0	1.8	95	53	3.2	ţ	0	М	06	06.	06	40	30	1.5	53	214	32	30	N/A	32	32	9.5	Å	1.9	86.1	409	0011	750	1.39	1
Data She	By:	Date:	Time:	range	.5 - 1(reg FNT)	1-4 (reg OGT)	20-50	0.1 - 2.0		A DEPOSIT OF MALE DAVID			1-5 (reg SLT)	60-220	50 - 55								Service and the service of the servi		1 - 1.5	< 40	200 - 250	< 50	100	100	0₽	≤30			<1.0	<83 Net	<300 Net	1100 - 1150	750 - 850	< 10	5
				units	mdg	scfm	psig	in. H2O	in. H2O	in. H2O	in. H2O	in. H2O	mdg	liters	psi	gpm	Hq	%	A or M	Chq/Pull	Chq/Pull	Cha/Pull	%	%	mdp	с °	ပ	υ ·	ပ	°C/NA	ပ	ပိ	%	A or M	psi	Кg	Кg		с °	Кg	201
		RSM, July 2003		DESCRIPTION	Feed Nozzle Cooling Flow	Film Cooler Air Supply	Film Cooler Pressure	Melter Vacuum-Magnehelic	EVS AP	Film Cooler AP	HEME AP	Svstem AP	EVS HX Cooling Flow	EVS Scrub Tank Volume	EVS Nozzle Pressure	EVS Nozzle Flow	EVS Scrub Solution pH	OG Control Valve position	OG Control Valve mode	Feed Pump (tubing) Condition	Recirc Pump Tubing (Master)	Recirc Pump Tubing (DavTnk)	Dav Tnk Aoitator Setting	Master Tank Agitator Setting	Blower Cooling Flow	Feed Nozzle Temp (FNT)	Melter Off-Gas Jumper Temp	Post EVS Off-Gas Temp	Post HEME Off-Gas Temp	Hg-Bed Temperature (or NA)	Scrub Liquid Temp (SLT)	EVS HX Cooling Temp	Feed Pump Setting	Feed Pump Control Mode	Feed Pressure	Dav Tank Wt	Master Feed Tank Wt	Overflow Temp	Discharge Can Temp	Glass Scale	·· > ···

Date: 2/6/03

Sheet 2 of 13

Reviewed and Approved: $\mathcal{R}_{k}/\mathcal{S}_{z}$

WTPSP TSS# 91

250 749 758 9.33 9.95 0.36 2200 2300 drapping of 20222 65.7 64.28 62.84 406 0,15 34 0200055 3 200 22 2 4 10.0 2.5 240 406 5.0 53 3.4 3.2 0 5 ð 0 ¢, 0 Ł δ ð Ο Cot or 2100 0.8 2.1 10,5 406 0.3 20 g 1.4 0 ALC 0 0 ข้ 1159 N.C. N.C. 72.87 71.25 69.47 68.0 Clear 0 272 8.37 8.45 3.4 50 297 21-2 00-25 0 Ł গুপু 0 싱 34 3 4 0 8 1/29 1900 406 72 130 6/000 241 5 3.9 Å 2.1 55 ZY 34 ~ ^ ŋ 5 b 0 6 Routine Status Sheet 14.5 NC ų Z 750 Cler 42 121 244 ۹ 3 7,50 1754 57 3.0 4 40C r M 3 6.2 22 かる 0 Xp 0 ٤ 2 QX ð 12963 17/29/03 35 750 J.F 2,0 222 1 en 107 147 217 ٥Ę र्ष গ্ৰহ O 35 oκ 1 9 Ś ત્ર 74.53 No Ch. ON Alan \$5 407 J 0 S ŝ of the 22 E 5 JN N J 10 12 500 ц ふう 331 € õ Ó range 5 - 1(reg FNT) 1.4 (reg OGT) 20-50 0.1 - 2.0 1-5 (reg SLT) 60-220 50 - 55 Data Sh 1100 - 1150 750 - 850 200 - 250 <83 Net <300 Net 1 - 1.5 9 v 40 <1.0 50 Ñ 100 Å ₽f 100 By: Date: Time: psig in. H2O in. H2O in. H2O in. H2O in. H2O °C % A or M On/Off A or M Chg/Pull Chg/Pul °C/NA time hh:mm gpm scfm gpm liters Chg/Pu gpm is d ပ ပ ပ ပ <u>.</u> 822° ç ς υ units X 퓐% %
 DESCRIPTION
 unit

 Feed Nozzle Cooling Flow
 9

 Film Cooler Arr Supply
 5

 Film Cooler Arr Supply
 5

 Metter Vacuum-Magnehelic
 in.

 FILM Cooler Arr Supply
 6

 Metter Vacuum-Magnehelic
 in.

 FILM Cooler Arr Supply
 in.

 FILM Cooler Arr Supply
 in.

 FILM Cooler Arr Supply
 in.

 FY Stem Arr
 in.

 System Arr
 in.

 Strub Solution Pith
 in.

 Feed Pump Tubing (Master)
 in.

 Recirc Pump Tubing (Master)
 in.

 Master Tank Agitator Setting
 in.

 Master Tank Agitato RSM, July 2003 Narm Condition

TP-RPP-WTP-252

Reviewed and Approved: UL UL

Sheet 4 of 13

Date: 3/6/63

WTPSP TSS# 91

12.30 0.2 04 2.4 3.6 3.2 (50 00 0 0 249 0 5 5 10 0 0 0 0 77.35 26.05 0.72 0.72 244 360 360 0.3 6.1 50 0.1 27 24 114 0 3 3 . OK ð 5 01/2: 0406 ۲, ۲ 2 3 53.5 2.58 s, v 3 24 21 1 0 Ò Ð ok z ž 14,05 360 111 St. 2 -1150 3 Ś 4 546 8 ž Ì 3,2 0 4 Š 2 7 óκ ß 20 0.3 0 C Data Sheet #2: Routine Status Sheet 0220 0220 360 360 3.2 06. 250 + 0.75 5 2 0.0 214 0 0 2 20 т 17 5 114 360 36 2.0 2000 35 0.6 R 0 28 0 20 Z 4/14 ٥ X 0 7 0 34 9,0% 81.78 251 360 7/30 6.75 0 45 N.C. ソキ 2012 r Z 3.7 ٤ d 3 S 0,2 2 34 M 24 1100 - 1150 .5 - 1(reg FNT 1-4 (reg OGT) 1-5 (reg SLT) 750 - 850 <300 Net 200 - 250 <83 Net 0.1 - 2.0 range <1.0 20-50 50 - 55 < 10 < 60-220 1-1.5 °26664 8 < 40 By: Date: Time: Kg On/Off A or M psig in. H2O in. H2O in. H2O in. H2O in. H2O Chg/Pull Chg/Pull Chg/Pull gpm liters A or M C / NA gpm scfm is m Ha dpm 0 Kg Kg ပ ပ ပ ပ с С ပ ပ units % R % % % EVS Scrub Solution PH EVS Scrub Solution PH OG Control Valve position OG Control Valve mode Feed Pump (tubing) Condition Recirc Pump Tubing (Master) Recirc Pump Tubing (DayTnk) C1 Day Tnk Agitator Setting Master Tank Agitator Setting Blower Cooling Flow Feed Nozzle Temp (FNT) Melter Off-Gas Jumper Temp Post EVS Off-Gas Temp Post HEME Off-Gas Temp Post HEME Off-Gas Temp Scrub Liquid Temp (SLT) EVS HX Cooling Temp Feed Pump Setting Feed Pump Control Mode Feed Nozzle Cooling Flow Film Cooler Air Supply Film Cooler Pressure Metter Vacuum-Magnehelic EVS AP Film Cooler AP HEME AP System AP EVS HX Cooling Flow EVS Nozzle Pressure EVS Nozzle Flow RSM, July 2003 DESCRIPTION Day Tank Wt Master Feed Tank Wt Overflow Temp Discharge Can Temp eed Pressure Glass Scale

TP-RPP-WTP-252

050

0414

0505

6220 2110

00

OFF

cleas

off

time hh:mm

Alarm Condition

Date: 2/6/3

Reviewed and Approved: K 4

WTPSP TSS# 91

2002025025 229 0 20 2000 000 01/0 01/0 S 0 নিম্থ ব্যুণ্ডমু 013 101 9.050 125 ē 200 NONE CAN IOZY 283 1. 0 5 0 365 14.07 لأرافح 0.75 (30 12 1/30/03 1/30/03 JJ.ZZ 00000 moral 0200 292 clear 5.0 רות 0 Q 9 ¢ 32 262 362 11 262 362 140 14 500 500 246 238 00 どの 3 d Dκ Z OK 3 Ň ZOOOZZ - mail Dic Data Sheet #2: Routine Status Shee 010 1020 0 v 0.0 3 10 264 10 40.00 COUL 0.7 active 1:0 Wind - Per 3 mm 2 022200 2 N N ž ٢ 3236 -19 0.0 5 9 0 STE. 80025 361 1-4 (reg OGT) 20-50 .5 - 1(reg FNT) gpm 1-5 (reg SLT) liters 60-220 1100 - 1150 750 - 850 <300 Net <83 Net 200 - 250 50 - 55 0.1 - 2.0 range 1 - 1.5 < 40 9 v < 50 55888 v.1 ц, By: Date: Time:
 EVS Scrub Solution pH
 pH

 OG Control Valve position
 %

 OG Control Valve position
 %

 Feed Countol Valve mode
 A or M

 Recirc Pump (ubing) Condition
 CopPull

 Recirc Pump Tubing (Master)
 Chg/Pull

 Recirc Pump Tubing (DayTnk)
 Chg/Pull
 17 K94 (1 A or M psig in. H2O in. H2O in. H2O On/Off in. H2O in. H2O scfm C/NA mdg psi gpm gpm ပ ပ ပ psi Ċ ပ 집 ပ units ပ % % % Feed Nozzle Temp (FNT) Metter Off-Gas Jumper Temp Post EVS Off-Gas Temp Post HEME Off-Gas Temp Hg-Bed Temperature (or NA) Aaster Tank Agitator Setting Melter Vacuum-Magnehelic EVS HX Cooling Flow EVS Scrub Tank Volume EVS Nozzle Pressure Feed Pump Control Mode eed Nozzle Cooling Flow Scrub Liquid Temp (SLT) Day Tnk Agitator Setting RSM, July 2003 DESCRIPTION EVS HX Cooling Temp Film Cooler Air Supply Master Feed Tank Wt Discharge Can Temp Film Cooler Pressure **Blower Cooling Flow** Feed Pump Setting EVS Nozzle Flow EVS AP Film Cooler AP HEME AP Alarm Condition **Verflow Temp** Feed Pressure Jay Tank Wt Glass Scale ystern **AP**

TP-RPP-WTP-252

* montolité due to procese demonstre

Sheet 6 of 13

20%

1025

2237

time hh:mm

Date: 3/1/63

Reviewed and Approved:

WTPSP TSS# 91

0.65 1.1 100 5.2.2 33.0 0418 0.0 12.0 12 4 4.1 97.99 97. 5.70 4.3 3 32 040 255 в J 1/2 5 217.5 0 61 W/W 0306 100 0211 200 6 3 ź アレ 3 11 5 0 0 Ł ó 99.15 53.1 Zit 220 0215 0.65 69 5.2 pulled 10 3 \$ 7 Ð 2 2 29 JH ۰. ۲ ٥ 3 ž 4 15) (`` 30 02.001 2.82 Clear 750 3.4 0.2 2312 0019 0109 ß 00 34 Ø Sul 010 20 05 Ś 8 2.50 2:2 102.10 CLERS CLERV 5 12 50 218 2100 75.20 1 Å L N/4 0 11 Data Sheet #2: Routine Status, Sheet 105.63 103.57 510 749 N S 0% С 0 220 50 アレイ ろん 9~ 00 Ω 0 μ Ó 好ら ξ 10 0 0 1.32 752 clear 7/30 2.32 124 0.65 ころ 10 m2/10 314 22 OZ 202 _ R 35 기 0 ŝ 10,6 Q N 4 ž 36 Ц 0 30 11-1 1-1 248 اک ک 07.57 2103 750 clars ې 6,3 217 5 SiH 27 S C K C K 0 0 0 Σ 36 ŝ 7 4 30 range .5 - 1(reg FNT) 1-4 (reg OGT) 20-50 1-5 (reg SLT) 60-220 50 - 55 1100 - 1150 <83 Net</p><300 Net</p> 750 - 850 200 - 250 0.1 - 2.0 <1.0 < 10 1 - 1.5 6 6 6 8 8 < 50 < < 40 off By: Date: Time: psig in. H2O in. H2O in. H2O in. H2O A or M °C/NA On/Off A or M Chg/Pull Chg/Pull Chg/Pul gpm liters gpm scfm gpm °C 비 비 씨 S S S S S ပ ပ ပ ç ပ္ပ Å S bsi ç units 8 8
 Film Cooler AP
 in.

 HEME AP
 System AP
 in.

 System AP
 EVS HX Cooling Flow
 in.

 EVS HX Cooling Flow
 EVS Storub Tank Volume
 in.

 EVS Nozzle Pressure
 EVS Storub Tank Volume
 in.

 EVS Storub Solution pH
 OG control Valve position
 A

 Rescire Pump Tubing (Master)
 Ci
 Control Valve position

 DG Control Valve position
 OG control Valve position
 Ci

 Preed Pump (tubing) Condition
 Ci
 Part Applicator Setting

 Recire Pump Tubing (Master)
 Ci
 Day Tnk Applicator Setting

 Master Tank Applicator Setting
 Blower Cooling Flow
 Post HEME Off-Gas Temp

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME

 Post HEME Off-Gas Temp
 Post HEME Off-Gas Temp
 Post HEME

 Post HEME Off-Gas Temp
 Post HEME
 Post HEME

 Post HEME Feed Nozzle Cooling Flow Film Cooler Air Supply Film Cooler Pressure Melter Vacuum-Magnehelic Overflow Temp Discharge Can Temp Glass Scale RSM, July 2003 DESCRIPTION Day Tank Wt Master Feed Tank Wt Alarm Condition VS AP

TP-RPP-WTP-252

B.31

Date: RULOJ

Reviewed and Approved: 24

2107

time hh:mm

Sheet 7 of 13

WTPSP TSS# 91

2 0 205 3.0 0 0 C C 108/18 0.6 30 ā Data Sheet #2: Routine Status, whee 0,60 5 0 53.6 2.0 いう 00 0.60 200 5 Ó. 0.62 52.5 Q C õ .5 - 1(reg FNT) 1-4 (reg OGT) 20-50 1-5 (reg SLT) 60-220 50 - 55 0.1 - 2.0 range By: Date: Time: psig in. H2O in. H2O in. H2O in. H2O in. H2O gpm liters mdg scfm gpm units . S
 System AP
 in

 EVS HX Cooling Flow
 i

 EVS Scrub Tank Volume
 EVS Nozzle Fressure

 EVS Nozzle Fressure
 EVS Nozzle Fressure

 EVS Nozzle Fressure
 EVS Nozzle Fressure

 EVS Nozzle Fressure
 EVS Nozzle Fressure

 EVS Scrub Solution pH
 OG Control Valve position

 OG Control Valve position
 P

 Peed Pump (ubing) Condition
 C

 Recirc Pump Tubing (Master)
 C

 Recirc Pump Tubing (DayTnk)
 C

 Post EVS Off-Gas Jumper Temp
 Post EVS Off-Gas Jumper Temp

 Post EVS Off-Gas Jumper Temp
 Post EVS Off-Gas Jumper Temp

 Post EVS Off-Gas Jumper Temp
 Post EVS

 Post EVS Off-Gas Jumper Temp
 Post EVS
 </tr

Feed Nozzle Cooling Flow Film Cooler Air Supply Film Cooler Pressure Melter Vacuum-Magnehelic EVS AP

Film Cooler AP HEME AP

RSM, July 2003

DESCRIPTION

200

0

02

2.6

2027

eg

0220

ξ õ 2

0 0 Ø Ø อั Ø

20

د ک

Outled

A or M Chg/Pu Chg/Pu

ž

ž ž

2

25

2 9

22

202

1-15

8

ş

220

22

200 - 250

ပ ပ

< 40 < 50

E C

1 - 1.5

29

3 22

> 3 S

345

õ

0

2F

34.1

50

đ

Z A

N/A

2

100 췅 ñ

C / NA

° ° % psi Υ Σ Š

ပ

3

TP-RPP-WTP-252

Sheet 8 of 1

Date: 7/6/63

80

6210

0600

040 8

94.12

<83 Net

v1.0

A or M

<300 Net

~

ta a

4,65 0150

× 10

off

On/Off

Ř

Overflow Temp Discharge Can Temp Glass Scale Alarm Condition

time hh:mm

ŝ

ပ

Reviewed and Approved: \mathcal{K}

WTPSP TSS# 91

2015 2000 C/ er 2302. 235 0.6 202 2 2222 10 с Г 212 বহাই ي م 2 36 M Q 0 Ś 2307 0,6 3.4 222 3,73 010 11 26 916 612 μ 1/3/1 Ž 0 0 õ 2:08 2208 028 713,403 و. 0 0022 Ĵ õ 2222 ч 0 30 2.5 35 312 103 0 ŝ 2 £ 0 0 С 2102 7/3/ NZ NC. 222 с, г ۵ N 3,0 0 ŝ 200 Ň 7.8 كما 0 53 36 0 34 210 ٤ 0 5 M വ Clear ZOOS N.C. ANG. 74.57 و 2 870 15/5 15/5/ 5.1 و ک 522 001 ò ž 0 M 13 8 <u>٥</u> 9.0 õ z 0 0 ව 0 1807 1406 76.14 0,6 3, 10 3.0 Clerk 2.9 5 2 at a 3 2 5 4 3 Data Sheet #2: Routine Statug ž d Ł ō 77.25 2100 0.6 112 112 112 光。 310 15/2/31 3 0 218 2981 y A 20 23 2.9 10 Ч Ф 20 44. z J Ň JW 0.0 h õ Q, A DIELE LENH F د 2 n z 124 0.27 100 9 e 362 2.0 601 3 3 6 53 224 2 ¢ Wo 2 h ዮ 6 range .5 - 1(reg FNT) 1-4 (reg OGT) 20-50 0 0.1 - 2.0 1-5 (reg SLT) 60-220 50 - 55 1100 - 1150 200 - 250 750 - 850 <300 Net <83 Net 1 - 1.5 < 10 v.0 < 40 < 50 <u>5</u>58 3 븅 By: Date: Time: A or M Kg On/Off psig in. H2O in. H2O in. H2O in. H2O in. H2O V or M Chg/Pull Chg/Pull Chg/Pull % C / NA hh:mm gpm scfm gpm psi Eg. ပ ပ ς β S S <u> 외</u> 이 gpm °. units Ha
 DESCRIPTION
 Iu

 Feed Nozzle Cooling Flow
 Film Cooler Air Supply

 Film Cooler Air Supply
 Film Cooler Air Supply

 Film Cooler Air Supply
 Ir

 Melter Vacuum-Magnehelic
 Ir

 EVS AP
 Ir

 Film Cooler AP
 Ir

 For Scrub Tank Volume
 Ir

 EVS Scrub Scrub Flow
 Ir

 For Scrub Flow
 Ir

 OG Control Valve position
 Ir

 OG Control Valve position
 Ir

 OG Control Valve position
 Ir

 Day Thk Agitator Setting
 Ining (DayTink) C

 Day Thk Agitator Setting
 Blower Cooling Flow

 Peed Nozzle Tamp (FNT)
 Melter Off-Gas Jumper Temp

 Post EVS Off-Gas Jumper Temp
 Post EVS ITemp

 Post EVS Off-Gas Jumper Tem time Feed Pump Setting Feed Pump Control Mode Feed Pressure Day Tank Wt Master Feed Tank Wt Overflow Temp Discharge Can Temp Alarm Condition RSM, July 2003

TP-RPP-WTP-252

Reviewed and Approved: 0-4

Sheet 2 of 13

Date: 3/1/63

WTPSP TSS# 91

Data Sheet #2: Routine Status Sheet

		Data She	eet #2: R	outine Sta	atus Spee		4	1	¢	¢
		Bv:	al a	20	600	54	EV -	101	Ż	
RSM July 2003	4	Date:	6/1/05	81	1/9/	8/1	6/1	1 911	1/00	811
		Time:	2010	2020	0307	00100	0528	2020	0%(0	0202
DESCRIPTION	units	range								
Feed Nozzle Cooling Flow	gpm	.5 - 1(reg FNT)	0.57	0.6	0,56	0.56	0.54	6.55	2.55	550
Film Cooler Air Supply	scfm	1-4 (reg OGT)	3	1.8	2.8	2.8	5.6	3.4	2	24
Film Cooler Pressure	psig	20-50	2	2	к	N	2	1.7.	3.5	S,
Melter Vacuum-Magnehelic	in. H2O	0.1 - 2.0	3.5	3.6	3.1	3.2	5.0	207	2	215
EVSAP	in. H2O		0	Ŷ	0	0	0	0	0	3
Film Cooler AP	in. H2O		0	0	0	0	0	0	ა	0
HEME AP	in. H2O		2.0	0.2	0,7	2.0	0.2	0.2	م. ه	2.0
Svetem AP	in. H2O		0	0	٥	0	0	0 41	٥	0
EVS HX Cooling Flow	map	1-5 (reg SLT)	2.9	3.0	2.9	2.9	5.3	4 \$ tot	2.9	2.2
FVS Scrub Tank Volume	liters	60-220	105	105	105	105	ЮG	106	106	901
EVS Nozzle Pressure	psi	50 - 55	541.5	54.5	54.9	SS	55	3	55	52
EVS Nozzle Flow	mdp		1.6	2.2	2.6	5.0	2.5	5.2	216	216
EVS Scrub Solution pH	Ha		1	6)	١	[1	1	,
OG Control Valve position	%		0	0	0	Q	0	θ	9	d
OG Control Valve mode	A or M		Z	14		2	pr	Ł	M	Z
Feed Pitmn (trihing) Condition	Cha/Pull		Rull	06		OKA	06	ok	٥	
Recirc Pump Tubing (Master)	Chq/Pull	85	10	040		oft	off	た	off	250
Recirc Pump Tubing (DavTnk)	Cha/Pull		05	04.		OL	06	OK	04	
Dav Trk Agitator Setting	%		2/1	26	UN V	NC	a.50	kc	24	N.C.
Master Tank Agitator Setting	· %		00	W C	MC	NC	20	ЪL	90	j. Z
Blower Cooling Flow	dpm	1 - 1.5	1,2	1.2	9 1	4.4	1.6	9''	٩٣	
Feed Nozzle Temp (FNT)	ပိ	< 40	50	62.	29	29	λŶ	27 27	27	28
Melter Off-Gas Jumper Temp	с °	200 - 250	222	t22	220	211	205	206	202	20
Post EVS Off-Gas Temp	ပ	< 50	34	32	33	33	33	33	32	22
Post HEME Off-Gas Temp	ပိ	100	34	32	32	32	3/	\$	31	2
Hq-Bed Temperature (or NA)	°C/NA	100	2/4	2/4	N/A	2/4	2/4	4/4	-	1
Scrub Liquid Temp (SLT)	ပ	140	35	32.5	33	32.5	32.5	32.5	31	32
EVS HX Cooling Temp	ပ	₹30	25	32	33	33	35	32	32	25
Feed Pump Setting	%	A THE REAL PROPERTY AND	9.0	9.3	35	9.5	9.5	95	9.5	s,
Feed Pump Control Mode	A or M		4	×	4	A	4	۲	A	Ŧ
Feed Pressure	psi	<1.0	0.9	1.7	1,5	1.8	- 8	1,5	51	0
Day Tank Wt	Кg	<83 Net	PH. 10	73.3	71.6	10.1	100.7	98,2	96.65	1212
Master Feed Tank Wt	5 Z	<300 Net	202	202	aor	202	101	131	12	131
Overflow Temp	ပိ	1100 - 1150	1130	1130	1133	639	1122	1135	1128	131
Discharge Can Temp	° C	750 - 850	869	869	010	876	876	670	869	200
Glass Scale	Кg	< 10	4.44	4.97	51.19	5.84	5.66	21.9	25.2	ار م
Alarm Condition	On/Off	off	4	010	Ъ	04	OF	44	40	012
time	hh:mm		0107	0200	0312	0405	0534	0100	0818	0909

TP-RPP-WTP-252

Date: <u>3/6/6</u>3

Reviewed and Approved: 2/1

WTPSP TSS# 91

2,65 0,65 204 205 60 d no 36 臣 3 Q q 0 2 ð ۵ť 250 湝 2 3.1 0208 34 30 okt ø 85781843 6 1506 N¢. 129 04 or Be 29 0 Σ 8 34 0 2:1 0 ó 117 5734 1409 مره 30 34 3,5 80 ž n 0 200 9 6 Z 8 0 Į Q 6 No. 3 5 0,0 3.5 n h 2 5 Ř 21 34 100 z 町 Data Sheet #2. Routine Status Sheet 9392 92.29 70.78 1203 و، و 970 N N N N ہ م 37 3.0 37 5 3 í? q Z 0 ð 000 20 hog 124 222 3,4 ēy 1000 و۔ ن 0.7 33 20 75 194 Ч б 3 Ō 2 ٥ 0 0 Cerst-K 604 870 e Ö 3,0 2.5 0 M 5 0 95 220 3 6 2 106 3 3 34 J 0 5 range 8 .5 - 1(reg FNT) - 1-4 (reg OGT) 20-50 20-50 1100 - 1150 8 1-5 (reg SLT) <83 Net <300 Net 750 - 850 200 - 250 0.1-2.0 50 - 55 1 - 1.5 v1.0 < 10 60-220 < 40 3,<u>6,5,6,8</u>,8 ff By: Date: Time:
 System AP
 in. H2O

 EVS HX Cooling Flow
 in. H2O

 EVS Scrub Tark Volume
 gpm

 EVS Scrub Tark Volume
 gpm

 EVS Scrub Tark Volume
 gpm

 EVS Nozzle Pressure
 psi

 EVS Scrub Scrub Tark Volume
 ph

 EVS Scrub Tark Volume
 ph

 EVS Scrub Scrupe
 pH

 OG Control Valve position
 Aor M

 Recire Pump Tubing (Master)
 Aor M

 Recire Pump Tubing (DayTink)
 ChgPull

 Raster Tark Agitator Setting
 %

 Master Tark Agitator Setting
 %
 gpm scfm psig in. H2O in. H2O in. H2O in. H2O in. H2O A or M ° C / NA On/Off hh:mm ပ ပ ပ္ ပ ပ ပ <u>s</u> S ပ ပ 훱 units % 집 Provincy I county I c time Feed Nozzle Cooling Flow Film Cooler Air Supply Film Cooler Pressure Melter Vacuum-Magnehelic Teed Pump Setting Teed Pump Control Mode Feed Pressure RSM, July 2003 DESCRIPTION Day Tank Wt Master Feed Tank Wt Overflow Temp Discharge Can Temp Alarm Condition EVS ΔP Film Cooler ΔP HEME ΔP Glass Scale

TP-RPP-WTP-252

Reviewed and Approved: 📿

Sheet IL of 13

Date: 3/1/03

WTPSP TSS# 91

TP-RPP-WTP-252

		Data She	set #2: Ro	outine Sta	atus Shee	t	9	4	4	
		BV:	220	9 3	93	2	34	or.	14	X
RSM July 2003		Date:	8/1/8	8112	201118	816103	3/2	3/2	6/2	812
		Time:	2003	and	39.95	0000	0103	1010	0306	0400
DESCRIPTION	units	range								
Feed Nozzle Cooling Flow	gpm	.5 - 1(reg FNT)	و	90	٥	2.6	0. 0	0. 0	0.6	624
Film Cooler Air Supply	scfm	1-4 (reg OGT)	910	818	2 K	3.0	10 10	8.2	20	2
Film Cooler Pressure	psig	20-50	6	6	Ъ	2	2	N	5	3
Melter Vacuum-Magnehelic	in. H2O	0.1 - 2.0	3.6	3.6	3,6	ŝ	3.8	5.8	3.4	3.4
EVS AP	in. H2O		0	0	Q,	0	0	0	0	0
Fitm Cooler AP	in. H2O		0	0	٥	0	0	0	0	0
HEME AP	in. H2O		0,2	0.2	D:2	0.2	0,2	0.2	0.2	0.2
Svstem AP	in. H2O		0	0	0	0	0	0	0	0
EVS HX Cooling Flow	db	1-5 (reg SLT)	9.9	3,0	3.0	3.0	3.0	3.0	3.0	ы 0
EVS Scrub Tank Volume	liters	60-220	0	0	111	112	13	112	112	<u>()</u>
EVS Nozzle Pressure	psi	50 - 55	5 29	59	59	52	52.1	52.1	52.1	52.1
EVS Nozzle Flow	gpm		311	31	31	5.1	1.1	3.1	3./	3:/
EVS Scrub Solution pH	На		1		-	(١	١	١	(
OG Control Valve position	%		0	0	0	0	0	0	0	0
OG Control Valve mode	A or M		W	N	W	М	У	Х	Z	Z
Feed Pump (tubing) Condition	Chq/Pull		0 V	ol.	ğ	Pull	ok.	ola	Óľ.	96
Recirc Pump Tubing (Master)	Chg/Pull		0KP	OUN	of V	~L/	244	del	000	040
Recirc Pump Tubing (DayTnk)	Chg/Pull		0 K	DIG	ØK	06.	ok	06	ole	ok
Day Tnk Aditator Setting	%		NC	NC	NC	40	20	20	NG	NC.
Master Tank Agitator Setting	%		NC	NC	NC	001	NC-	50	20	5
Blower Cooling Flow	gpm	1 - 1.5	1,8	1.75	1,75	125	44	1 1 1	1.7	1.4
Feed Nozzle Temp (FNT)	ပိ	< 40	39	99	Гđ	29	29	62	29	28
Melter Off-Gas Jumper Temp	° C	200 - 250	311	306	201	761	149	203	102	194
Post EVS Off-Gas Temp	°C	< 50	36	34	5	72	35	20	35	35
Post HEME Off-Gas Temp	°c	100	30	30	35	3 4/	2	20	25	3/
Hg-Bed Temperature (or NA)	° C / NA	100	ATA	đ 2	₹ Z	44	2/1	NIF	214	N 14
Scrub Liquid Temp (SLT)	ာိ	40	64	۶	31	32	32	32	22	20
EVS HX Cooling Temp	ပ	\$30	33	39	53	35	37	35	2	3
Feed Pump Setting	%		0	0	ιo	12.	9	6	9.0	9.0
Feed Pump Control Mode	A or M		Å	đ	£	A	Æ	¥	A	ł
Feed Pressure	psi	<1.0	2	- 10	0.8	0,0	1.7	1.1	1.5	1.0
Day Tank Wt	Кg	<83 Net	611-	76.2	73.X	20.7	69.6	68.1	61.0	64.5
Master Feed Tank Wt	Кg	<300 Net	lag	139	139	129	124	130	621	130
Overflow Temp	ပ	1100 - 1150	1131	1134	1135	1,72	1131	1128	900	400
Discharge Can Temp	ပ	750 - 850	870	869	867	861	810	830	019	810
Glass Scale	Кg	< 10	356	4.06	ж Т	5.27	حر ، جرح	5.85	6.14	6.49
Alarm Condition	On/Off	off	OFF	040	oft	BN	oth	diff.	00	01
time	hh:mm	a provinsi in the second of th	1100	dille th	9330	0034	0102	0205	0311	OTOY

Sheet \underline{N} of $\underline{/3}$

Date: 6/6/63

Reviewed and Approved: $\mathcal{R}\mathcal{U}\mathcal{R}$

WTPSP TSS# 91

		Data Sh	eet #2; Rout	ine Status She	at			
		By:	1 mg					
RSM. July 2003		Date:	8/2/03					
		Time:	0505					
DESCRIPTION	units	range						
Feed Nozzle Cooling Flow	gpm	.5 - 1(reg FNT)	0.56					
Film Cooler Air Supply	scfm	1-4 (reg OGT)	3.5		-			
Film Cooler Pressure	psig	20-50	2		_	-		
Melter Vacuum-Magnehelic	in. H2O	0.1 - 2.0	3.4					
EVS AP	in. H2O		0					
Film Cooler ΔP	in. H2O		0					
HEME AP	in. H2O		0.2			-		
System ΔP	in. H2O		0		_			
EVS HX Cooling Flow	gpm	1-5 (reg SLT)	8.2					
EVS Scrub Tank Volume	liters	60-220	//3					
EVS Nozzle Pressure	psi	50 - 55	52.2					
EVS Nozzle Flow	mdg		3.2			-		
EVS Scrub Solution pH	F		[
OG Control Valve position	%		0			_		
OG Control Valve mode	A or M		M					
Feed Pump (tubing) Condition	Chg/Pull		06.			-		
Recirc Pump Tubing (Master)	Chg/Pull		240			_		
Recirc Pump Tubing (DayTnk)	Chg/Pull		ok.					
Day Tnk Agitator Setting	%		40	•*				
Master Tank Agitator Setting	%		100					
Blower Cooling Flow	gpm	1 - 1.5	1.7					
Feed Nozzle Temp (FNT)	ပိ	< 40	19	-	•			
Melter Off-Gas Jumper Temp	ပိ	200 - 250	195					
Post EVS Off-Gas Temp	ာ	< 50	32					
Post HEME Off-Gas Temp	ပိ	100	26					
Hg-Bed Temperature (or NA)	° C / NA	100	2/4	-		-		
Scrub Liquid Temp (SLT)	င္	1 40	32	-		-		
EVS HX Cooling Temp	° C	≤30	32					
Feed Pump Setting	%		9.0					
Feed Pump Control Mode	A or M	Hard Street Street Street	4					
Feed Pressure	psi	<1.0	1.0					
Dav Tank Wt	ę,	<83 Net	(e5.1			_		
Master Feed Tank Wt	БХ	<300 Net	130					
Overflow Temp	ပ	1100 - 1150	1128			-		
Discharge Can Temp	ပိ	750 - 850	869			-		
Glass Scale	Кg	< 10	6.52					
Alarm Condition	On/Off	off	00			_		
time	hh:mm		0509				-	

Sheet (J of 13

Date: 3/6/63

Reviewed and Approved: 26

TP-RPP-WTP-252

Data Sheet #3: Off-Gas Monitoring Data

2
ΥΩ.
9
۹.
F
2
1
<u>n</u>
Ω.
œ
റ്
F
•

WTPSP TSS# 91

TI-RPP-WTP-264

			NBC							
		By:	Sati	HDS	GRY-	Cett	C A	CAR	(V)	CKK4 gym
RSM, July 2003		Date:	7/28	2/23/03	7128/03	129/09	1/28/03	20/8KIK	7128/09	7/28/0%
		Time:	10:40	\$6:11	11.54	1960	1902	2003	2100	23,00
DESCRIPTION	units	Range								
Chromatograph										
8	kppm	5-7	5:5	L	7,6	7.7	5.2	1. 8	5.4	k. 4
	mdd	<1e4	Ø	1	1	}	٢	\	لعمريم	
ff-Gas Residence Time	s	0.8-1.2			¢					
Test										
	mge//			۲ ۰ ۰۶	6, 01	20.2	24.3	20.3	401-14	20.0
0	mdd		2	200	135	150	<u> የ</u>	46,3	98.01	16.6
X	mdd			200	157	(ف ح	1.00	HEO!	112.1	153.5
0	mqq	<3E4		4.9	2.4	2,1	1	1	· / ·	5
0	7.ppm			146.0	106.10	0.771	112.0	0.212	86X4 2	(. H.Y.)
<u>, 6</u>	mqq	<2E4		Q	0	0	•	١		٦
Analytical	000000000000000000000000000000000000000	1000000		000000000000000000000000000000000000000						
emental Ho Concentration	ma/m ³			ſ	08/ 70	0.07	١]		
the Concentration	mo/m ³				0.m1/2 9	1	١	1		
mole Source	M/E/H/C	<u>е</u>				2		١		
Mace Flow	200000000	00000000	000000000000000000000000000000000000000	WINDOW						
Tracer Chan-1	nol		0	0.7	0,1	<i>b</i> 10	0')	01		1
elter Bubbler #1. Chan-2	ma	0-0.5	29	0.129	0.235	455.0	122.0	40.234	8,252	0.253
elter Bubbler #1, Chan-3	mq	0-0.5	. 36	0.136	0,245	0.192	0.243	0.244	0.251	0.26
e/H ₂ Sample Flow	mq	0.1-1								
2 Sample Conditions		1000 IS								
c Gauge	БН.	0 7	0	0	41	45	1.7	1.9	2.0	20
ter Change?	Νλ		2	Ń	2	N	ک	Ņ	Z	\sim
ter Temperature	ပ့	>100	Į)	XX	XX	2×2	E A	hat	lig
es Gauge	bsid	<15	5.2	5.5	5.0	5.0	5.0	5.0	20	5.0
eirite OK?	Yes/Chg		X	· ×		ħ	¥	۴	3	7
Gas Gauge Pressures	100000000									
enum	"on	0.5-4.0	ب م'ب	1	2.2	3.1	216	οiα	y Ŝ	26
ost EVS	wc"		1	1		/	١	ţ	i	(
ost HEME	*C	e	1	(0.1	21/2-	61	118	107	115
ost Charcoal Trap	wc"		33	١	01	11 22	1.1	115	- 1 -	n L
OST HEME	WC ^H					V.				
ost Cntri Valve	wc"		51	(28	20	78	30	29	29
Cylinders Tank Pres										
.ec	psig	>200	200	6 86	6800	680	01/	670	670	-020
e GC	psig	>200	1250	2481	1800	1000	06115	1230	306	[P 2:0
e Tracer	psig	>200		١	١	1	λ	(

Reviewed and Approved: DU/ Sc. R

Date: <u> 7/6/6</u>3

Sheet 1 of 23

WTPSP TSS# 91

8 03:26 **1** 03(ş т. 1 2 4 109.0 0.10 9 29 ind d 20.2 1490 le. 49 بلج 02 3.0 0 0 9 10/24 262 N 0 0.63 0.55 5 0 620 0721 . 30 510 0 20, ź ç ٥ 103.8 2/29/03 2/24/0) 103.6 26.0 0.12 3 20.3 262 650 0611 253 2.2 1520 6.2 C d <u>.</u>.) ļ l 1401 Ś 6.2.8 132.2 4.2 0445 2,02 (212) 0.09 0.68 1.0 30 0 N V l 3 601 0 56.38 6.2 262 5 3.03 Ş Data Sheet #3: Off-Gas Monitoring Data 510 0313 20.3 20 T. 38 0.2 4 0.18 ~ 0 2 ٥ 0 410 112.8 (52120 20.3 1680 103.3 0.31 1.0 253 262 2.4 0 66 { *Exhaust of Meiter (M), EVS (E), HEME (H), or Carbon Bed (C) # Multiple tanks for the He Tracer can be switched by manual valving if need be. 1 0 0 0 4-21.4 2/22/03 6.92 4 Ó 201 0.5 0100 253 010 1 P 261 C 0000 0100 €.4 100.7 (ſ h 3812 U f 77 $\langle \gamma \rangle$ Q 1 0-0.5 0-0.5 Range <3E4 By: Date: Time: 5-7 <1e4 <2E4 0.1-1 >100 <15 >200 80 0.5-4.0 psig psig mag mag mg/m³ M/E/H/C mq mq ц mdd 6H" kppm lpm time hh:mm mg/m³ units bpm -mpg-2 100000 ۶ l v psig ۳° Elemental Hg Concentration He/H2 Sample Conditions Vac Gauge Filter Change? Filter Temperature Pres Gauge Dreinte OK? He Tracer, Chan-1 Metter Bubbler #1, Chan-2 Metter Bubbler #1, Chan-3 Dreirite OK? Off-Gas Gauge Pressures Total Hg Concentration Sample Source* MKS Mass Flow Gas Cylinders Tank Pres^{*} Off-Gas Residence Time DESCRIPTION RSM, July 2003 PS Analytical Ma M Post EVS Post HEME Post Charcoal Trap He/H₂ Sample Flow Gas Chromatograph Post Cntrl Valve He Tracer Ar GC He GC AmTest <u>şğgg</u> so₂ 운 ŕ õ

TP-RPP-WTP-252

Reviewed and Approved: OC 4 2 2

Date: 3/1/63

Sheet 2 of 23

B.40

WTPSP TSS# 91

prerune geographic control volve extra brain). Sheet 2 of 13 turber CLARID. 0.253 20.31 COH10 12.61 N N 2 6.9 3,2 1000 5812 57 9 5 Ì S. Ø Ś \hat{o}_{1} values agreen to have been Jown stream of truth wh 0.262 000 0.25 20.25 12 bé 670 0940 515 # the magness 5.9 2 5 2 è 0 679 607 realing 0.216 0 1.25% 0.411 2005 200 6.99 0 2 Kay Ö or Nearly 0.58.454 oll e Y 0,252 0061 2253 6.34 S 01-7 1,00 50 A they 5 0,262 あり C 5 5 ç V These 2 avia 0 0.262 19103 0,253 ×0% Data Sheet #3: Off-Gas Monitoring Data 0,296 1.174 1<u>2</u>01 0.04 0.04 20.21 320 5.6 (i) 127 ϡ 174 10 129/03 Date: 8/6/63 6-63 163 0.253 50 1 440 0.26 6 80 0 0.e 1.00 5 11:35 7 2.9 3.4 2 *Exhaust of Melter (M), EVS (E), HEME (H), or Carbon Bed (C) # Multiple tanks for the He Tracer can be switched by manual valving if need be. 0 3 20:01 18-4-17-6-53 04-15 CM 20.29 10.9 \$05 2.05 150 1.00 1-9 4.9 > ₹ ļ þ 0 l ŗ 1 1.7 29/03 K D S 0.5-4.04 3-03.2 * -<u>50:</u> 20.28 6-74 692.0 <u>د</u> 0.101 100.1 • 50 00-8.253 1480 Z Z 0 1 1 19 0 0-0.5 >200 >200 >100 Range 5-7 <1e4 <3E4 ₹0 <15 By: Date: Time: <2E4 0 1 0.8-1 s |0.8-1 psig psig psig mdd mdd ₽IJĶ kppm mdd - Clarke mg/m³ M/E/H/C ng ng psig units Hudd mqq mg/m ۲Ç md I time hh:mm ပ္ Reviewed and Approved: 24 22 42 Elemental Hg Concentration Total Hg Concentration Sample Source* MKS Mass Flow He Tracer, Chan-1 Metter Bubbler #1, Chan-2 Metter Bubbler #1, Chan-3 He/H2 Sample Conditions Dreirite OK? Off-Gas Gauge Pressures Plenum Off-Gas Residence Time Gas Cylinders Tank Pres RSM, July 2003 Post HEME Post Charcoal Trap Post HEME DESCRIPTION Gas Chromatograph He/H₂ Sample Flow Filter Temperature Post Cntri Valve Vac Gauge Filter Change? Pres Gauge **PS Analytical** Post EVS Tracel б AmTest လိုင် လိုင် õ . £ £ g £ õ ŕ

5/20/12

mater

TP-RPP-WTP-252

WTPSP TSS# 91

Data Sheet #3: Off-Gas Monitoring Data

			· · · ·	1	141	200 -	641	A. A	Dire Sa	540
	_	ò.		(Kry)	- Jay		2100/07	100	120	3/30
RSM, July 2003		Uate:	7/29/27	129/07	7124127	104012	1120102	170	0, 0, 0	1300
		Time:	20.00	21/00	22:00	22.57	000 6	0101	2772	0.00
DESCRIPTION	units	Range								
Gas Chromatograph										
He	kppm	5-7	49	6.45	6,13	6.36	6.19	6.36	6.34	0
, T	maa	<1e4	0	0	0	0	0	0	0	0
Off Cac Dacidonace Time		0 8-1 0	>			X				
		0.0	000000000000000000000000000000000000000	000000000000000000000000000000000000000						
Amiest			000000000000000000000000000000000000000	000000000000000000000000000000000000000	000000000000000000000000000000000000000			2 * 2 · /		1000
02	mqa%		20,34	20.33	20.34	20.30	20.23	40.04	42'02	10.20
ON	bpm	- the state of the	130	601	97.4	(lb)	109	6625	0//	44
NOX	maa	All Distances	135	11	9.80	(23	103	43.60	. 120	88
00	mod	<3E4	1.7	e)	0	つ しんき	0	0.649	0	0,2
	mac 72		101	6 293	0 264	0.246	0.23	@ 499	0.199	0.2
200		10EA	C.F.		· ~ · · ·	0	C	0	9	0
302	hidd	7454		<i>www.www.www</i>		000000000000000000000000000000000000000	000000000000000000000000000000000000000	000000000000000000000000000000000000000	200000000000000000000000000000000000000	000000000000000000000000000000000000000
PS Analytical										
Elemental Hg Concentration	mg/m ³)	(0		1		1	
Total Ha Concentration	mg/m ³		(١	4.6 0	ĺ	1	5	ι	(
Sample Source*	M/E/H/C	正正素的 的		1	(ع ۲	/	(Landara .	l	ĥ
MKS Mass Flow		0000000								
He Tracer Chan-1	uni Mul	-			_		1.0	1,0	1.0	×.0
Moltor Dishiftor #1 Chan 2		0-0 5	5	2202	55	.252	2.5.5	252	253	253
			10/01	200	142	510	717.	6.76	262	292
Melter Bubbter #1, Cnan-3	E		102 -	- 900	1000	1000	660	2	(
He/H ₂ Sample Flow	mq	0.1-1)	ĺ	\	1				000000000000000000000000000000000000000
He/H2 Sample Conditions										
Vac Gauge	۶H	Å 8	1.7	15	1,9	6.6	2.0	4,0	410	2-
Filter Change?	N/λ	No. of Concession, No. of Conces	$\tilde{\mathbf{x}}$	Л	Ŋ	ž	2	N.	N	2
Filter Temperature	ပ္	>100	hind-	had	her	her	hor	(107	100/	1401
Pres Gauge	psig	<15	5,3	5.	51	5-1	5.0	5	2	2,
Dreirite OK?	Yes/Chg		2	>	7	Y	¥	759	Y	Y
Off-Gas Gauge Pressures										
Plenum	wc"	0.5-4.0	301	30	3,0	2.0	3.3	212	5.5	Z.7
Post EVS	wc"		49202	0	0	0	0.0	, 5, D	6.0	0.0
Post HEME	wc"	STATES IN	A. 5-	1.5	15	1.5	1.6	115	1.4	<u>, , , , , , , , , , , , , , , , , , , </u>
Post Charcoal Trap	wc"		1.4	12	1.5	115	6.1	1:5	1:3	6.7
Post HEME	wc"	1. A. A. A. A. A.	. (1	١		[¢	(\
Post Cntrl Valve	wc"		0%	30	29	250	30	50	30	20
Gas Cylinders Tank Pres [#]										
Ar GC	psig	>200	099	660	hell	650	630	650	629	020
He GC	psig	>200	060	1150	1140	1120	1100	1100	(100	1040
He Tracer	psig	>200	((٢		,	1
time	e hh:mm	Ref. Room	20:00	2109	22 0 k	123955	5000	0116	02.25	0304
*Exhaust of Melter (M). EVS	3 (E), HEN	IE (H). C	or Carbon Bec	1 (C)					•	
# Multiple tanks for the He Tr	racer can	be swit	ched by manu	ual valving if n	need be.					
It Minishin mining in the second				,						

TP-RPP-WTP-252

Reviewed and Approved: RUSs 42

Date: 3/1/63

WTPSP TSS# 91

Data Sheet #3: Off-Gas Monitoring Data

RSM, July 2003			24	11	m	40		- C C C		
Kow, July 2003	-		2 142 1- 2	5/00/2	2/01/2	2/20/03	7/00/63	Enlos1 7.	1/ 40/03.	- UV./L
	<u> I</u> i	Uale 1	1/2/02	1150/00	Calacit	110010	C O C V	NI 91	8118	UUT !!
		Ime:	0401	0505	0608	0510	Section of the sectio		odeločičko	<u> </u>
DESCRIPTION	units	Range								
as Chromatograph										
He	kppm	5-7	0	0	0	0	F-16.7	6.7	Le i l	<u><u> </u></u>
H	maa	<1e4	0	0	0	0	0	Ó	٥	0
Off Gas Desidence Time		7 8-1 2	1	((
mTast										
1001	d'anne	COCOCOCOCOCOCOCOCOCOCOCOCOCOCOCOCOCOCO	7 2 2 2	1000	1000	20.12)	20.24	20.2	19.9
02	LLMN0/	Automatica Contraction	00.00	01 01	10.01	10.00	/	224	10.4	961
NO	mdd		40.01	22.07	14.00	10 22	•	22.0	10 1	141
NOX	mdd		94.62	49.15	20.43	11.22		60	1.0	
co	mdd	3E4	٥	0	9.9	9.4	\ \	4.6	1.1	2
CO ₂	mad o		0.18	0.16	0.18	0.23)	6.17	0.164	5.22
SO,	maa	<2E4	0	0.85	24.2	2.63)	1.99	0.17	0
S Analvtical										
Elemental Ha Concentration	mo/m ³			ſ	l	(~100 20			l
	E-1	1				١	2 2			١
1 otal Hg Concentration		and a first of the second s			(1	3			1
Sample Source	M/E/H/O	1				000000000000000000000000000000000000000	200000000000000000000000000000000000000	000000000000000000000000000000000000000	000000000000000000000000000000000000000	
KS Mass Flow							, ,		00000000000000000000000000000000000000	
He Tracer, Chan-1	md	-	0.1	1.0	0',	1.0	0'10	1.00	100	
Melter Bubbler #1, Chan-2	mq	0-0.5	1253	, 253	223	.253	263	2.23	644	2020
Melter Bubbler #1, Chan-3	mq	0-0.5	292.	202.	.202	.262	262	242	292	1202
He/H ₂ Sample Flow	mq	0.1-1		•	(101	ţ)	
A/H2 Sample Conditions										
Vac Gauge	бН"	Ş N	2.5	2.8	3.0	3.2	3.2	E-1-2-12AH		
Filter Change?	ΥN		6	2	NO	20	٩٧	N N	N	2
Filter Temperature	ပ့	~1 00	1405	Het	401	thet	Hel		Nº L	2
Pres Gauge	psig	<15	'n	- -	4:5	4.2	4.2	2.2	, 'r	<u>م</u> ر م
Dreirite OK?	Yes/Chg	確認施設	405	Ye5	405	Ye)			مكممومه	
ff-Gas Gauge Pressures										
Plenum	"MC"	0.54.0	5.0	3.0	3.0	2.8	4-53	~	 	~~~~
Post EVS	wc"		0.2	2.0	2:0	9	0	6.3	>	
Post HEME	wc"		2.1	2.7	2.1	.e 0			-	
Post Charcoal Trap	wc"		1.4	51	, · ·	0.)		0 .1		21
Post HEME	wc"	$\mathcal{A}(\mathcal{A},\mathcal{A}_{0},\mathcal{A}_{0},\mathcal{A}_{0},\mathcal{A}_{0},\mathcal{A}_{0})$	}	{))	1	1	24	
Post Cntrl Valve	wc"		30	30	30	30	1.30	30	30	20
as Cylinders Tank Pres*										
Ar GC	psig	>200	650	650	650	450	623	1	1	650
He GC	psig	>200	0601	1080	1080	1080	1050	1030	1010	D1 800
He Tracer	psig	>200)	5	1	١			١.	
time	hh:mm		0400	0511	2/00	070	68:30			D:8
The second secon			Carbon Bo	10/17					•	

Sheet Sof 12

Date: 7/1/63

Ge da

Reviewed and Approved: 22 L

TP-RPP-WTP-252

			Data	Sheet #3: C	off-Gas Monit	toring Data		1 - 11	1	- KHI	
		By:	Calkan	CART	000	CAP.	CAR Delot	18 F	3/1/2	10/12/2/	
RSM, July 2003		Time:	2000), 2000),	12000	20.00	20100	23.60	×300	0019	101 by	
DESCRIPTION	units	Range									
Gas Chromatograph								1 2 0 D	. 84		
He	Kppm		7-1-10	6	12.27	24	6	0		0	
112 Off Car Davidance Time	ind ,	10010	AD [3	10%	1 6.00	2	2	~		,	
OII-Gas Residence Line		0.0-1-2	000000000000000000000000000000000000000	000000000000000000000000000000000000000	000000000000000000000000000000000000000	000000000000000000000000000000000000000			000000000000000000000000000000000000000		
Antest				. 6 0 2 2	1000	10 00	10 00	19 26	10 01	19,45	
02	sudd %		14. 11	14, 87	19.87	1.84	17187	11 15 0 0	12.20	-00.00	107 62 MS
ON	mdd		701	10%	22	10/	121	011	10211	10. 213	freedows to have
NOX	mdd	ļ	14	(L)	14	20 1	121		11201		
co	mqq	<3E4	ß	0.1	0	8 - 0	<i>.</i>	5	0,0		suited
cO ₂	uudel 🎾		0.511	0.611	0,561	0.566	0.465	0. 78 /	0.312	0100	
SO ₂	mqq	<2E4	в	0	0	0	0000	1.58	1.28	1177	
PS Analytical											
Elemental Hq Concentration	n mg/m ³		\$	Ĺ	ι		۱	١	(
Total Hg Concentration	mg/m ³	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1))	1	Ļ	anti	ſ	(l	1	
Sample Source*	M/E/H/C)	١))	١	l	(1	
MKS Mass Flow											
He Tracer, Chan-1	hm	+	1	~	1.00	600	1.00	1,00	1,00	1,00	
Melter Bubbler #1, Chan-2	lpm	0-0.5	,252-	. 252	252	1252	.252	e 252	2.5.2	1353	
Melter Bubbler #1, Chan-3	lpm	0-0.5	1 193	. 184	, (ga	2921	0261	261	.261	- 10L-	
He/H ₂ Sample Flow	md I	0.1-1		(-	\ 	(١			
He/H2 Sample Conditions											
Vac Gauge	۶H	¶ ₹	1.1	15	111	1.8	67	Ъ,́О	0.2	213	
Filter Change?	ΝX		2	2	1	2	2	1	ala	N I	
Filter Temperature	ပ့	×100	2.2	101	1.2	18%	4 al	hit.	141	+3	
Pres Gauge	psig	<15	5.0	570	5.0	50	50	5.0	5.0	5.0	
Dreirite OK?	Yes/Chg	学会建立性	r	A	Y	ý	í,	4/	Ye5	>	
Off-Gas Gauge Pressures											
Plenum	wc"	0.5-4.0	3.5	31	303	3/3	303	ν,Υ	2.2.2.3.0	212	
Post EVS	wc"	Succession of the	1	١	1	1	1	}	0.4	P.0.	
Post HEME	wc"		319	1.4	1.4	1,4	1.1	1.4	1.1	14	
Post Charcoal Trap	wc"		1,5	1.2	ls	Status Status	1.5	115	1.3	X.	
Post HEME	wc"	建建合业组织)	1	1)	1	١	\		
Post Cntrl Valve	wc"		29	28	29	36	20	30	30	50	
Gas Cylinders Tank Pres [#]											
Ar GC	psig	>200	650	650	1670	660	670	1.1.0	670	6 70	
He GC	psig	>200	050	740	220	000	064	670	650	640	
He Tracer	psig	>200	2	~	$\langle \rangle$	>	1)			
tim	e hh:mm	ALC MART	19110	90:23	ac:07	21805	d are	3706	0025	022	
*Exhaust of Melter (M), EV: # Muitinle tanks for the He	S (E), HEM Fracer can	IE (H), o he switc	r Carbon Be	d (C) ual valvino if r	heed be.				•		

ТР-КРР-М

WTPSP TSS# 91

9:30 19.83 1361 01.0 トン 3.6 <u>1, êp</u> - 470-145 -100 00 3 7.1 ¢ C 146 :262 5:13 123 1,00 Helo 3 0 1 50 0 141 0.23 139.65 1.00 0905 126.83 ,202 19.83 0 R 650 00 3 2 n Ž 364 19.81 120.33 130.77 0.22 4.40 1.00 202. 260 S 610 .253 0 3.0 00 ł 7/31 65 0 8 Ŕ X-12-6 X Data Sheet #3: Off-Gas Monitoring Data 1.00 3.05 19.86 \bowtie .262 125.2 7 ر ۲ New Yes 610 0.6 A A A 0 407 4. S 0 9 14674 19.82 15).08 164.54 6.0 11.5 9.3 1.2 0,339 2,17 3 1.00 .262 402 2 15/6 6.50 019 7.7 6.54 -1 0 12 3 3 0HI Σ Ł ٩ 12153 108.60 103.29 1.00 0.25 2 6.99 19.81 620 0 2.2 Het 3/31 g 115.69 1.2 1.5 1262 19.83 7/31/03 0.27 .253 2.02 150 1404 1.2 620 2.4 0 1.0 6.9 ò 200 9.5 2.7 0 × 0-0.5 0-0.5 Range <2E4 0.5-4.05-7 <1e4 >100 >200 >200 >200 By: Date: Time: <3E4 v 15 ÷. s |0.8-1. 0.1-1 Å mqq mqq ₽¥.° bpm med o psig kppm M/E/H/C шdd mg/m mq I psig units Elemental Hg Concentration mg/m wc" psig Ň NC. psid "ov ş "v Sample Source* MKS Mass Flow He Tracer, Chan-1 Metter Bubbler #1, Chan-2 Metter Bubbler #1, Chan-3 Off-Gas Gauge Pressures He/H2 Sample Conditions Gas Cylinders Tank Pres⁴ Off-Gas Residence Time Total Hg Concentration RSM, July 2003 DESCRIPTION Gas Chromatograph He/H₂ Sample Flow Post Charcoal Trap Filter Temperature Post Cntrl Valve Filter Change? SO₂ PS Analytical Pres Gauge Post HEME Post HEME Dreirite OK Vac Gauge He Tracer Post EVS Plenum He GC AmTest ğ s S ° N N N 00 £ ŕ

TP-RPP-WTP-252

Sheet 7 of D

ÔG I

0005

0422

time hhrime $|\sqrt{2}/7|$ $|\sqrt{2}/5|$ $|\sqrt{2}/5|$ the hermal of Melter (M), EVS (E), HEME (H), or Carbon Bed (C) # Multiple tanks for the He Tracer can be switched by manual valving if need be.

Reviewed and Approved: N. L. Y. R.

Date: 7/6/63

ті-RPP-WTP-264 V

WTPSP TSS# 91

Data Sheet #3: Off-Gas Monitoring Data

		By:	H105	HDS	HDS	RW2	HIDS	(84		Sec.
RSM, July 2003		Date:	2191103	7/31/03	13403	7131 103	75403	2/21/03	ashut	2/31/22
		Time:	10:28	11:33	12:27	1305	14:21	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	7000	1900
DESCRIPTION	units	Range								
Gas Chromatograph										
He	kppm	5-7	6.99	6.99	6.90	7.0	N Reeds	× 7.5	7.47	202
H ₂	mqq	<1e4	0	Q	٥	0	٥	0	0	0
Off-Gas Residence Time	s	0.8-1.2								
AmTest										
02	%ppm		19.29	19.82	19,83	19.9	14,38	19.15	14.2	423
NO	bpm		148	136	103	131	133	OF	146	125
NOX	bm		158	145	112	171	140	1 28	151	138
co	bpm	<3E4	3.9	0	6	2.6	, 1, 7	ی غ	A.Z.	2
co ₂	20 ppm		0.218	0.218	0.205	0.285	202.0	0.499	0.429	0.379
SO ₂	mqq	<2E4	3.4	236	1.0	ġ	Ó	\ Q	ġ	в
PS Analytical										
Elemental Hg Concentration	mg/m ³		. b blim	7.0		01	1		ι	1
Total Hq Concentration	mg/m ³	ALC: NO	10.1. 6.	8.3		Я	,	40 m m	l	
Sample Source*	M/E/H/C		N N	ដ		¥	,	\	/	J
MKS Mass Flow										
He Tracer, Chan-1	hm	1	1,00	1,00	001	(.00	1,00	1,00	1.00	00
Melter Bubbler #1, Chan-2	nd	0-0.5	0.253	0.253	01253	253	0.253	230.0	0,257	0,252
Melter Bubbler #1, Chan-3	hpm	0-0.5	o. rich	E. 26%	0.2 W	262	0.262	0.262	0.790	8,362
He/H ₂ Sample Flow	d b	0.1-1	•		2				Jar. Alla	
He/H2 Sample Conditions										
Vac Gauge	۳Hg	0₽	i. F	+	1. T	1.4	ŗ	1.4	7.7	144
Filter Change?	٨N		Ø	Ņ	Ş	V	2	0	N/	1 m
Filter Temperature	ပ့	×100				14eT	(int	her	100	<u>h</u> M
Pres Gauge	psig	<15	5.3	5.2	5.5	5.5	5,5	515	52	5.2
Dreirite OK?	Yes/Chg		У	-/-	×	λ	λ	ر میں میں میں میں میں میں میں میں میں میں	<u>, 1</u>	×~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
Off-Gas Gauge Pressures										
Plenum	"ON	0.54.0	۲.e		7.5	1.7	5.0	5,0	270	5°8
Post EVS	"o		1.4	70	э	Q.	5.	3	200	20
Post HEME	"o		(19	5 1	114	1.9	5-	2,0	212	2,2
Post Charcoal Trap	"C"		۽ ر -		/.	(N N	11	1	0
Post HEME	wc"	Set A. routent.		0,7	5,0	dr	202	2.2	210	50
Post Cntrl Valve	wc"		20000000000000000000000000000000000000		bootototototo	3 0		20.000		
Gas Cylinders Tank Pres"					(~~~~~	007	UUUUUUUU	1 d M
Ar GC	bsid	>200	1	1		130	A at M	0.00	6.29	1000 C
He GC	bsid	>200	340	340	025	200	へりのた	1920	ADIU	7000
He Tracer	psig	>200					11.116			10 611
time	hh:mm	60 10 000	10:39	11:38		13:20		1604	1787	10 01
*Exhaust of Melter (M), EVS	(E), HEM	E (H), o	r Carbon Bec	(C)					•	
# Multiple tanks for the He Tr	racer can	be switc	thed by man	ual valving if n	eed be.					

ТР-КРР-WTР-252

Date: 8/6/03

Sheet 2 of 13

WTPSP TSS# 91

126 09/1 0.23 10,2 : 25 126 0216 777 0 Y 516 0.6 1,7 221 9.6 3 30 7 ź Q 202 19.20 20-1 2112 80 7.24 0.23 1.83 22 1800 11-3 124.2 N 0.3 0 Ş 19.19 0.279 17.91 140.64 000 7.20 1,00 1820 255 202. 201 1:42 1.0 202 200000 5.5 ţ C d' 3 701 19.22 0.275 23:05 Carlet-0,60 192. i. 00 2.5 989 5.0K 2 Eg 30 14 2068 5 0 ĸ あっ 221.10 (AH 7/107 29.9 0,281 169 40 19.27 1920 **Off-Gas Monitoring Data** 0.0 30 0.59 109 25. 9 Q, 8-2 99 0 Ł 6.88 CX 1 0.342 19.2 2001 198 1 1252 8 180 XI01 time hh.mm $|A| = 120^{-1} |A| = 20^{-1} |A| = 20^{-1} |A| = 20^{-1}$ •Exhaust of Meiter (M), EVS (E), HEME (H), or Carbon Bed (C) # Multiple tanks for the He Tracer can be switched by manual valving if need be. 45 1 5 31/03 0.350 19,26 200 Data Sheet #3: 74 eses 1950 11 50 2 NA W 7 \mathcal{O} ため 6.98 6 j/ 3.4 096 0 1 252 1922 0.330 1571 2.0 8 Ň 12 1980 CR H 06 ſ 2.7 68 0 0 0-0.5 .5-4.0 5-7 >200 >200 By: Date: Time: 1500 <1500</p> <1e4 <2E4 Range <3E4 0.1-1 Å >200 IM/E/H/C 0.8-1 psig psig kppm mqq шa Ipm "Hg Y/N °C Psig units mdd mqq mg/m³ mg/m³ шd hm mdd "o bpm psig ۳ ۲ "N ۳ ۲ ۳ ۲ WC. Elemental Hg Concentration Melter Bubbler #1, Chan-2 Melter Bubbler #1, Chan-3 e/H2 Sample Conditions Dreirite OK? Off-Gas Gauge Pressures Gas Cylinders Tank Pres Off-Gas Residence Time AmTest Total Hg Concentration RSM, July 2003 DESCRIPTION He/H₂ Sample Flow Post Charcoal Trap Gas Chromatograph Filter Temperature He Tracer, Chan-1 Post Cntrl Valve Sample Source* MKS Mass Flow Vac Gauge Filter Change? Pres Gauge Post HEME PS Analytical Post HEME Post EVS He Trace Plenum He GO Ar GC ୶ୢୢୖ୰ୖୢୖୢ୰ୖୄ୰ୖ s02 ۳ ŕ

TP-RPP-WTP-252

Date: 3/6/63

Reviewed and Approved: ALL

Q.

Sheet $\underline{1}$ of $\underline{3}$

WTPSP TSS# 91

7.44 11:09 0,349 56 6.92 20.91 00 0 35 973 ч М 30 \$. . . r. 20 22 6.9 J 200 U -G 20 0.347 19.20 262 263 2520 3,44 00 0 7,05 21 215 2,2 ć Kr3 Ľ 0 â Юß 9.6° 6 60 1595 NA 2.93 0.369 012 1267 1060 HAN . 200000000 er A 19.79 5 he 200 1.00 3 212 C 5 à З 6.7 e 253 1267 6.97 0809 8 1600 0.339 34.6 19.91 10-10-٩ 20 ç 080 660 22 2.4 CX4 118 e 3.42 1262 Data Sheet #3: Off-Gas Monitoring Data 0205 ĝ 6020 1.00 7.06 1620 10 60 ĥ Ø. 0.29 19.83 198.42 202. 0543 B.3. Ń 21.3 1.00 7.48 650 1680 05.34 time [hh:mm] [action] [a Hel 3 30 3.20 0.20 1.00 94 2.8 .262 450 9.2 0 5.0 1 1910 N. 040 14. 126.47 60.2 22.0 500 8.0 222 10.4 7.48 00 24 8 Θ 1401 3 2 0.5-4.0 >100 ×200 ×200 0-0.5 0-0.5 Range By: Date: Time: <1e4 <3E4 <2E4 5-7 0.1-1 Å <15 <u>6</u> psig 탄^시ン psig mqq mqq kppm mdd tidd M/E/H/C md mq units mg/m3 mdq Elemental Hg Concentration mg/m³ ۶ ۲ wc" . ≥ psig psig He Tracer, Chan-1 Melter Bubbler #1, Chan-2 Melter Bubbler #1, Chan-3 fe/H2 Sample Conditions Dreirite OK? Off-Gas Gauge Pressures Total Hg Concentration Sample Source* MKS Mass Flow Gas Cylinders Tank Pres* Off-Gas Residence Time RSM, July 2003 DESCRIPTION He/H₂ Sample Flow Post Charcoal Trap Post HEME Gas Chromatograph Filter Temperature Post Cntrl Valve Fitter Change? Pres Gauge PS Analytical Post HEME Vac Gauge He Tracer Post EVS Plenum He GC Ar GC AmTest Ô ő so g 8 £ õ ŕ

.

TP-RPP-WTP-252

Date: <u>8/6/6</u>3

WTPSP TSS# 91

TI-RPP-WTP-264

RSM, July 2003 Description interviewer Description interviewer Description interviewer Description interviewer Description interviewer Milestimation interviewer Milestimatinterviewer Milestimation interviewer </th <th></th> <th></th> <th></th> <th>Data</th> <th>Sheet #3: U</th> <th>TT-Gas Monito</th> <th>ning uata</th> <th></th> <th>2414</th> <th></th> <th></th>				Data	Sheet #3: U	TT-Gas Monito	ning uata		2414			
Nam. July 2003 Team. July 2013 Team. July				Self and	CUANT INS	200	C& H 21, 10-30	10 2	01/05	11/02/11/8	81103	
DESCRIPTION units Rame F/I Part (2) Lot C/I C/I <thc i<="" th=""> C/I C/I</thc>	Kom, July 2003	-15	me:	20/1/2	120051	1462	1503	16:05	17:05	18:00	-CO:61	
ac Chronatograph (11) (11) (11) (11) (11) (11) (11) (11	DESCRIPTION	units F	ange									
He He Hom 57 The fifth b, 12 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 7.2 4.1 4.	Bas Chromatograph											
H- Inflext BPI (Fills) Cold D	He	kppm	5-7	101)-01	6:32	6.90	717	2	11.2	7.7	, ,	
Off-Gas Residence Time s 0.8-13 <td>H₂</td> <td>mqq</td> <td><1e4</td> <td>Q</td> <td>Ø</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>3</td> <td>0</td>	H ₂	mqq	<1e4	Q	Ø	0	0	0	0	3	0	
mintest montest fram frag	Off-Gas Residence Time	s 0	.8-1.2	. \$	l	١	١					
0, 100 内容m 1(1) 2(1)	AmTest											
NO	°0	wee %		(83)	i1,86	19.45	19.88	19.87	N.	19.86	19.42	
Kix	CN	maa	- THE PARTY	110	Ĩ	704	422	228	S.	222	59	
CO PPM SEA P C <thc<< td=""><td>NOX</td><td>maa</td><td></td><td>321</td><td>22.7</td><td>272</td><td>299</td><td>249</td><td>. ب</td><td>240</td><td>イト</td></thc<<>	NOX	maa		321	22.7	272	299	249	. ب	240	イト	
Co ₂ Kµm co 357 0, 4/1 0.357 0, 4/1 0.357 0, 4/1 0.4/2	CO	шaa	3E4	0	0	0	0	0	X	с	0	
SO ₂ Ppm <2E4 1/16 0.57 0 <th0< th=""> <th0< th=""> 0</th0<></th0<>	co,	Mee %		0.359	0.417	0.285	0.463	0.532	5	0.412	0.293	
S Analytical Mail To the second second and the second and the second sec	, SOS	maa	<2E4	1 76	0.59	0	0	0		0	۵	
Elementation Total Hg Concentration Total Hg Concentration Mg/m % f M kg/m M	oc Analutical	200000000000000000000000000000000000000										
Total Hg Concentration Mit/ Internation Mit/Internation Mit/Internation <th linternation<="" th=""> Mit/Internaternation</th>	Mit/Internaternation	- 5 Allalytical Flowertol Ho Concentration			9 6		14 11-P-2	4 7	R 16 H05	H.7.	12 1. 1 & A	
Instruction ME/HIC M Construction Mit Rise Source ME/HIC M Construction Mit Mit He Tracer Merine Merine Mit Mit Mit Mit He Tracer Merine Mit Mit Mit Mit Mit He Tracer Mit Mit Mit Mit Mit Mit Mit He Tracer Im 0.1 1 Mit		ma/m3		8 C		0 71	0 1	24	1. 7.	Dr W		
Sample Solue Multiruly Numericuly Numeri			Prifector (NS) antiactificities, factor	117		712	2			A K		
Micro Mass riow Micro Mass	Sample Source	M/E/H/C			000000000000000000000000000000000000000		sector (consector)	description				
He Iracer, Cnan-1 I. Wor	WKS Mass Flow				000000000000000000000000000000000000000		. 20		60	1 00	1,00	
Meller Bubbler #1, Chan-2 pm 0-0-3 3.52 1.242 0.242 <th0.24< th=""> 0.242 <th0.24< th=""></th0.24<></th0.24<>	He Tracer, Chan-1	<u></u>	- 0	1,000	1.60		150	1. 10	2000	0.553	0 063	
Meller Bubbler #1, Chan-3 Ipm 0-03 C2-02 C2-02 <thc2-02< th=""> C2-02 <thc2-02< <="" td=""><td>Melter Bubbler #1, Chan-2</td><td>ud.</td><td>0-0.2</td><td>-252-</td><td>2520</td><td>\$ 57 S</td><td>2520</td><td>10. 29 8</td><td>- C12 - C</td><td>2.14</td><td>5-7-7</td></thc2-02<></thc2-02<>	Melter Bubbler #1, Chan-2	ud.	0-0.2	-252-	2520	\$ 57 S	2520	10. 29 8	- C12 - C	2.14	5-7-7	
Herlty Sample Flow Ipm 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1-1 0.1	Melter Bubbler #1, Chan-3	щ	0-0.5	2921	12.62	1366	Take.	D. 144	C. 564	E1. 25 de	104-7	
Heirt Sample Conditions Ho LO LO <thl< thr=""> List List <th< td=""><td>He/H₂ Sample Flow</td><td>mq</td><td>0.1-1</td><td>١</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<></thl<>	He/H ₂ Sample Flow	mq	0.1-1	١								
Vac Gauge *Hg 40 10 140 10	He/H2 Sample Conditions											
Filter Change? YIN A N A N	Vac Gauge	6H"	9 10	1.0	3,1	1.0	0	>	140	4 1	1,4,1	
Filter Temperature °C >100 LAY LAY <thlay< th=""> LAY <thlay< th=""></thlay<></thlay<>	Filter Change?	Νλ		X	N	2	2	N	2	2	N.	
Pres Gauge psig <15 5.5 5.1 5.5 5.4 5.3 5.4 7.3 Derine OK? Ves/Chg V <td>Filter Temperature</td> <td>ပ့</td> <td>~100</td> <td>1.54</td> <td>ten</td> <td>15 y</td> <td>404</td> <td>2011</td> <td>بر ک</td> <td>464</td> <td>he^T</td>	Filter Temperature	ပ့	~100	1.54	ten	15 y	404	2011	بر ک	464	he ^T	
Dreinte OK? Yes/Chg V Y	Pres Gauge	psig	<15	5.5	5.1	sis	5.4	5.2	4,5	5. 2.	5	
Off-Gas Gauge Pressures Mcc 0.54.0 3.0 3.14 3.15 3.14 3.15 0.14.5 3.15 0.14.5 3.15 0.16.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 0.14.5 <th0.14.5< th=""> 0.14.5 <th0.14.5< td="" th<=""><td>Dreirite OK?</td><td>Yes/Chg</td><td></td><td>Ì</td><td>7</td><td>۲</td><td>></td><td>X</td><td></td><td></td><td>000000000000000000000000000000000000000</td></th0.14.5<></th0.14.5<>	Dreirite OK?	Yes/Chg		Ì	7	۲	>	X			000000000000000000000000000000000000000	
Plenum wc 054.0 3.0 3.14 3.14 4.13 2.15 5.15 Post HEME wc wc 5.15 3.5 3.5 1.1 1.0	Off-Gas Gauge Pressures						200000000000000000000000000000000000000				17 V	
Post EVS wc 2.14 2.55 3.65 1.7 1.0 1.16 1.0 1.16 1.0 1.16 1	Plenum	"O	0.5-4.0	0,6	714	310	hit	6.2	1.7	2 4 4		
Post HEME wc 2.4 2.5 3.5 1.6 1.0 1.	Post EVS	wc"		١	1	1	\ <u>_</u>		ہ د	5.9.0	2.2	
Post Charcoal Trap wc 3.5 3.5 3.6 1.7 1.0 1.4 Post HEME wc wc 3.5 3.6 3.7	Post HEME	wc"	- 191	7.4	25	8	18			م د ب	<u>ر</u> د -	
Post HEME wc wc 30 30 30 37 33	Post Charcoal Trap	wc"		2.5	215	2.5	210	L . 1	0.1	>	2,1	
Post Critir Value wc" 300 520 50 500 50 500 50 500 500 50 500 500 500 500 500 500 500 1300 1300 1410 <td>Post HEME</td> <td>wc"</td> <td></td> <td>\</td> <td>1</td> <td>J</td> <td>)</td> <td>1</td> <td>ر مر</td> <td>, in the second second</td> <td>ů v</td>	Post HEME	wc"		\	1	J)	1	ر مر	, in the second	ů v	
Case Cylinders Tank Pres ⁴ 200 6 70 6 6 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Post Cntrl Valve	wc"		30	30	30	30	25	74	2		
Ar GC Psig >200 670 680 680 680 1460 1400 1400 1400 1400 1400 1400 140	Gas Cylinders Tank Pres [#]											
He GC Psig >200 (オスレ 15.05 /5.06 (チョレ 141.0 141	Ar GC	psig	>200	670	h & O	040	1980 -	/	()) .	
He Tracer psig >200 ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	He GC	psig	>200	1320	15:05	1500	1500	207	1420	1410	1 280	
time hh:mm (1.207 (3.00 / 14.07 5.10 16.15 7.05 ***********************************	He Tracer	psig	>200	1	J	\ \	/					
•Exhaust of Metiter (M), EVS (E), HEME (H), or Carbon Bed (C)	time	hh:mm		Lovel	1306	1401	1510	16:15		80:02	19:30	
	Ume Ume af Matter (M) EV/S		E E	K Carhon Rec	1 1 2 1 6	1 17 0 1		1.1.011		· · · ·		
		ייייייייייי			r (c) referencier	ad ha						

ć ļ

B.49

Date: 2/1/62

Reviewed and Approved: \mathcal{K}

Sheet IL of 13

WTPSP TSS# 91

1180 1.00 262 0305 0.3 ð . ئ 7.3 E e 0 Ŀ Ĵ 5 N 253 1200 90 30 650 0 5 o, 1202 6 50 12.20 088 S1 81 3,5 0.3 1.30 ~ ١ 200 ŝ L L'G 1.2.1 7,0 ð 5,1 199 06 9.5 2,2 1260 <u>1,8;1</u> 0.440 61.253 0.262 50:00 14 ر ار 20 30.7 200000 0 er. 4 5 2 G - Sales 0.453 14.84 0.262 1.00 0,42 305 1280 Monitoring Data 63 63 さ ×, 200 ~ ej C 3 Θ 22:06 19.86 0.707 0.253 300 1,9 0.26 1,46 19 chte Ma-0.1 •Exhaust of Metter (M), EVS (E), HEME (H), or Carbon Bed (C) # Multiple tanks for the He Tracer can be switched by manual valving if need be. 461 hurt Ţ 5.3 Data Sheet #3: Off-Gas 121:09 10.3. 13.3 0.466 19.94 0.29 2 0.253 0.361 226 2.04 e 1330 0,0 ١ C 2 0 12 3.0 494.0 NP2-1-1.00 20:13 1.51 0.261 0 1340 $\langle \gamma \rangle$ 6.253 20 198 204 0 d'e d_ 1 29 à 0 ŝ, >200 >200 Range 0.5-4.0 0-0.5 0-0.5 By: Date: Time: 5-7 <1e4 <3E4 <2E4 0.1-1 100 s |0.8-1. >100 <15 ~200 psig psig mg/m³ M/E/H/C units kppm udd udd ppm рт шd шd nd "Hg Y/N °C psig ~med~ bpm 2-ppm psig time hh:mm mg/m³ 000000 Elemental Hg Concentration Melter Bubbler #1, Chan-3 He Tracer, Chan-1 Meiter Bubbler #1, Chan-2 Pres Gauge Dreirite OK? Off-Gas Gauge Pressures **He/H2 Sample Conditions** Gas Cylinders Tank Pres Off-Gas Residence Time Total Hg Concentration RSM, July 2003 DESCRIPTION Vac Gauge Filter Change? Filter Temperature He/H₂ Sample Flow Post Charcoal Trap Gas Chromatograph Post Cntrl Valve Sample Source* MKS Mass Flow SO₂ PS Analytical Post HEME Post HEME Post EVS He Trace Plenum He GC AmTest <u>م</u> <u>ဂီ ဍဍိ</u>ဒိုပို့ ŕ

TP-RPP-WTP-252

Date: 2/6/63

Reviewed and Approved: CLA

Ś

Sheet 12 of 23

WTPSP TSS# 91

TI-RPP-WTP-264

RSM. July 2003 Dir. My 2013 Dir. Dir. <thdir.< th=""> Dir. Dir.</thdir.<>	
RSM, July 2003 Their Phylor Phylor DESCRIPTION Their Phylor DESCRIPTION DESCRIPTION <thdescription< th=""> DESCRIPTION DESCRIPT</thdescription<>	
DESCRIPTION Time: $MOr.$ M	
DESCRIPTION units Range Print S/1 T/1 C/2 He ppm S/1 T/1 C/2 He ppm s/1 T/1 C/2 He ppm relet ppm relet ppm Or.Grass Residence Time s 0.813 Ois ppm Mail G.2 Ois ppm mail G.2 Ois ppm relet L/2 Ois ppm relet L/2 Ois ppm relet L/2 Soc. ppm relet L/2 Soc. ppm relet L/2 L/2 Soc. ppm relet L/2 L/2 L/2 Soc. som relet L/2 L/2 <th></th>	
ability Chrometograph Sector Dim $2/3$ $1/1$ $2/3$ $2/1$ $2/3$	
d_{das} kpm 5.7 7.11 6.72 2.7 2.11 6.72 2.7 2.11 6.72 2.7 2.11 6.72 2.7 2.11 6.72 2.7 2.11 2.7 2.11 2.7 2.11 2.7 <	
η_1 η_1 η_2 η_1 η_1 η_1 η_2 η_2 η_1 η_1 η_2 η_1 η_1 η_2 η_1 η_1 η_2 <t< td=""><td></td></t<>	
Off-Gas Festlence Time s $[0,1,3]$ $$ $$ NO ppm $gr/2$ </td <td></td>	
Treat Prom Prove Prove <th< td=""><td></td></th<>	
0_{1} pm p_{1} p_{2} p_{1} p	
NGDpmdpm $\psi_1 \chi_1^2$ $\psi_1 \chi_2^2$ $\psi_1 \chi_2^2$ NOxppmmesc $\psi_1 \chi_2^2$ $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ NOxppmsets $\psi_2 \chi_3^2$ $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ C0ppmsets $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ S0S0ppmsets $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ S0S0ppmsets $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ S0S0ppmmesc $\psi_1 \chi_3^2$ $\psi_1 \chi_3^2$ S0S0ppm $\psi_1 \chi_3^2$ $\psi_2 \chi_3^2$ $\psi_1 \chi_3^2$ S0S0sets $\psi_1 \chi_3^2$ $\psi_2 \chi_3^2$ $\psi_2 \chi_3^2$ S0S0 $\chi_2 \chi_3^2$ $\chi_2 \chi_3^2$ $\chi_2 \chi_3^2$ $\psi_2 \chi_3^2$ Meler Bubbler #1, Chan-2pm 0.11 $$ $\psi_1 \chi_3^2$ Meler Bubbler #1, Chan-2pm 0.11 $$ $\psi_2 \chi_3^2$ Meler Bubbler #1, Chan-2pm 0.11 $$ $\psi_2 \chi_3^2$ Meler Bubbler #1, Chan-2pm 0.11 $$ ψ_2^2 Meler Bubbler #1, Ch	
NOXppmppm $\mathcal{M}_{11}\mathcal{A}_{11}^{\prime}$ $\mathcal{M}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{M}_{12}\mathcal{A}_{11}^{\prime}$ O_{2} ppm $CE4$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ SO_{2} ppm $CE4$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ SO_{2} ppm $CE4$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}^{\prime}$ $\mathcal{L}_{12}^{\prime}$ SO_{2} ppm $CE4$ $\mathcal{L}_{12}\mathcal{A}_{11}^{\prime}$ $\mathcal{L}_{12}^{\prime}$ $\mathcal{L}_{12}^{\prime}$ $Simple Source*mm/m*Simple Source*mm/m*$	
00 ppm $SE4$ $4,2$ $2,7$	
S02 ppm $< 2E4$ $L_3 \mathcal{J}$ $\mathcal{J}_1 \mathcal{J}_2$ $\mathcal{J}_1 \mathcal{J}_2$ Analytical month $2E4$ $L_3 \mathcal{J}_1$ $\mathcal{J}_1 \mathcal{J}_2$ $\mathcal{J}_1 \mathcal{J}_2$ Calementation month $2E1$ $L_3 \mathcal{J}_1$ $\mathcal{J}_1 \mathcal{J}_2$ $\mathcal{J}_1 \mathcal{J}_2$ Calementation month 1 $L_1 \mathcal{J}_2$ $L_1 \mathcal{J}_2$ $L_1 \mathcal{J}_2$ Sample Source ME/H/G $$ $$ $$ $$ $$ Gample ratio 1	
Analytical Analyt	
Elemental Hg Concentration mg/m ³	
Total Hg Concentration mg/m ³	
Sample Source* MEHIC	
CS Mass Flow Meller Bubbler #1, Chan-2 pm 1 L_0D_2	
Image: Chan-1	
Weiter Bubbler #1, Chan-2 Ipm 0-0.5 $Z\delta/2$ <	
Weiter Bubbler #1, Chan-3 Ipm 0-0.5 Z_0Z Z_0Z Z_0Z Z_0Z Helf Sample Flow Ipm 01-1 \neg \neg \neg \neg Helf Sample Flow The \neg \neg \neg \neg \neg Helf Sample Flow The \neg \neg \neg \neg \neg Helf Sample Flow The \neg \neg \neg \neg \neg Helf Change? YiN \neg \neg \neg \neg \neg Piller Temperature \neg \neg \neg \neg \neg \neg Press Gauge Psig <15 5 7 5 7 Press Gauge ViN \sim γ γ γ γ Press Gauge vc^{\prime} 0.54 3.2 3.7 γ γ Post Claun wc^{\prime} 0.54 3.7 2.7 3.7 σ Post EVS wc^{\prime} 0.7 0.7 0.7 σ σ σ	
Helly, Sample FlowIpm0.1-1 \neg H2 Sample Conditions M $0.1-1$ \neg H2 Sample Conditions M $0.1-1$ 1.5 H2 Sample Conditions γ M 1.5 Hac Gauge γ N M α couper γ 1.5 α couper γ 1.5 γ 1.5 3.4 5.4 γ 5.4 5.4 5.4 γ	
H2 Sample Conditions H3 Sample Conditions $(ac Gauge$ Hg $\exists 0$ L_{1}^{2} L_{2}^{2} L_{3}^{2} $(ac Gauge$ γ γ γ γ γ γ $(ac Gauge$ γ γ γ γ γ γ $(ac Gauge$ γ	
lac Gauge "Hg sto L_{f} L_{2} L_{2} L_{2} L_{2} Tiller Change? YN max L_{f} L_{f} L_{f} L_{f} Tiller Change? YN max L_{f} L_{f} L_{f} L_{f} Tiller Change? PN NN max L_{f} L_{f} L_{f} Tiller Change? PSig L_{f} L_{f} L_{f} L_{f} L_{f} Detrite OK? Vex/Change P_{f} L_{f} <	
Itler Change? YIN λb	
iller Temperature °C >100 $h_0 F_1$ $h_0 F_1$ $h_0 F_1$ $h_0 F_1$ Pres Gauge Disition OK3 $Y_2 S_1$ $Y_2 S_1$ $Y_2 S_1$ $Y_2 S_1$ Pres Gauge Disition OK3 $Y_2 S_1$ $Y_2 S_1$ $Y_2 S_1$ $Y_2 S_1$ Press Gauge Pressures wc^* $0.54.0$ 3.2 3.1 9.1 Plenum wc^* $0.54.0$ 3.2 3.1 9.1 0.1	
Tes Gauge psig <15 5,4	
Dietrite OK? Yes/Choi Yés Yes/Choi Yés/Choi	
I-Gas Gauge Pressures wc* 0.5-4.0 3. 2 3. / 1 Pienum wc* 0.5-4.0 3. 2 3. / 1	
Plenum wc ⁿ 0.5-4.0 3.5 5.7 <t< td=""><td></td></t<>	
Dost EVGs Wc	
Osit HEME wc	
Dost Charcoal Irap wc wc i	
- Ost File ME	
-ost Cumr vare w	
Ar GC Psig >200 650 650 650 750 Psig >200 650 650 150 Psig >200 650 750 160 Psig >200 7/50 1/20 Psig >200 7/50 1/20 Psig >200 7/50 Psig >200 Psig >20	
He GC psig >200 // 50 // 20 He CC bsig >200 // 50 He CC bsig >200 // 50 He CC bsig >200 // 50 He CC bsig >200 fsig >	
time hitmm 2 2404 0504	
time intervention time into a contract of the	8888

Sheet 13 of 13

Date: 7/6/63

 \mathcal{X}

Reviewed and Approved:

Appendix C

Graphical Temperature, Electrical Parameter, and Noncondensable Off-Gas Data

Graphical Temperature Data



Melter and Off-Gas Temperatures

Melter and Off-Gas Temperatures



Melter and Off-Gas Temperatures







Melter and Off-Gas Temperatures





Melter and Off-Gas Temperatures

Graphical Electrode Parameter Data



RSM Electrode Parameters

RSM Electrode Parameters



RSM Electrode Parameters



RSM Electrode Parameters



RSM Electrode Parameters



RSM Electrode Parameters	RSM	Electrode	Parameters
--------------------------	-----	-----------	------------



Graphical Noncondensible Off-Gas Data



Melter Exhaust Gas Concentrations

Melter Exhaust Gas Concentration, Condition #1: Hg/Cl,Red = Lo/Lo/Lo




Melter Exhaust Gas Concentration, Condition #2: Hg/Cl,Red = Lo/Hi/Lo

Melter Exhaust Gas Concentration, Condition #3: Hg/Cl,Red = Lo/Hi/Hi





Melter Exhaust Gas Concentration, Condition #4: Hg/Cl,Red = Hi/Lo/Lo

Melter Exhaust Gas Concentration, Condition #5: Hg/Cl,Red = Hi/Hi/Lo





Melter Exhaust Gas Concentration, Condition #6: Hg/Cl,Red = Mx/Hi/Lo

Melter Exhaust Gas Concentration, Condition #7: Hg/Cl,Red = Mx/Hi/Hi



Graphical Mercury Off-Gas Data

Melter Exhaust Concentration, Condition #1: Hg/Cl/Red = Lo/Lo/Lo



Mercury Vapor Off-Gas Concentration, Condition #2: Hg/Cl/Red = Lo/Hi/Lo





Mercury Vapor Off-Gas Concentration, Condition #3: Hg/Cl/Red = Lo/Hi/Hi

Melter Exhaust Concentration During Idling





Mercury Vapor Off-Gas Concentration, Condition #4: Hg/Cl/Red = Hi/Lo/Lo

Mercury Vapor Off-Gas Concentration, Condition #5: Hg/Cl/Red = Hi/Hi/Lo





Mercury Vapor Off-Gas Concentration, Condition #7: Hg/Cl/Red = Mx/Hi/Hi



Mercury Vapor Off-Gas Concentration, Condition #6: Hg/Cl/Red = Mx/Hi/Lo

Appendix D

Piping Hardware Photographs

Appendix D: Piping Hardware Photographs

3-RSM Off-Gas Line Component Views





Film Cooler Inlet



Film Cooler Exhaust Port, View A



Inlet of Curved Elbow, View B



Outlet of Curved Elbow, View C



Inlet to Horizontal Pipe Segment, View D



Outlet of Horizontal Pipe Section, View E



Inlet to Pipe Reducer, View F



Outlet of Pipe Reducer, View G



Inlet to EVS



HEME Filter Bed Inlet

Distribution

No. of Copies

OFFSITE

No. of

Copies

ONSITE

2	Savannah River Technology Center		
	Jim Marra		
	Savannah River Site		
	Building 773-43A		
	Aiken, South Carolina 29808		

Harold Sturm Savannah River Site Building 773-A Aiken, South Carolina 29808

11	Battelle—Pacific Northwest Division	
	R. W. Goles	K6-24
	W. W. Buchmiller	K6-24
	L. K. Jagoda	K6-24
	D. E. Kurath	P7-28
	B. D. MacIsaac	K4-26
	S. Rassat	K6-24
	R. L. Russell	K6-24
	Project Office (2)	P7-28
	Information Release (2)	H4-02

3 Bechtel National, Inc.

J. F. Doyle	H4-02
J. M. Perez	H4-02
L. L. Petkus	H4-02